ECONSTOR Make Your Publications Visible.

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Zichella, Giulio

Doctoral Thesis Entrepreneurial Cognition: Three Essays on Entrepreneurial Behavior and Cognition Under Risk and Uncertainty

PhD Series, No. 30.2017

Provided in Cooperation with: Copenhagen Business School (CBS)

Suggested Citation: Zichella, Giulio (2017) : Entrepreneurial Cognition: Three Essays on Entrepreneurial Behavior and Cognition Under Risk and Uncertainty, PhD Series, No. 30.2017, ISBN 9788793579330, Copenhagen Business School (CBS), Frederiksberg, https://hdl.handle.net/10398/9507

This Version is available at: https://hdl.handle.net/10419/209038

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



NC ND https://creativecommons.org/licenses/by-nc-nd/3.0/





COPENHAGEN BUSINESS SCHOOL

SOLBJERG PLADS 3 DK-2000 FREDERIKSBERG DANMARK

WWW.CBS.DK

ISSN 0906-6934

Print ISBN: 978-87-93579-32-3 **Online ISBN: 978-87-93579-33-0**



ENTREPRENEURIAL COGNITION **ON ENTRFPR** IAI **COGNITION UNDER RISK** The PhD School in Economics and Management **PhD Series 30.2017**

Giulio Zichella

THR

AND UNCE

CBS K COPENHAGEN BUSINESS SCHOOL

S

RTAINTY

ENTREPRENEURIAL COGNITION

Three essays on entrepreneurial behavior and cognition under risk and uncertainty

Giulio Zichella

Supervisors: Toke Reichstein, Valentina Tartari The PhD School of Economics and Management Copenhagen Business School Giulio Zichella Entrepreneurial Cognition. Three essays on entrepreneurial behavior and cognition under risk and uncertainty

1st edition 2017 PhD Series 30.2017

© Giulio Zichella

ISSN 0906-6934

Print ISBN: 978-87-93579-32-3 Online ISBN: 978-87-93579-33-0

"The PhD School in Economics and Management is an active national and international research environment at CBS for research degree students who deal with economics and management at business, industry and country level in a theoretical and empirical manner".

All rights reserved.

No parts of this book may be reproduced or transmitted in any form or by any means, electronic or mechanical, including photocopying, recording, or by any information storage or retrieval system, without permission in writing from the publisher.

ENGLISH SUMMARY

Research in decision making and cognition has a long tradition in economics and management and represents a substantial stream of research in entrepreneurship. Risk and uncertainty are two characteristics of the decision environment. It has long been believed that entrepreneurs who need to make business judgments in such environments are less risk- and uncertainty-averse than non-entrepreneurs. However, this theoretical prediction has not been supported by empirical evidence. Instead, entrepreneurs have been found to be more susceptible to cognitive biases and heuristics. These cognitive mechanisms, which represent deviations from rational judgment, help entrepreneurs simplify their decision-making and carry out decisions in a timely manner. As a result, a growing stream of research in entrepreneurship focuses on the cognitive differences between entrepreneurs and non-entrepreneurs when faced with risk and uncertainty.

The purpose of this dissertation is to contribute to this latter stream of research by examining how individuals differ in their cognition and behaviors in situations of risk and uncertainty in a controlled environment. More specifically, the dissertation explores how entrepreneurs and nonentrepreneurs differ in their behavioral susceptibility to prior outcomes, increasing degrees of risk, risk perception, and predictive information. The empirical analyses are based on data from a laboratory experiment that I designed and conducted in October 2014. Individuals participating in the experiment were selected based on their entrepreneurial intentions, with a high degree of comparability and a limited impact of prior entrepreneurial experience. The data include measures of behavior in situations of risk and uncertainty with real monetary incentives, risk perception, and a number of individual characteristics, including personality traits, cognitive biases, and demographics.

In the first essay, I explore how individuals with and without entrepreneurial intentions differ in terms of risk behavior by testing their sensitivity to prior monetary outcomes and an increasing degree of risk. In the second essay, I explore whether individuals with and without entrepreneurial intentions differ in their risk perceptions by examining how they focus their attention on different aspects of risk, specifically either possible monetary outcomes or the probabilities of obtaining these outcomes. In the third and final essay, I test individuals' sensitivity to a lack of predictive information when making choices under uncertainty. In sum, the dissertation contributes to a more nuanced understanding of entrepreneurial cognition in situations of risk and uncertainty by illustrating the direct link between cognition and behavior. Since the dissertation focuses on individuals with limited entrepreneurial experience, it makes important practical contributions with respect to novice entrepreneurs and their cognition in cases of risk and uncertainty. As a result, it provides important insights into how entrepreneurs and non-entrepreneurs differ from a cognitive perspective.

DANSK SAMMENDRAG

Beslutningstagning og kognitiv forskning har en lang tradition i økonomi og ledelse og udgør en væsentlig del af forskningen om iværksætteri. Risiko og usikkerhed er to kendetegn for beslutningsmiljøet. Iværksættere, der tager forretningsbeslutninger i sådan et miljø, har længe menes at være mindre usikkerhed- og risikoavers sammenlignet med ikke-iværksættere. Dog er sådan en teoretisk forudsigelse ikke blevet underbygget af empiriske beviser. I stedet har iværksættere vist sig at være mere modtagelige til at anvende kognitive bias og heuristik end sammenlignelige grupper. Disse kognitive mekanismer, som repræsenterer en afvigelse fra rationalitet i beslutningsprocessen, hjælper iværksættere med at forenkle beslutningsprocessen og træffe beslutninger i tide. Som følge heraf viser en voksende andel af forskningen om iværksættere når de står over for risiko og usikkerhed skyldes kognition.

Formålet med denne afhandling er at bidrage til den sidstnævnte forskning. Dette gøres ved at undersøge i et kontrolleret miljø, hvordan individer adskiller sig i kognition og adfærd når de er står over for risiko og usikkerhed. Afhandlingen undersøger mere specifikt, hvordan iværksættere og ikke-iværksættere er forskelligt modtagelige i deres adfærd når det kommer til forudgående resultater, stigende grader af risiko, risikoopfattelse og forudsigende informationer. De empiriske analyser er baseret på data fra et laboratorieeksperiment, som jeg designede og udførte i oktober 2014. Personerne der deltog i forsøget blev udvalgt på baggrund af deres iværksætterintentioner med en høj grad af sammenlignelighed og en begrænset effekt af forudgående iværksættererfaring. Det empiriske data inkluderer mål på adfærd under risiko og usikkerhed med reelle monetære incitamenter, risikoopfattelse og en række individuelle karakteristika herunder personlighedstræk, kognitive bias og demografi.

I det første essay undersøger jeg, hvordan iværksættere og ikke-iværksættere er forskellige i deres risikoadfærd ved at teste deres følsomhed over for tidligere monetære resultater og stigende grad af risiko. I det andet essay undersøger jeg, om iværksættere og ikke-iværksættere er forskellige i deres risikoopfattelse ved at undersøge, hvordan de fokuserer på forskellige aspekter af risiko og særligt hvordan de enten fokusere på de potentielle monetære resultater eller sandsynligheden for at få dem. I det tredje og sidste essay undersøger jeg iværksættere og ikke-iværksætteres følsomhed over for manglende forudsigende information, når de træffer beslutninger i uvished. Kort sagt bidrager afhandlingen til en mere nuanceret forståelse af iværksætteres kognition, når de står over for risiko og usikkerhed ved at vise den direkte forbindelse til adfærd. Dette giver et vigtigt indblik i hvordan iværksættere og ikke-iværksættere adskiller sig fra hinanden ud fra et kognitivt perspektiv.

ACKNOWLEDGEMENTS

This dissertation is a promise kept. A potential has become reality. At the beginning of my journey as a Ph.D. student, I promised myself that I would pursue my aspirations to grow professionally and personally in academia. Staying true to such a promise has proved to be an exciting, yet challenging, journey. What I have experienced has helped me more than I could ever have imagined. Therefore, I am thankful to all who shared this journey with me. During my years working at Copenhagen Business School, I have been fortunate enough to be surrounded by people whose advice and positive criticism helped me shape my research, avoid mistakes, and arguably become a better person, both professionally and personally. Thank you all for such a great, inspiring time together.

I wish to thank my supervisors for navigating with me through the rough waters of experimental research. I am immensely grateful to my first supervisor, Toke Reichstein. Thanks to his support, insightful criticism, and expertise, my research reached a level I certainly would not have been able to achieve alone. Therefore, grazie, Toke. As an intellectually (over)curious person, I will never forget your advice on how to keep focus and think straight. I also would like to express my very great appreciation to my second supervisor, Valentina Tartari. Her honest feedback and guidance, especially in the critical early stages of my Ph.D., proved indispensable in the long run.

I would like to express my deep gratitude to Jing Chen, Frederic Delmar, and Lars Frederiksen for agreeing to be part of the committee. Your comments were instrumental to further refining my dissertation. A big thanks to Mirjam Van Praag and Orsola Garofalo for taking the time to review my work before submission, and giving comments and suggestions on my experimental design.

During my Ph.D., I had the opportunity to be part of teaching activities. In this respect, I would like to thank, in alphabetical order, Per Vejrup-Hansen, Christian Erik Kampmann, Francesco Rullani, Leonardo Santiago, and Sof Thrane for their guidance. The administrative staff did an excellent job of support. Therefore, thanks for your precious work, help, and patience to Blazenka, Gitte, Katrine, Jette, Lena, and Mie. I want to express my gratitude to the late Annelise Klüwer. I am also thankful to many other people, in particular to Keld Laursen for supervising me as a master's student and guiding me toward an academic career; to Peter Løtz for supporting my research and sharing insights on career opportunities; to Jens Frøslev Christensen, Finn Valentin, and Lee Davis for inspiring me during my innovation studies in the Master of Innovation and Business Development program; the local Ph.D. coordinators, Thomas Rønde and H.C. Kongsted; and all the Ph.D. Days participants for helping me to sharpen my arguments when presenting. Many more people inspired me and made my stay at the INO Department a great experience: Kristina, Christoph, Jörg, Lars Bo, Asma, Ulrich, Mark, Peter, Marion, Vera, Katla, Rene, Henrik, Thomas, Nadika, and Francesca. Thank you all.

My journey to a doctorate has been particularly fruitful due to my fellow Ph.D. colleagues. I want to thank particularly my cohort—Anders, Cecilie and Jacob—for all the laughs, the hard

work, the dinners, the conferences, and the mutual inspiration we shared. Your positive attitudes and friendship mean a lot to me. In addition, I feel very grateful to Olga and Jon for their support, especially during the final stages of our Ph.Ds. Thanks also to the first cohort of Ph.D.s at INO—Virgilio, Maria, Lucas, Arjan, Maggie, Milan, and Gouya—as they helped me to refine my research focus. A big thanks, and good luck for your work in progress, to the current Ph.D. cohort: Aga, Diego, Theo, Adrian, Ahmad, Davide, Stefan, and Hannah.

Two years ago, I had the chance to spend five months as a visiting scholar at the University of California, Los Angeles. This experience gave me the chance to receive feedback on my work and contributed immensely to my understanding of cognitive mechanisms in decision-making under risk and uncertainty. For this reason, I would like to offer my special thanks to my host, Craig R. Fox, and his research group at the Anderson School of Management, particularly Carsten Erner and Helen Colby. I also wish to thank Marie Louise Mors, Wolfgang Sofka, Diego Stea, and Francesco Di Lorenzo for their precious help with the job market. Otto Mønsted Fond and the Entrepreneurship Platform at CBS were generous enough to grant me financial support for my research activities.

As I move on with my academic career, I want to express my gratitude to all the great people I was blessed to work with at the Operations Management Department. I want to thank in particular Sof Thrane for his precious guidance in both teaching and research; Carsten Ørts Hansen for his support and vision; and Andreas, Birte, Cristiana, Christian, Eric, Juliana, Liping, Mikkel, Pernille, and all the people I had the pleasure to interact with. Thank you all.

I want to thank my friends and family for supporting me during my Ph.D. I am particularly thankful to the members of the Italian Choir at Jesu Hjerte Kirke and my bandmates, Dawn Gazers. Thank you for making me laugh and for praying for me, both of which were very important during these years: Larissa, Loredana, Maria, Chiara, Francesco, Giuseppe, Father Robert, Giulia, Paola, Francesca, Marco, Lone, Federico, Benjamin, and Lucia. A special thanks goes to my brothers, sisters and cousins: Flora Emma, Elisabetta, Maria Giulia, Francesca, Alfonso, Pietro, and Gianluca.

Finally, I want to thank my mother, Anna; Mane; and my wife, Jenny. Thank you for being the best traveling companions anyone could wish for.

This dissertation is dedicated to the loving memory of my father, Giorgio Zichella (1947-2012). A promise kept for you.

Giulio Zichella

Frederiksberg, July 2017

TABLE OF CONTENT

CHAPTER 1: INTRODUCTION	8
CHAPTER 2. A RISK WORTH TAKING: RISK AS THE ENTREPRENEUR'S CONDIT CHOICE	IONAL 21
CHAPTER 3. ENTREPRENEURS' FOCUS OF ATTENTION AND PERCEPTION OF FIRISK	INANCIAL 71
CHAPTER 4. IT'S ALL OR NOTHING: ENTREPRENEURS' WILLINGNESS TO BEAF UNCERTAINTY	₹ 106
CHAPTER 5, CONCLUSIONS	

CHAPTER 1: INTRODUCTION

This dissertation consists of three chapters that explore how two groups—individuals with and without entrepreneurial intentions—differ in their cognition and behavior in situations of risk and uncertainty. Chapter 2 finds that contextual factors make individuals with entrepreneurial intentions more willing to choose risk than individuals without entrepreneurial intentions. Chapter 3 shows how the two groups differ in two cognitive respects (focus of attention and risk perception) when choosing between investment opportunities. Chapter 4 shows that individuals with entrepreneurial intentions are less sensitive than individuals without entrepreneurial intentions to a lack of information on the probabilities of monetary outcomes.

This dissertation seeks to contribute to entrepreneurship literature by exploring the cognition and behavior mechanisms that underlie entrepreneurial actions and how these mechanisms differ between individuals with and without entrepreneurial intentions in situations of risk and uncertainty (Mitchell et al., 2002, 2007). Business environments push individuals to make decisions with limited information, and the ways in which they select and make use of such information play an important role in guiding their behavior (Busenitz & Barney, 1997). For this reason, I focus my research efforts on understanding how specific pieces of information increase risk and uncertainty, taking into account how cognitive biases affect individuals with entrepreneurial intentions versus individuals without entrepreneurial intentions. More specifically, I explore: (i) how prior monetary gains increase individuals' bias towards risk-taking under different degrees of risk; (ii) how *risk perception*, defined as the subjective assessment of risk in a given opportunity (Weber et al., 2002), and *focus of attention*, defined as an individual's selection of available information through attention (March & Shapira, 1987, 1992), are unique aspects of entrepreneurs' decision making in situations of risk; and (iii) how

individuals with entrepreneurial intentions are less sensitive to an absence of information about probabilities and, hence, more willing to choose uncertain monetary opportunities.

Risk and Uncertainty in Entrepreneurship: An Overview of the Chapters' Complementarity

Since the work of Frank Knight (1921), risk and uncertainty, defined as measurable and immeasurable risk, respectively, have been treated as two separate concepts in entrepreneurship research. Entrepreneurs operate under both risk and uncertainty, depending on the information available. While it is unrealistic to think about entrepreneurs as individuals operating under a complete lack of predictive information (e.g. on returns of investment opportunities), it is important to understand how entrepreneurs use available information to make decisions. For this reason, in this dissertation, I distinguish both conceptually and empirically between risk and uncertainty in Chapters 2 and 4. In particular, in Chapter 2, I manipulate information on prior winnings and degree of risk, while in Chapter 4, I manipulate predictive information. In Chapter 3, I qualitatively assess individuals' use of available information with a focus on risk perception. Overall, these three chapters present complementary views on how cognition and decision making in situations of risk and uncertainty differ among individuals with different entrepreneurial intentions.

Risk is an important factor in the entrepreneurial environment. An entrepreneur can be defined as someone who runs and bears the risk of a business. In this light, research on entrepreneurs' risk-taking has focused primarily on understanding whether (and how) entrepreneurs differ from non-entrepreneurs in their stable and contextual risk preferences. Based on prior research in economics, entrepreneurs are typically considered (and, to some

degree, still believed to be) less risk averse than non-entrepreneurs (Kihlstrom & Laffont, 1979). Yet, empirical evidence has not confirmed such perceptions (Brockhaus, 1980; Holm et al., 2013; Koudstaal et al., 2015; Stewart & Roth, 2001), hinting instead that there are cognitive differences between entrepreneurs and non-entrepreneurs with respect to risk-taking. To test these differences, in Chapters 2 and 3, I focus only on financial risk. Overall, I argue for and find support for the greater importance of two specific cognitive biases—the prior gain effect and the risk propensity effect—in individuals' increased risk-seeking behavior. Furthermore, I test how individuals with and without entrepreneurial intentions differ in their risk perceptions, which is important for explaining behavior in cases of risk (e.g. opportunity evaluation and choice; Keh et al., 2002; Simon et al., 2000).

Uncertainty is another important element in the entrepreneurial environment. Especially when introducing highly innovative products or services, entrepreneurs often lack predictive information on key variables, such as sales forecasts or returns. Research on entrepreneurs' behavior in cases of uncertainty is relatively more recent than research on risk and often questions how entrepreneurs make use of predictive information (McKelvie et al., 2011). According to effectuation theory, entrepreneurs use heuristics and biases to shape (rather than predict) an uncertain future (Sarasvathy, 2001). In Chapter 4, I operationalize uncertainty as the complete lack of information on monetary opportunities, arguing that individuals with entrepreneurial intentions have a greater propensity to bear uncertainty than individuals without entrepreneurial intentions.

The importance of understanding how entrepreneurs act in situations of risk and uncertainty—and how such behavior can be optimized—is underlined by the emergence of a recent research stream in entrepreneurship that uses experimental evidence to explore the topic (for a review, see Shepherd, 2015; Shepherd et al., 2015). Cognitive research requires a careful examination of the causal link between behavior and cognition. In such cases, an experimental design (as opposed to the use of questionnaires) is most appropriate. The overall challenge for this dissertation is to observe and measure differences in cognition and behavior in situations of risk and uncertainty between comparable samples of individuals with and without entrepreneurial intentions.

Empirical Setting and Sample Relevance

To answer these research questions, I collected data from a quasi-laboratory experiment that I designed and conducted in October 2014. The subjects comprised 72 individuals with different entrepreneurial intentions. The sample was chosen for several reasons: first, it sought to facilitate an understanding of cognition before entrepreneurial experience. Entrepreneurial intentions¹—defined here as an individual's active interest in starting a firm—are appropriate for testing whether some of the cognitive mechanisms found in experienced entrepreneurs can also be found in individuals with limited entrepreneurial experience. By using students, I can compare individuals with similar personal and professional backgrounds, but different entrepreneurial intentions. Second, a sample of individuals with entrepreneurial intentions, arguing for a high predictive power of planned behavior over decisions of self-employment (Krueger et al., 2000, 2007). Third, using individuals with entrepreneurial intentions is relevant for practice.

¹ The more general definition of entrepreneurial intentions is: "the cognitive state temporally and causally prior to action" (Krueger, 2009, p. 51). In 2016, I surveyed the current status of individuals with entrepreneurial intentions. Approximately 80% of the respondents answered that they had started a firm within two years of the experiment. I cannot rule out the possibility that entrepreneurial intent fully captures differences between entrepreneurs and non-entrepreneurs; yet, I do believe that these results strongly indicate that I have worked with a valuable proxy for entrepreneurship.

limited entrepreneurial experience (e.g. students, novice entrepreneurs). It is important for these institutions to understand how to support these individuals in an effective way (Amezcua et al., 2013). Finally, using the same sample of individuals throughout the dissertation allows me to compare results and provide a complementary picture of differences in cognition and behavior in situations of risk and uncertainty between individuals with different entrepreneurial intentions.

The experiment was programmed in Z-tree (Fischbacher, 2007) and consisted of three parts. In the first part, subjects were presented with a series of real money games involving a total of 24 unique binary choices. Each choice consisted of two monetary opportunities with identical expected values, but different combinations of risk and uncertainty. In the second part, subjects were presented with a series of repeated choices between investments. The design for the simplified investment opportunities was adapted from prior research (Sarasvathy et al., 1998). Finally, the subjects were given a questionnaire capturing demographics, cognitive biases, and personality traits. The models used to analyze the data included regressions (for choices in situations of risk and uncertainty) and verbal content analysis (for focus of attention and risk perception). With particular reference to the latter, three independent coders performed the content analysis and successfully followed the codification protocols.

Contributions to Extant Research

The extant research has yielded mixed evidence regarding entrepreneurs' greater propensity towards risk. My first contribution shows how two specific contextual mechanisms grounded in cognitive psychology and economics—the prior gain effect and the risk propensity effect—play a primary role in observed differences between the behaviors of entrepreneurs and non-entrepreneurs in situations of risk. In particular, prior monetary gains increase entrepreneurs' risk-taking more than non-entrepreneurs'. By contrast, increasing degrees of risk reduces the differences in observed risk-taking behaviors among groups, indicating similar levels of bias in situations of high risk. This latter finding helps to explain why differences between entrepreneurs and non-entrepreneurs in terms of risk-taking propensities are often not visible (Brockhaus, 1980) without considering the effect of cognitive biases. These findings offer two practical contributions. First, since entrepreneurs often use their experience when pursuing risky opportunities, prior monetary gain from investment decisions may push them to take unnecessary risks in subsequent investment decisions. For this reason, it is important for private stakeholders to consider the potentially detrimental effect of entrepreneurs' successful prior investment decisions. Second, my results indicate that entrepreneurs are not blind risktakers; rather, they shy away from high predicted monetary outcomes when the risk is too high. For this reason, educational programs aimed at helping entrepreneurs make more thoughtful decisions should focus on the contextual elements (e.g. information, expert opinions, etc.) that drive entrepreneurs to invest in new ventures.

Though risk perception is an important variable in fostering individuals' behavior in situations of risk, very little is known about entrepreneurs' risk perceptions and whether (and how) it differs from those of non-entrepreneurs (Shepherd et al., 2015; Simon et al., 2000). My second contribution shows differences between individuals' risk perceptions are driven by different foci of attention. In particular, I find that individuals focus on different characteristics of investment opportunities, with entrepreneurs focusing more on monetary outcomes and non-entrepreneurs focusing more on probabilities. Such differences in focus translate into differences in perception, with entrepreneurs perceiving more risk when they focus on the size of the monetary outcomes than when they focus on probabilities. These findings offer two practical

contributions. First, it is advisable to inform stakeholders that entrepreneurs use a different logic than non-entrepreneurs and that differences in foci of attention affect investment selection (Sarasvathy et al., 1998, 2001). For this reason, both entrepreneurs and stakeholders (particularly investors) need to align their assessment of risk in new investment decisions by considering their foci on control and attention. Second, I show that risk perception does matter in entrepreneurs' choices as it increases with their focus on investments' desirable elements. This finding contributes to our understanding of whether entrepreneurs perceive the riskiness of their actions, providing an interesting point of departure for future research on entrepreneurs' perceptions of risk.

Entrepreneurs must often make decisions under uncertainty, which can be defined as immeasurable risk (Knight, 1921). Thus, my third contribution is showing how an absence of information regarding probabilities does not impact entrepreneurs' decisions to the same extent that it does for non-entrepreneurs. In particular, entrepreneurs are more likely than non-entrepreneurs to choose consistently between two prospects that have everything in common except for probability information. This result, while confirming the findings of Chapter 3 (i.e. that entrepreneurs do not focus their attention on probabilities) also confirms that entrepreneurs have a greater willingness to bear uncertainty if the stakes at play (in this case, monetary outcomes) are desirable.

In sum, this dissertation advances our understanding of the roles of cognition and information in driving differences between entrepreneurs and non-entrepreneurs, ultimately helping stakeholders (investors and public institutions) in their understanding of the motives behind entrepreneurial action.

14

Contributions to Entrepreneurship in Practice

The dissertation seeks to contribute to the field of entrepreneurship in practice by informing public-private stakeholders about entrepreneurs' behavior, cognitions, and perceptions in situations of risk and uncertainty. By illustrating the cognitive differences between individuals with and without entrepreneurial intentions, the three chapters present several complementary suggestions for managing relationships between entrepreneurs and their stakeholders. In so doing, this dissertation offers three main contributions: first, stakeholders who work with entrepreneurs, especially those who work with novice or student entrepreneurs (e.g. incubators, accelerators), need to carefully manage entrepreneurs' bias towards prior gains in future risk choices (e.g., by reviewing entrepreneurs' recent past experience and financial results). The results of Chapter 2 suggest that individuals with entrepreneurial intentions are more likely than individuals without such intentions to choose risky monetary opportunities right after having received a monetary gain, even when such a gain is due to a random process. Second, such stakeholders and entrepreneurs need to strategically align their assessment of risk with respect to business opportunities (e.g., by rationally analyzing and comparing all of the elements involved in investment opportunities). The results of Chapter 3 suggest that, in line with the reasoning of Sarasvathy et al. (1998), individuals with entrepreneurial intentions tend to perceive less risk than individuals without such intentions because of their focus of attention. Stakeholders must consider that entrepreneurs do not use predictive information on investment feasibility as a primary factor in investment choices and, as a consequence, should carefully manage entrepreneurs' critical assessment of information. Finally, Chapter 4 bridges the findings of Chapters 2 and 4 by testing individuals' risk behavior and cognition in situations of Knightian (1921) uncertainty, defined as the absence of predictive information on probabilities. In particular, in situations of uncertainty, individuals with entrepreneurial intentions are less sensitive than individuals without entrepreneurial intentions to the lack of probabilities when choosing between opportunities. The results of Chapter 4 suggest that stakeholders should understand how entrepreneurs evaluate monetary opportunities in terms of possible monetary objectives.

Limitations and Avenues for Future Research

As part of a critical assessment of my work, I acknowledge several limitations that also represent potential avenues for future research and contributions in the field of entrepreneurship. First, the definition of entrepreneurship is here limited to individuals with entrepreneurial intentions. Despite the research and practical importance of intentions in entrepreneurship², my results are more relevant for groups of individuals at the beginning of their entrepreneurial careers, particularly novice entrepreneurs. It is worth noting that the definition of entrepreneurial intentions used in this dissertation is further restricted by individuals' types of intentions. Virtually all of the students participating in the experiment intended to start firms based on an innovative idea or service (i.e. they did not intend to be "self-employed" as, for example, journalists). Clearly, this affects the type of novice entrepreneurs to whom the results speak. Future research may compare individuals at different stages in their entrepreneurial careers with relevant comparison groups (e.g. managers). Second, my measurement of the effect of monetary rewards on choice behavior is limited because I do not use a long-term perspective. This was a deliberate choice, given my focus on testing the causal link between cognition and action in situations of risk and uncertainty. However, since monetary rewards are typically delayed in time, my research is particularly relevant for choices made in a limited time frame. Future

 $^{^{2}}$ Intentions are described as an accurate predictor of entrepreneurial self-selection (Krueger, 2000) and, as a post-experiment sample check, a large share of those who intended to start a firm in the next three years actually started a firm.

research may explore the effect of rewards on entrepreneurial risk-taking when time is manipulated as a critical variable. Monetary rewards delayed in time would require a careful assessment of time preferences (Abdellaoui, Diecidue, & Öncüler, 2011; Andreoni & Sprenger, 2012) and, particularly for entrepreneurs, learning effects for binary choices (Das & Teng, 1997; March & Shapira, 1992; Minniti & Bygrave, 2001; Oskarsson, Van Boven, McClelland, & Hastie, 2009). Third, despite the power of experiments to identify precise causal links between cognition and behavior, experiments always represent a simplification of reality. Individual cognition and stable traits influence risk behavior concurrently. I acknowledge that I cannot identify all of the stable traits that may influence behavior. However, in each chapter, I address strategies to test (and exclude) possible alternative explanations for the results. Fourth, due to experimental constraints in terms of time, I did not measure (and therefore cannot control for) several elements pertaining to the psychology of the entrepreneur that may affects risk behavior, including locus of control (Mueller & Thomas, 2001), self-efficacy (Cassar & Friedman, 2009), and need for achievement (Stewart et al., 2003). Finally, family culture is an important element explaining entrepreneurial decisions in situations of risk (e.g. terminating an investment, Sharma & Manikutty, 2005). Though I control for some family demographics (e.g. parental selfemployment, size and industry of parents' firms), culture remains a source of unobserved heterogeneity.

REFERENCES

Abdellaoui, M., Diecidue, E., & Öncüler, A. (2011). Risk preferences at different time periods: An experimental investigation. *Management Science*, *57*(5), 975-987.

Amezcua, A. S., Grimes, M. G., Bradley, S. W., & Wiklund, J. (2013). Organizational sponsorship and founding environments: a contingency view on the survival of business-incubated firms, 1994–2007. *Academy of Management Journal*, *56*(6), 1628-1654.

Andreoni, J., & Sprenger, C. (2012). Risk preferences are not time preferences. *The American Economic Review*, *102*(7), 3357-3376.

Brockhaus RH (1980) Risk taking propensity of entrepreneurs. Academy of Management Journal, 23(3), 509–520.

Busenitz, L. W., & Barney, J. B. (1997). Differences between entrepreneurs and managers in large organizations: Biases and heuristics in strategic decision-making. *Journal of Business Venturing*, *12*(1), 9–30.

Cassar, G., & Friedman, H. (2009). Does self-efficacy affect entrepreneurial investment? *Strategic Entrepreneurship Journal*, *3*(3), 241–260.

Das, T. K., & Teng, B. S. (1997). Time and entrepreneurial risk behavior. *Entrepreneurship: Theory and Practice*, 22(2), 69–71.

Fischbacher, U. (2007). z-Tree: Zurich toolbox for ready-made economic experiments. *Experimental Economics*, *10*(2), 171–178.

Kahneman, D. & Tversky, A. (1979). Prospect theory: An analysis of decision under risk. *Econometrica: Journal of the Econometric Society*, Vol. 47, 263–291.

Keh, H. T., Foo, M. D., & Lim, B. C. (2002). Opportunity evaluation under risky conditions: The cognitive processes of entrepreneurs. *Entrepreneurship Theory and Practice*, 27(2), 125–148.

Knight, F. H. (1921). Risk, uncertainty and profit. New York, NY: Hart, Schaffner and Marx.

Koudstaal, M., Sloof, R., & Van Praag, M. (2015). Risk, uncertainty, and entrepreneurship: Evidence from a lab-in-the-field experiment. *Management Science*, 62(10), 2897–2915.

Krueger, N. F., Reilly, M. D., & Carsrud, A. L. (2000). Competing models of entrepreneurial intentions. *Journal of Business Venturing*, 15(5), 411–432.

Krueger, N. (2009). Entrepreneurial intentions are dead: Long live entrepreneurial intentions. In *Understanding the entrepreneurial mind* (pp. 51–72). New York, NY: Springer.

March, J. G., & Shapira, Z. (1987). Managerial perspectives on risk and risk taking. *Management Science*, 33(11), 1404–1418.

March, J. G., & Shapira, Z. (1992). Variable risk preferences and the focus of attention. *Psychological Review*, 99(1), 172-183.

McKelvie, A., Haynie, J. M., & Gustavsson, V. (2011). Unpacking the uncertainty construct: Implications for entrepreneurial action. *Journal of Business Venturing*, *26*(3), 273–292.

Minniti, M., & Bygrave, W. (2001). A dynamic model of entrepreneurial learning. *Entrepreneurship: Theory and Practice*, 25(3), 5–12

Mitchell, R. K., Busenitz, L., Lant, T., McDougall, P. P., Morse, E. A., & Smith, J. B. (2002). Toward a theory of entrepreneurial cognition: Rethinking the people side of entrepreneurship research. *Entrepreneurship Theory and Practice*, 27(2), 93–104.

Mitchell, R. K., Busenitz, L. W., Bird, B., Marie Gaglio, C., McMullen, J. S., Morse, E. A., & Smith, J. B. (2007). The central question in entrepreneurial cognition research. *Entrepreneurship Theory and Practice*, *31*(1), 1–27.

Mueller, S. L., & Thomas, A. S. (2001). Culture and entrepreneurial potential: A nine country study of locus of control and innovativeness. *Journal of Business Venturing*, *16*(1), 51–75.

Oskarsson, A. T., Van Boven, L., McClelland, G. H., & Hastie, R. (2009). What's next? Judging sequences of binary events. *Psychological Bulletin*, *135*(2), 262-285.

Sarasvathy, S. D. (2001). Causation and effectuation: Toward a theoretical shift from economic inevitability to entrepreneurial contingency. *Academy of Management Review*, *26*(2), 243–263.

Sarasvathy, D. K., Simon, H. A., & Lave, L. (1998). Perceiving and managing business risks: Differences between entrepreneurs and bankers. *Journal of Economic Behavior & Organization*, 33(2), 207–225.

Sharma, P., & Manikutty, S. (2005). Strategic divestments in family firms: Role of family structure and community culture. *Entrepreneurship Theory and Practice*, *29*(3), 293–311.

Shepherd, D. A. (2015). Party on! A call for entrepreneurship research that is more interactive, activity based, cognitively hot, compassionate, and prosocial. *Journal of Business Venturing*, *30*(4), 489–507.

Shepherd, D. A., Williams, T. A., & Patzelt, H. (2015). Thinking about entrepreneurial decision making review and research agenda. *Journal of Management*, *41*(1), 11–46.

Simon, M., Houghton, S. M., & Aquino, K. (2000). Cognitive biases, risk perception, and venture formation: How individuals decide to start companies. *Journal of Business Venturing*, *15*(2), 113–134.

Stewart, W. H., & Roth, P. L. (2001). Risk propensity differences between entrepreneurs and managers: A meta analytic review. *Journal of Applied Psychology*, *86*, 145–153.

Stewart Jr., W. H., Carland, J. C., Carland, J. W., Watson, W. E., & Sweo, R. (2003). Entrepreneurial dispositions and goal orientations: A comparative exploration of United States and Russian entrepreneurs. *Journal of Small Business Management*, *41*(1), 27–46.

Thaler, R. H., Johnson, E. J. (1990). Gambling with the house money and trying to break even: The effects of prior outcomes on risky choice. *Management Science*, *36*(6), 643–660.

Weber, E. U., Blais, A. R., & Betz, N. E. (2002). A domain-specific risk-attitude scale: Measuring risk perceptions and risk behaviors. *Journal of Behavioral Decision Making*, 15, 263–290.

CHAPTER 2. A RISK WORTH TAKING: RISK AS THE ENTREPRENEUR'S CONDITIONAL CHOICE

Giulio Zichella - Copenhagen Business School, gz.ino@cbs.dk

Toke Reichstein - Copenhagen Business School, tr.smg@cbs.dk

ABSTRACT

It is theorized that entrepreneurs are more likely than non-entrepreneurs to make risky choices. However, the empirical evidence addressing this claim is mixed. This paper offers new insights into entrepreneurs' choices related to risk, investigating the circumstances in which individuals with entrepreneurial intentions are more or less likely to choose risk over certainty. Drawing on prospect theory, we formulate hypotheses about the greater likelihood that individuals with entrepreneurial intentions will choose risk immediately after a positive gain, but shy away from risk as the degree of risk increases. We test the hypotheses using data collected in laboratory-based real money games experiments and find support for the hypotheses, indicating that entrepreneurs' bias towards risk is circumstantial. These results have fundamental implications for our understanding of the factors guiding entrepreneurial choices in situations of risk.

Keywords: risk, entrepreneur, money games, lab experiment, prior gain effect, risk propensity effect

JEL Classifications: L26, D81

INTRODUCTION

How do individuals with and without entrepreneurial intentions differ in their propensity to choose risk? Prior research offers mixed evidence relating to this research question. Some studies suggest that entrepreneurs are more likely to choose risk than non-entrepreneurs (see e.g. Begley & Boyd, 1988; Carland III et al., 1995; Chen et al., 1998; Cramer et al., 2002; Kihlstrom & Laffont, 1979; Mullins & Forlani, 2005; Stewart et al., 1999), while others suggest that there are no differences between these groups (see e.g., Brockhaus, 1980; Holm et al., 2013; Koudstaal et al., 2015; Simon et al., 2000). The absence of a definitive answer to this question represents a conundrum in entrepreneurship research. It is important to resolve this dilemma because both individuals' recognition of entrepreneurial opportunities and their propensity to pursue them are central aspects explaining self-selection into entrepreneurship, which is a relevant phenomenon for public welfare (Baron & Ensley, 2006; Kihlstrom & Laffont, 1979; Shane & Venkataraman, 2000). Further, a better understanding of this issue might help entrepreneurs make more rational and thoughtful decisions and help their stakeholders understand and manage their actions. Finally, by investigating this question, we respond to Shepherd et al.'s (2015) call for a better understanding of the contextual factors that magnify or diminish the effects of individual characteristics on entrepreneurial decision-making (e.g. whether an increasing degree of risk influences entrepreneurs and non-entrepreneurs differently).

The premises of this study are that real decision-makers are influenced by cognitive biases (Busenitz & Barney, 1997; Deligonul, Hult, & Cavusgil, 2008; Kahneman & Tversky, 1979; March, 1994), that entrepreneurs do not choose risk at all costs, and that their choices are circumstantial. We use entrepreneurial intentions as a proxy for entrepreneurship, since intentions are a powerful predictor of entrepreneurship (Krueger, Reilly, & Carsrud, 2000). We

investigate two circumstances that might cause entrepreneurs to respond to risk differently than non-entrepreneurs. First, drawing on the arguments proposed by Thaler and Johnson (1990) and Post et al. (2008), individuals may exhibit increased risk-seeking behavior when they have recently (i.e. immediately before) experienced gains. This has been documented in the presence of gambling both with only positive monetary outcomes ("prior gain effect;" Battalio et al., 1990) and with positive and negative outcomes ("house money effect;" Thaler & Johnson, 1990). Both of these effects represent extensions of the decreasing absolute risk aversion (DARA) prediction that individuals will hold more money in risky assets as their wealth increases (Holt & Laury, 2002). Drawing on prospect theory, we extend our explanation by focusing on the effects of immediate prior gains on subsequent choices, while controlling for accumulated wealth. Since entrepreneurs are characterized by representativeness³ and, hence, make choices based on a few observations (Forbes, 2005; Keh et al., 2002; Simon et al., 2000), we suggest that a gain may generate a more immediate bias towards risk among entrepreneurs than among non-entrepreneurs. That is, entrepreneurs may exhibit a bias towards a potential gain when they have recently experienced a gain from a previous choice. This is consistent with the finding that entrepreneurs often rely on their personal experience when evaluating opportunities (Cassar & Craig, 2009; Fern et al., 2012; Parker, 2006; Stuart & Abetti, 1990).

Second, an increased level of risk has been shown to be key in explaining the differences among individuals' revealed preferences (Rothschild & Stiglitz, 1970; Sarasvathy et al., 1998; Simon et al., 2000). However, it has been argued that individuals tend to exhibit a decreasing tendency to choose options characterized by high levels of risk. Entrepreneurs, however, value small gains/winnings. As a result, their perceived utility from small gains has a greater chance of

³ The representativeness heuristic is defined as the degree to which an event is similar in essential characteristics to its parent population and reflects the salient features of the process by which it is generated (Kahneman & Tversky, 1972).

exceeding the disutility from choosing low-risk options than among non-entrepreneurs. Consequently, entrepreneurs may be more likely to choose risk in general, but at higher levels of risk, the difference is evened out because the disutility of the increasing risk exceeds the potential gain, even for entrepreneurs.

Entrepreneurship research investigates risk primarily by eliciting entrepreneurs' propensities indirectly through their responses to questionnaires where there is nothing at stake (Brockhaus, 1980; Sarasvathy et al., 1998). However, there is solid body of research in economics that argues for the importance of using real incentives when comparing groups' decision-making, since real incentives increase risk-averse behavior (Holt & Laury, 2002; Kachelmeier & Shehata, 1992; Weber & Milliman, 1997). Furthermore, the presence of real monetary incentives is necessary to account for prior experience (e.g., monetary feedback, Post, Assem, Baltussen, & Thaler, 2008; Thaler & Johnson, 1990) and low-probability events (Schade et al., 2012). A few scholars have tested the risk preferences of entrepreneurs in the presence of real monetary incentives (Holm et al., 2013; Koudstaal et al., 2015). In this paper, we use quasilaboratory experiments⁴ with real monetary incentives. We investigate the behaviors of students enrolled in a general business economics program at a major business school. After the first semester, one group actively selected to enroll in a designated entrepreneurship program, motivated by the desire to pursue careers as entrepreneurs. Thus, our experiment included a sample of "entrepreneurs," defined as individuals demonstrating entrepreneurial intentions (the cognitive state that precedes the decision to launch a new venture) (Krueger et al., 2000). The second sample consisted of general business economics students with no such declared intentions. The choice was made to use students rather than a random sample from the greater population in order to limit variations across subjects in terms of demographics and professional

⁴ We use a quasi-laboratory experiment design because the subjects are not randomly selected, but are instead assigned to two groups based on a specific characteristic: their entrepreneurial intentions (Shadish et al. 2002).

experience, thereby isolating the effects of prior gain and risk propensity. Furthermore, the use of students with entrepreneurial intentions limited concerns related to external validity, particularly with respect to novice entrepreneurs (Krueger et al., 2000; Liñan & Chen, 2009).

The experiment required subjects to make choices between certainty and risk in the presence of real monetary incentives. These incentives took the form of one-to-one payoffs from gambling gains. We separately introduced two exogenous manipulations: feedback on monetary outcomes and an increasing variance of risky outcomes. A control for individual risk attitude (Holt & Laury, 2002) was included to separate the effects of these manipulations. The results revealed that, with respect to a certain monetary gain, individuals with entrepreneurial intentions were more likely to favor risky options than individuals from the control sample. However, this finding held only when controlling for prior gains and degree of risk. Furthermore, prior gains had a larger positive impact on the risk choices of individuals with entrepreneurial intentions. In addition, the degree of risk negatively moderates the likelihood that an individual with entrepreneurial intentions will choose risk over certainty, compared to the control sample. Overall, these results confirm the importance of empirical investigations of entrepreneurial decision-making in the presence of real monetary incentives and suggest that the differences in risk preferences between entrepreneurs and non-entrepreneurs are circumstantial.

The remainder of the paper is organized as follows: Section 2 briefly reviews the relevant literature and proposes arguments leading to hypotheses on the prior gain effect and the risk propensity effect. This section seeks specifically to articulate the reasoning justifying testing the different effects comparing entrepreneurs with non-entrepreneurs. Section 3 describes the data and the construction of the sample and provides details concerning the experiment and the methods used to test the proposed hypotheses. Section 4 presents the results. Section 5 concludes and discusses the implications of the findings.

25

THEORY AND HYPOTHESES

Understanding entrepreneurs' behavior vis-a-vis risk is considered important because entrepreneurship is characterized by risk. It is theorized that entrepreneurs are less risk-averse than non-entrepreneurs. Kihlstrom and Laffont's (1979) model predicts that fewer risk-averse individuals become entrepreneurs and that the least risk-averse entrepreneurs run larger firms. One of the underlying arguments with regard to entrepreneurs is that they are well equipped to cope with environmental uncertainty and complexity because they are characterized by cognitive biases, heuristics, *overconfidence*, *representativeness*, and a belief in the *law of small numbers* (Busenitz & Barney, 1997). These characteristics better equip them to make effective decisions in entrepreneurial settings, which lead them to self-select into entrepreneurship. Cognitive biases and a heuristics-based decision-making approach favor the performance of entrepreneurial activities, since entrepreneurs often face high levels of monetary risk.

These theoretical arguments are at the basis of numerous studies that attempt to determine whether entrepreneurs differ from non-entrepreneurs in terms of making risky choices, where risk is considered to involve situations in which a decision-maker chooses between different alternatives with complete information about the probability of different outcomes (Ellsberg, 1961; Knight, 1921). However, empirical investigations suggest that the association is less straightforward, and there is a growing body of work on risk in entrepreneurship that provides mixed evidence. Numerous studies find empirical support for the proposition that entrepreneurs exhibit higher risk-taking propensities than non-entrepreneurs (see e.g. Begley & Boyd, 1988; Carland III et al., 1995; Stewart et al., 1999). For instance, Moskowitz and Vissing-Jorgensen (2002) found that entrepreneurs do not spread their risk when making portfolio investments, suggesting a lower aversion to risk. Moreover, in a study

investigating school children over a period of 40 years, Cramer et al. (2002) found evidence of a negative association between risk aversion and self-selection into entrepreneurship.

On the other hand, other studies suggest that entrepreneurs are no different from nonentrepreneurs in their propensity to choose risk. Brockhaus (1980) found no empirical evidence to support differences in the risk-taking propensities of entrepreneurs and non-entrepreneurs. Koudstaal et al. (2015) showed that entrepreneurs perceive themselves to be more risk-tolerant than non-entrepreneurs, but also showed that the same subjects, when exposed to incentivized gambles, do not exhibit a higher propensity to choose risk over certainty. We contribute to these findings by showing how circumstantial factors may make entrepreneurs more likely than nonentrepreneurs to choose risk, thereby providing a possible explanation for entrepreneurs' perceptions of their own propensities for risk-taking. Holm et al. (2013) hinted that circumstance-dependent factors may influence the differences in observed risk propensities. Specifically, they find that entrepreneurs are found to differ from non-entrepreneurs in terms of their willingness to accept risk only when choices involve strategic uncertainty, which depends on a decision-maker's beliefs about his or her own performance relative to that of others.

The mixed results regarding risk propensities in entrepreneurship research might suggest that the association is attributable to the contingencies of the decision, as indicated by Holm et al. (2013). We extend this perspective by proposing two contingencies that might provide a more fine-grained understanding of these mixed empirical results for entrepreneurs compared to non-entrepreneurs. First, differences in risk tolerance between entrepreneurs and comparable others might depend on an individual's prior monetary gains/losses. Second, differences between entrepreneurs' and non-entrepreneurs' propensities to choose risk may depend on the level of that risk.

Prior gains and the biased entrepreneur

Prospect theory suggests that individual decisions are affected by gains or losses that alter people's expectations and aspirations. In the words of Kahneman and Tversky (1979), "there are situations in which gains and losses are coded relative to an expectation or aspiration level that differs from the status quo" (p. 286). Decision-makers may, hence, be influenced by and adapt their aspirations based on prior experience when evaluating real monetary opportunities (Oskarsson et al., 2009). Both their satisfaction and their dissatisfaction may be dynamically adapted to feed back into and generate expectations about future outcomes in the short run (Andersen et al., 2014; March, 1994). This cognitive bias may cause individuals to perceive random events as non-random and, ultimately, cause them to adapt their aspirations to their cognitively biased view of opportunities.⁵

Thaler and Johnson (1990) and Post et al. (2008) provided evidence that individuals exhibit a high propensity to choose options characterized by risk after experiencing a monetary gain.⁶ The reasoning here is that the individual's reference point changes. Having already won a gamble, individuals reason that they may win again in the subsequent gamble. This behavior, which is based on a biased cognitive assessment of a random sequence of gains, is

⁵ Research on individual decision making recognizes that decision makers use multiple reference points when constructing preferences out of choice options (Heath et al., 1999; Köszegi & Rabin, 2006; Stewart et al. 2003). How prior gains affect risk-taking behavior is not part of the original prospect theory, but has attracted substantial attention due to the potential of such gains to change individuals' reference points (Novemsky & Dhar, 2005; Post et al., 2008; Weber & Johnson, 2009). In this paper, the reference point corresponds to a monetary outcome that a subject expects to attain.

⁶ Thaler and Johnson's (1990) paper develops the concept of the house money effect, referring to an individual having a sense of "gambling with the house money" rather than his or her own money. This effect is investigated by comparing an individual's behavior when gambling with his or her own money to his or her behavior when gambling with money won from the "house." This phenomenon does not apply in our gambling experiment because individuals do not gamble money won in prior gambles; in other words, they are not putting their winnings at risk. The present experiment is a simplified version of the house money experiment that does not adopt the house money effect terminology.

fundamentally different from a simple adaptation to accumulated wealth⁷. Accordingly, individuals who have achieved prior gains exhibit a lower aversion to risk.

The prior gains effect may be stronger for individuals with entrepreneurial intentions than comparable others for three reasons. First, the effect may be amplified by a self-serving bias, which applies especially to entrepreneurs (Baron, 1998; Parker, 2009). Individuals affected by a self-serving bias attribute positive outcomes to their own skills or talents, while attributing losses to external factors beyond their control. Thus, positive gains may boost their selfconfidence in making the right decision, whereas losses may have no such effect. Accordingly, the effect of prior gains may be particularly strong among entrepreneurs because, to a greater extent, they may tend to associate the gains with personal ability, resulting in overconfidence and an increased tendency to choose risk over certainty.

Second, the effect of prior gains may be amplified by the illusion of control, a cognitive bias that persuades an individual that he or she can control or influence a situation even if it is exogenously given or due to a random process (Langer, 1975; Taylor & Brown, 1988). Individuals may be affected by the illusion that they can predict the outcome of a random event. Among such individuals, a gain from a prior gamble may amplify an existing illusion of control. Since entrepreneurs have been shown to suffer from this cognitive bias (De Carolis et al., 2009; Keh et al., 2002; Simon et al., 2000), it is likely that amplification of the effect of prior grains is stronger for entrepreneurs than for comparable others. For this reason, the gains from a prior gamble may have a stronger effect on entrepreneurs than on comparable others.

⁷ While prospect theory is well-suited to explain risk-seeking behavior as a result of a biased cognitive assessment of gains, decreasing absolute risk aversion is well-suited to describe increased risk-seeking as a result of an adaptation to accumulated wealth. The cognitive bias approach may, hence, be somewhat more instrumental in disentangling individuals' specific effects, while the DARA seems to be a more global mechanism that holds for all. To separate the two effects, we focus our analysis on the effects of prior gains, distinguishing between the effects on those with entrepreneurial intentions and those without, while controlling for accumulated wealth.

Third, entrepreneurs have been shown to learn and adapt extensively from experience (Shepherd et al., 2015). In risk-taking, individuals respond to prior outcomes by avoiding alternatives that have produced poor past outcomes (Dendrell, 2007). This adaptation influences individuals' risk propensity. Since entrepreneurs are particularly prone to adapt extensively from experience, they are also more likely to be sensitive in their choices and, hence, more likely to make use of adaptive techniques as a response to risk⁸ (Patel & Fiet, 2009).

These arguments suggest that individuals with entrepreneurial intentions positively moderate the effect of prior gains on the choice of a risky gamble.

HYPOTHESIS 1: Individuals with entrepreneurial intentions will be more likely to choose risky options after a monetary gain than individuals without entrepreneurial intentions

The "risk propensity" effect

Prospect theory tells us that different presentations of the same monetary opportunity result in different choice behaviors. As one possible psychological explanation, Kahneman and Tversky (1979) introduced the isolation effect, which causes individuals to disregard components that are common to all of the prospects being considered. In a set of gambles with the same expected value and same probabilities for risky outcomes, we posit that individuals may disregard shared components and respond differently to the increasing degree of risk (i.e. the increasing variance in the size of monetary gains). Indeed, variance in risky outcomes is one indicator of monetary risk: an objective assessment of the risk inherent in a situation (Fox et al., 2015; Weber & Milliman, 1997).

⁸ Patel and Fiet (2009) investigated adaptive techniques as a response to risk, but did not compare individuals with entrepreneurial intentions and individuals without.

Entrepreneurship research indicates that entrepreneurs, despite not possessing a greater ex ante propensity for risk, react differently than managers (e.g. bankers) when coping with increasing risk. Entrepreneurs tend to accept risk as a given and focus on their personal commitment on achieving success, thus seeing opportunities where others see risk (Sarasvathy et al., 1998). However, Forlani and Mullins (2000) argued that entrepreneurs may be cautious about the increased variability in gains, suggesting that while entrepreneurs believe in their ability to manage lower-variability scenarios, they are less certain of this ability at higher levels of variability. However, this ability-based argument assumes that the individual plays an active role and may not be directly transferable to purely experimental settings, in which the scenarios are given and the subjects play an exogenous role (except in the choice between risk and certainty).

We argue that entrepreneurs generally value and experience utility from gaining when engaging in activities characterized by risk. The experienced utility is comparable to the utility gained from an individual winning (or not) in a game. This is one potential reason entrepreneurs value even the smallest gains and, hence, are willing to exert great efforts to secure gains. The utility entrepreneurs gain from potential gains makes them more likely to choose risk than comparable others. Yet, this effect is only applicable at lower levels of risk. As risk increases, entrepreneurs experience a shift in focus away from the potential utility of gains and instead focus on the risk involved. Accordingly, as risk increases, entrepreneurs portray an increasing likelihood to choose risk to the same degree as comparable others. This suggests that the propensity of entrepreneurs to choose risk may not be significantly different from that of nonentrepreneurs at high degrees of risk.

31

HYPOTHESIS 2: The propensity of individuals with entrepreneurial intentions to choose risk, compared to individuals without entrepreneurial intentions, is negatively moderated by the degree of risk.

DATA AND METHOD

Sample

To investigate our hypotheses, we identified a sample of individuals who were willing to participate in a lab experiment. Our sample was drawn from students enrolled in a general business economics undergraduate program at a major business school. We chose to use students for several reasons. First, students exhibit representative heterogeneity on many important characteristics (e.g. big five personality traits, family background) that make them suitable for generalizing findings to the greater populations (in our case, entrepreneurs and nonentrepreneurs). Second, we used students in order to minimize the impact of entrepreneurial experience on inter-group differences in risk-taking. This choice at least partially circumvents differences in risk propensities attributable to variations in prior or current entrepreneurial encounters. Third, we specifically used students in general business economics so that we could compare a group with entrepreneurial intentions with a group without such intentions. In particular, subjects were sampled at the beginning of their second year of study and were enrolled in the same study track, making them generally comparable. This high level of comparability allows us to rule out many alternative explanations attributable to unobserved factors.

At the end of their first semester, all general business economics students were offered the chance to select into a specialized program focused on entrepreneurship topics. The students
had to apply to join the program by submitting their motivation for wanting to study entrepreneurship. Of approximately 600 students, 49 applied to the program, and all were accepted, suggesting that selection was based strictly on student preference. Almost all of the applications suggested that the students' motivation for selecting into the program was a desire to establish their own business. Thus, we consider those students who enrolled in the entrepreneurship program to have entrepreneurial intentions and those who did not enroll in the program to not have entrepreneurial intentions (since they actively chose not to specialize in entrepreneurship and to continue instead with their general business economics studies). These two groups define the populations from which we generalize our results: novice entrepreneurs (students with entrepreneurial intentions) and non-entrepreneurs (students without entrepreneurial intentions).

We sent separate e-mails to the two populations to secure subjects for the experiment. The students were able to sign up for the experiment through a website designed to register them and obtain initial details. In total, 18 students from the entrepreneurship population and 27 students from the control population signed up, yielding a total of 45 individuals for the investigation. The sampled students represented approximately 40 percent (i.e. 18 of 45) of the total population of entrepreneurship students and approximately 5 percent (27 out of 550) of the population of non-entrepreneurship students. In order to achieve meaningful comparisons, we checked whether the samples were representative of their respective populations in terms of general characteristics (age and gender) and found that they generally reflected the demographics of their respective populations.

A preliminary analysis of the sampled subjects revealed that 66.7 percent (12) of the students with entrepreneurial intentions intended to start a firm within the next three years, while only 14.8 percent (4) of the group of students without entrepreneurial intentions had similar

ambitions. This provides some support for our sampling method, suggesting that entrepreneurial intentions were greater among the entrepreneurship students than among the general business economics students⁹.

The experiment was accompanied by a questionnaire designed to capture individual demographics, psychological profiles, cognitive biases, and economic circumstances. These dimensions were used to ensure that the samples were comparable on dimensions that might otherwise cause spurious correlations and, hence, potential biases in the estimates.

Table 1 provides descriptive statistics for the two groups across categorical variables drawn from the questionnaire data. It also provides test statistics on differences. We found very few differences between the two groups. We used a t-test to check for differences in ages and found that the average age for both groups was 21.5 years old. The t-statistics were insignificant at 0.460. Overall, we believe that these findings suggest that the samples are comparable and justify our categorization of the groups as representing students with and without entrepreneurial intentions.

Insert Table 1 about here

Real money games

Prior studies have used a multitude of research designs to investigate how entrepreneurs differ from non-entrepreneurs in their propensity to make choices characterized by risk. However, two of these research designs are most common in testing what differentiates

 $^{^{9}}$ As a robustness check, we exclude from the analysis subjects with mixed entrepreneurial intentions and find that the results hold (see Column 3, Table 8).

entrepreneurs with regard to risk and uncertainty: questionnaires (see e.g. Brockhaus, 1980; Busenitz & Barney, 1997) and money games (see e.g. Cramer et al., 2002; Weber & Milliman, 1997). Questionnaire-based research mainly uses sets of scenarios and case-based examples to represent real-life decisions in order to understand respondents' decision-making rationales. By contrast, money games tend to be simple structures (e.g., gambles) that resemble investment opportunities and seek to understand respondents' behaviors in tasks involving stakes. Both incentivized (Koudstaal et al., 2015; Weber & Milliman, 1997) and non-incentivized experiments have been utilized (Forlani & Mullins, 2000; Keh et al., 2002). In this study, we used a real money games experiment. In real money games, subjects are confronted with repeated binary choices between two different gambles. The games involve real money, and at the end of the experiment, the subjects receive an amount of money corresponding to their winnings based on their choices.

Real money games are particularly useful for capturing risk. First, according to *decision affect theory* (Mellers et al., 1997), individuals are expected to choose monetary opportunities with greater subjective emotional value. The presence of real monetary incentives in money games increases the emotional value attached to participants' decisions, sharpening both the research measurements and the comparisons of the decisions related to risk (Mellers et al., 1999; Rottenstreich & Hsee, 2001). Second, greater monetary incentives improve subjects' effort and focus during the experiment (Cramer et al., 2002; Gneezy & Rustichini, 2000). Third, real money games are the best way to measure the "prior gains effect," as indicated by Thaler and Johnson (1990) and Post et al. (2008).

Experiment design

The 45 subjects participated in two separate experiment sessions conducted in a computer laboratory in October 2014 according to group (i.e. those with entrepreneurial intentions [n=18] and those without entrepreneurial intentions [n=27]). Subjects were invited to participate using an online system and were not informed about the purpose of the study with respect to either the group comparisons (with and without entrepreneurial intentions) or the risk-taking behavior. The experiment, which was programmed on Z-Tree (Fischbacher, 2007), followed a between-subjects design that sought to explore the behavioral differences in real money game behaviors between the two groups with different entrepreneurial intentions. Upon arrival, each subject was randomly assigned a computer. No personal belongings, including mobile phones, calculators, and pens, were allowed in the room. Communication was strictly forbidden during the experiment. At the end of the experiment, all subjects were paid a flat cash fee of Danish Kroner (Dkr) 100 (approx. \$17.20) for their participation in addition to any winnings from the monetary gains. On average, each session lasted 90 minutes.

The experiment was divided into three parts: an introduction, a money games section, and a questionnaire section. The timeline of the experiment is presented in Table 2.

Insert Table 2 about here

The questionnaire section included questions on investment scenarios, demographics, psychological traits (e.g., Big Five; John et al., 2008) and cognitive bias (e.g., overconfidence;

Fischhoff et al., 1977). At the end of these three parts, participants were paid individually and dismissed.

In the first part, the experimenter briefly explained the structure of the experiment and presented the rules. In particular, the experimenter explained that during the money games, each decision would be rewarded monetarily based on the subjects' choices. The experimenter further stressed that communication was not allowed throughout the entire experiment. In the second part, the subjects were presented with money games. Following well-established prior research on risk propensity measurements in the lab (Fox et al., 2015), two exogenous manipulations were used in a new context to test differences between subjects with and without entrepreneurial intentions. The two manipulations-namely, feedback on the immediate prior monetary gain ("prior gain effect") and an increasing degree of risk ("risk propensity effect")—were adapted here following the definitions of Thaler and Johnson (1990) and Fox et al. (2015), respectively¹⁰. Each subject was faced with 24 unique binary decisions, which were presented sequentially in the same order for everyone. The experiment used four different gambling combinations: a certain win versus a risky gamble, a certain win versus an uncertain¹¹ gamble, a risky gamble versus a risky gamble, and a risky gamble versus an uncertain gamble. In this paper, we consider only one combination type (a certain win versus a risky gamble), producing a total of six unique decisions for individuals. Of these six decisions, two are not included in the current analysis. Instead, we used these as manipulation checks, testing the prior gain effect manipulation by giving no feedback on the size of the monetary winning¹². Thus, a total of 180 observations (four choices each from 45 individuals) were available for the investigation.

¹⁰ A large body of prior literature, particularly in neuroeconomics, has used money games to study individual propensities under different combinations of risk, uncertainty and ambiguity (see e.g. Hsu et al., 2005; Huettel et al., 2006).

¹¹ In uncertain gambles, we define uncertainty in the Knightian (1921) sense, as the lack of probability of obtaining possible monetary outcomes.¹² As expected, when no feedback was given on monetary outcomes, the two groups did not significantly differ in

their propensities to choose risk.

During the money games, the subjects were presented with a choice between a certain gain and a risky choice, which triggers a 50 percent chance of a greater gain and a 50 percent chance of a smaller gain than the certain option. The four different gambles are presented in Table 3.

Insert Table 3 about here

Throughout the decision rounds, we kept the gamble probabilities and expected values constant (50% and Dkr 14, respectively). As the difference in outcomes for the risk scenario (Option B) increased, the degree of risk for the decision round also increased. This approach follows the standards for capturing risk prescribed by Rothschild and Stiglitz (1970) and, more recently, Fox et al. (2015). An individual subject's payoff was attributable to random draws and did not reflect individual abilities.

The total payoff across all gambles was Dkr 15,036, which was distributed among the 45 participants at the end of the experiment. The total payoffs across the four considered gambles were Dkr3,204, an average of approximately Dkr71 per subject. However, the difference between the two groups evened out across the considered gambles. Individuals with entrepreneurial intentions received an average payoff of approximately Dkr73, which was not significantly different from the corresponding number for individuals without entrepreneurial intentions (approximately Dkr70). The payoffs of individuals with and without entrepreneurial intentions are graphically illustrated in Figures 1a and 1b, respectively.

Insert Figures 1a and 1b about here

Main variables

The main variable of interest was a dummy variable indicating whether, in each gamble, an individual chose risk over certainty. The risky choice applied to 40 percent of the gambles. Individuals with entrepreneurial intentions chose risk in approximately 47 percent (34 out of 72 decisions) of the gambles, while individuals without entrepreneurial intentions chose risk in approximately 35 percent (38 out of 108). These numbers were not significantly different according to a Chi-square test (see Table 4).

Insert Table 4 about here

To test the prior gains effect, we used a dummy variable measuring whether a subject had received a "prior gain" in the previous gamble. In other words, this dummy indicated whether, in the previous gamble, an individual experienced a monetary gain greater than the certain option. Table 3 presents the descriptive statistics for such prior gains. Individuals with entrepreneurial intentions experienced greater gains than expected about 42 percent (33 out of 72) of the time, while the corresponding number for individuals without entrepreneurial intentions was only 10 percent (11 out of 108). Therefore, individuals with entrepreneurial intentions were "luckier" than the other group in their immediately preceding gambles.

Finally, to test hypothesis 2, we created a variable to measure the level of risk the subjects faced in a given choice. We followed the convention prescribed by risk–value models (in this case, mean–variance; see e.g. Jia et al., 1999), taking the difference between the upper and lower potential gain of a given gamble. Accordingly, the risk associated with decision 1 in Table 2 is 12, and the risk associated with decision 4 is 28. These values represent strictly exogenous givens dictated by the gamble design.

Controls

The identified groups were not exogenously given. The subjects self-selected into either an entrepreneurship or a standard business economics study program, which then defined the studied groups. Accordingly, we sought to control for confounding factors that might have caused bias in the results. First, we controlled for personality traits, which have been shown to be associated with entrepreneurship and economic choices (John et al., 1991, 2008; Zhao & Seibert, 2006). Certain personality traits characterize a group's composition and are useful for controlling for the unobserved time-invariant factors that drive choice. In particular, we relied on the so-called big five personality traits, controlling for extraversion, openness to experience, neuroticism, conscientiousness, and agreeableness. To capture these traits, we followed the method used by John et al. (1991, 2008). The results of the factor analysis are presented in Appendix Table A1.

Overconfidence is a cognitive bias believed to be associated with choosing more risky investment opportunities (Barber & Odean 2001; Fischhoff et al., 1978), and it is particularly common among entrepreneurs (Busenitz & Barney, 1997; Koellinger et al., 2007; Koudstaal et

al., 2015)¹³. We controlled for this particular cognitive bias using the method proposed by Fischhoff et al. (1977). In particular, we operationalized overconfidence by comparing individuals' confidence in answering a series of two-choice questions about health statistics (based on 2010 data from the World Health Organization). We expected that individuals scoring highly for overconfidence would have a higher propensity for risk.

The investigation measured subjects' gambling behaviors. However, gambling behaviors can be pathological (see e.g. Stinchfield, 2000; Winters et al., 1998). We controlled for potential pathological gambling bias by including Stinchfield's (2000) pathological gambling index, which measures the degree to which an individual is a gambler, as a control in the analysis. Specifically, we asked five questions about the subjects' frequency of engaging in activities that included gambling (e.g. "gambling in Casinos," "buying lottery tickets," etc.). We expected to find a positive relationship between pathological gambling and a propensity for risk.

Finally, we controlled for a series of demographic variables that have proven to be significant for entrepreneurship or that are associated with specific behaviors related to risk. We controlled for age because aging is associated with changes in personal wealth, attitudes towards independence, and work effort, all of which affect entrepreneurship (Levesque & Minniti, 2006; Levesque et al., 2002) and risk aversion (Holt & Laury, 2002). We also controlled for gender, since we expected women to be more risk averse than men (Croson & Gneezy, 2009) and since gender differences exist in rates of entrepreneurial entry (Langowitz & Minniti, 2007). To measure gender, we included a dummy for whether a subject was female. We also controlled for whether an individual was an international student, defined as being a student with a nationality other than Danish. Finally, we controlled for parental entrepreneurship in at least one of a

¹³ We chose not to use incentivized measures of overconfidence, since a recent study found that overconfidence may not differ between entrepreneurs and top managers (Koudstaal et al., 2015).

subject's parents and for the subjects' general income, using a dummy for whether a subject earned more than DKr 50,000 a year (suggesting a non-transitory job).

Method

The experiment was designed to predict the binary choices of the two groups given their subjection to various exogenous stochastic shocks. For this reason, a logistic regression technique was applied to explain the binary choices of the individuals based on the group to which they belonged (i.e. individuals with or without entrepreneurial intentions) and the effects of the exogenously inflicted prior gains and degrees of risk.

We have shown that the two groups were comparable across multiple dimensions. We have also included control variables to make the results more robust and convincing. However, since the two identified groups were self-selected and not randomly assigned, there may be some unobservable factors that could cause bias in the estimates. There may also be some sampling conditions that play a role in generating the results. We consider these potential sources of bias in a supplement to the analysis.

RESULTS

Table 5 presents the descriptive statistics and correlation coefficients for all the considered variables. The subjects received a prior gain in 22 percent of the gambles. Since the subjects chose risk in 40 percent of the gambles, this reflects an approximately 50% chance of winning. None of the correlations in Table 5 are great enough to raise concerns about potential multicollinearity.

Insert Table 5 about here

Table 6 presents the results of the regression analysis. The first column includes only the control variables and the dummy for whether a given subject had entrepreneurial intentions or not. This regression confirms the univariate test, suggesting that subjects with and without entrepreneurial intentions are unlikely to differ in terms of risk aversion. The marginal effect is 0.160 and not statistically significant. The second column reports the results including only the entrepreneurial intentions dummy and the variables used to operationalize the main effects of the prior gains effect and the risk propensity effect (i.e. the degree of risk). Column 3 introduces the interactions between the entrepreneurial intentions dummy and the two explanatory variables. Since these are included sequentially, the entrepreneurial intentions dummy becomes significant, indicating that individuals with entrepreneurial intentions are generally more likely to choose risk over certainty (i.e. they are less risk averse). Yet, this result only appears when coupled with other conditions, which may be one of the reasons prior research on entrepreneurs and their tendency to choose risk have not produced cohesive findings. In other words, entrepreneurs' choices regarding risk seem to be conditional. The results in Column 3 show an insignificant main effect with regard to degree of risk (estimate = 0.009).

The prior gains effect is insignificant when considering the main effect, suggesting that the effect did not hold for the sample of individuals without entrepreneurial intentions. Since the marginal effects of the interaction between entrepreneurial intention and prior gain also prove insignificant, there is no support for hypothesis 1: *Individuals with entrepreneurial intentions will be more likely to choose risky options after a monetary gain than individuals without* *entrepreneurial intentions*. These findings are consistent with those in column 4, for which we also included controls.

In terms of the risk propensity effect, the results suggest a general tendency among the subjects to shy away from risk as the risk increases. The estimate for degree of risk is significant at a 1% level in column 4, exhibiting a marginal effect of -0.04. We also find a significant negative marginal effect associated with the interaction between risk and entrepreneurial intention. The marginal effect associated with the interaction between degree of risk and entrepreneurial intention is significant at a 1% level at -0.044, providing support for hypothesis 2: *The propensity of individuals with entrepreneurial intentions to choose risk, compared to the propensity of individuals without entrepreneurial intentions, is negatively moderated by the degree of risk.*

Insert Table 6 about here

We investigated the hypotheses further by considering in more detail the marginal effects of the prior gains and the risk propensity effect based on the results shown in Table 6, column 4. The marginal effects with respect to prior gains are depicted in Figure 2a, and the contrasted marginal effects are shown in Figure 2b. These figures reveal new insights that suggest that the prior gains value (compared to a lack of a prior gains value) has some marginal significance at just below the 5% level. There is, in fact, a difference in the marginal effects when considering prior gains outcomes, though this difference is weak and at the very limit of what can be considered significant.

The corresponding figures for degree of risk are shown in Figures 3a and 3b. Here, it is evident that, at low values of risk, subjects with entrepreneurial intentions exhibit significantly higher propensities for risk than those without. This difference disappears at higher levels of risk. Figure 3a shows that the reason is that subjects with entrepreneurial intent start out with higher propensities for risk, but 'catch up' in terms of risk aversion as the degree of risk increases. This is consistent with the theoretical argumentation and lends further support to Hypothesis 2.

Insert Figures 2a and 2b about here

Insert Figures 3a and 3b about here

In terms of the control variables, the results were mostly consistent with the predicted associations. The women in our sample exhibited lower propensities to choose risk than males, and international students seemed to be more likely to choose risk than native students. In addition, we found that accumulated earnings had a strong effect, suggesting that those who have accumulated wealth during the game are also more likely to choose risky options. The findings concerning the pathological aggregate were odd: In our sample, subjects with gambling habits tended to choose risk less often than those without.

Supplementary analysis

To further investigate the results (or lack thereof), three supplementary studies were conducted. First, we explored the panel structure of the data. Second, we considered the lack of results on prior pay by considering the amount a subject gained in the prior gamble (rather than a dummy variable). Finally, we limited the sample used to those subjects most likely to fit our theoretical profile of entrepreneurial intent.

Panel data analysis

The results may be biased due to our use of a panel dataset in which the same subjects are considered repeatedly. There may be unobserved elements that could cause bias in the estimates. To investigate this, we ran both fixed effects and random effects models and used a Hausman test to select between the two. The Hausman test is depicted in Table 7. Only the variables that varied within individual subjects were considered, since the fixed effects model could not estimate coefficients for the remaining variables. Fortunately, these variables are also the one considered most prominently in this study. Table 7 clearly shows that the coefficients of the fixed effects and random effects specifications provide similar estimates. Indeed, the test statistics indicate that these coefficients are not statistically significantly different. Accordingly, we choose to consider the random effects model, since this model appears to be both consistent and efficient.

Insert Table 7 about here

The results of the random effects model are shown in column 1 of Table 8. Overall, these results confirm the prior results, although with no significance for the main effect of the degree of risk. The random effects specification allowed us to look at the within-group correlations to assess the nature of the consistency of the choices made by each subject. The rho of the random

effect specification was only 0.12, indicating a relatively low correlation between the subjects' choices after controlling for the observables. Table 9 reports several intra-class statistics for propensity to choose risk over certainty at the 1^{st} , 25^{th} , 50^{th} , 75^{th} , and 99^{th} percentiles. At the median propensity, the marginal probability of choosing risk was 0.348, while the corresponding number for the joint probability of any two given choices was 0.122. These numbers varied greatly across percentiles, with a 0.000 joint probability at the 1^{st} percentile of observed propensity and a 0.912 joint probability at the 99th percentile. At all percentiles, these numbers suggested that an individual was 1.010 times more likely to choose risk if they had chosen risk in the previous round than an individual who had not chosen risk in the previous round.

Insert Table 8 about here Insert Table 9 about here

Prior pay instead of prior gain

To further explore the weak results of the marginal effects on the prior gain effect, we reconfigured the positive feedback variable to be continuous, measuring the magnitude of the prior gain (prior pay) (rather than a dummy indicating simply whether prior gain was greater

than expected gain). The results are reported in column 2 of Table 8. The marginal effects and contrasted marginal effects are shown in Figures 3a and 3b, respectively.

Insert Figures 3a and Figure 3b about here

The estimate of prior pay did not prove significant in the main effect. There was, however, a weak significance in the interaction term. Figures 3a and 3b provide a more detailed and comprehensive indication of how prior pay correlates with the propensity to choose risk over certainty when comparing individuals with and without entrepreneurial intentions. The marginal effects were significantly higher for individuals with entrepreneurial intentions than for individuals without entrepreneurial intentions when considering the higher values of prior pay. There was no significant difference for the lower values. This might suggest that the amplification effect of the illusion of control and the self-serving bias on the house money effect is particularly strong at higher values of prior gains and insignificant at lower values. Since winnings of DKr 14 and below did not exceed the expected gain, the theory did not predict a significant contrasted marginal effect for these values. However, we also observe insignificant contrasted marginal effect for these values. However, we also observe insignificant to predict a suggests that the significance is perhaps more likely to stem from a change in behavior among those without entrepreneurial intentions than among those with entrepreneurial intentions, since it is in these margins that the changes are more pronounced.

Restricted sample

We also considered the sample of individuals included in the analysis. We found that some of the subjects with entrepreneurial intentions did not plan on starting a firm within the next three years, while some without entrepreneurial intentions were planning to do so. We ran the analysis without including these subjects to investigate the sensitivity of the results to this subsample. These results are depicted in column 3 of Table 8. Overall, the results were unchanged, and the estimates were not substantially altered. Due to a upward shift in the estimate, however, the significance of the interaction between prior pay and entrepreneurial intention becomes more significant, suggesting that the weaker results with regard to this hypothesis may be caused by the sampling.

CONCLUSION

In this paper, we addressed the following research question: How do individuals with and without entrepreneurial intentions differ in their propensity to choose risk? Moving beyond the notion that entrepreneurs have higher risk-taking propensities than comparable others, recent empirical studies have considered the circumstantial nature of the differences in risk-taking behaviors between entrepreneurs and non-entrepreneurs. Such studies seek to address the conundrum of entrepreneurs' high level of self-assessed risk-taking propensity and contradictory empirical evidence. Further, these studies challenge the view of the entrepreneur as an individual who is more willing and better able to cope with the risks associated with starting and running a business. To address this issue empirically, it is necessary to understand the circumstances under which entrepreneurs show different propensities for risk. However, endogeneity is a concern in studies exploring behavior because choices are often influenced by unobservable elements. To overcome this limitation, several recent studies have used data from experimental designs.

We have reported the results of a quasi-laboratory experiment with a number of distinguishing features. First, the results were based on repeated individual decisions with real monetary incentives. This is a special feature of our experiment, since each choice per individual involved real monetary gain. The work of Thaler and Johnson (1990) and Post et al. (2008), among others, suggests that this is the best way to compare individual sensitivity to past experience. Second, we compared individuals with entrepreneurial intentions—a significant predictor of entrepreneurship (Krueger et al., 2000)—with a well-defined group of individuals without entrepreneurial intentions within the same study program. The two groups were very similar in terms of demographics and experience, but clearly differed in terms of their career focus. Third, we collected a variety of additional background characteristics, both psychological and attitudinal, which allowed us to control for a variety of otherwise unobservable characteristics affecting decision-making, such as cognitive biases. Finally, we included both a subjective, survey-based measure of risk attitude and an incentivized, choice-based measure of risk-related preferences. This allowed us to compare subjective perceptions of risk attitude with objective measures based on actual choices with real financial consequences.

In line with previous studies, the individuals with entrepreneurial intentions in our sample were not significantly different from individuals without entrepreneurial intentions in their choices related only to risk (i.e. their level of risk aversion). However, when confronted with two specific exogenous manipulations grounded in cognitive psychology and prospect theory (the prior gains effect and the risk propensity effect, respectively), individuals with entrepreneurial intentions were found to differ significantly from the comparable group of individuals without entrepreneurial intentions. After controlling for a number of alternative

explanations, our results show that these differences arose from the distinctive ways in which the two groups coped with monetary risk. Specifically, the individuals with entrepreneurial intentions were influenced by the prior gains effect, since their propensities to choose risk were higher following the receipt of a monetary gain than the propensities of individuals without entrepreneurial intentions. Moreover, the gap between individuals with and without entrepreneurial intentions in terms of risk aversion closed at higher levels of risk, since the focus of the individuals with entrepreneurial intentions shifted from the potential utility of small gains to the risk involved. This latter result supports the work of both Forlani and Mullins (2000) and Kahneman and Lovallo (1993). Hence, our results provide further specific evidence regarding the circumstantial nature of entrepreneurs' propensities to choose risky monetary opportunities.

Limitations and implications

Our paper has several limitations related to the generalizability of the results and external validity. We chose to restrict the sample to students self-selecting into two distinct programs. While this has advantages in terms of the comparability of the considered groups, it also reduces generalizability. Therefore, the effects explored in this study should be tested across a wider range of subjects and possibly a group already engaged in entrepreneurial activity. Furthermore, while our definition of entrepreneurs follows standard conventions (see e.g. Krueger et al., 2000; Liñan & Chen, 2009; Roach & Sauermann, 2015), we acknowledge its limitations relating to its reliance on entrepreneurial intentions to empirically operationalize entrepreneurship. To limit alternative explanations with respect to sample construction, we used several post-experiment checks and found no reasons for concern. First, we checked whether subjects (both with and without entrepreneurial intentions) were in the same study concentration before entering the

entrepreneurship concentration. Coupling this check with the subjects' stated preferences (i.e. their written applications to the entrepreneurship concentration) suggested that socialization was not a relevant factor. Second, we recruited subjects without specifying the purpose of our study with respect to group comparisons or risk-taking. Third, we checked whether the topic of entrepreneurial risk-taking had been presented in recent classroom work and found no support for such a confounding factor. Finally, we enforced a strict no-communication policy during the laboratory experiment, thereby limiting any possible socialization effect during the experiment. Furthermore, our experiment offers a precise, yet narrowly focused study of the cognitive mechanisms guiding individuals' observed behaviors. This represents a limitation of our study. Real-life entrepreneurial settings involve a number of different dynamic factors that concurrently influence individuals' risk choices. In particular, knowledge and expertise (Heath & Tversky, 1991) may lead individuals to take risks in areas about which they feel confident or passionate (Cardon Wincent, Singh, & Drnovsek, 2009). Future research may test how prior knowledge and passion influence entrepreneurs' risk-taking by reducing (or increasing) the prior gains effect and the risk propensity effect.

This study contributes to our understanding of the mechanisms guiding entrepreneurial decision-making (i.e. the prior gains effect and the risk propensity effect). This has two direct implications for entrepreneurship research. First, entrepreneurs place significant value on monetary gains. However, in real life, entrepreneurs often face an environment that does not provide reliable information or feedback on monetary investments. This might be why entrepreneurs rely on heuristics when making decisions in real-life cases. It might also offer some insights into why successful entrepreneurs abandon entrepreneurial ventures in order to start new businesses. Serial entrepreneurship is not exclusive to necessity entrepreneurs; it also often characterizes some of the most successful entrepreneurs. Prior gains may lead

52

entrepreneurs to choose more risky career steps and embolden them to embark on new entrepreneurial ventures. This would suggest that this type of serial entrepreneurship is not strictly the result of lower capital barriers, but could also be explained by the individual behavioral traits of entrepreneurs.

Second, entrepreneurs respond to changes in the degree of risk associated with monetary opportunities. This moderating effect is unexpected and tempers the natural tendency of entrepreneurs to choose risky monetary opportunities. In real entrepreneurial ventures, monetary opportunities have the characteristics of investments, and it is often not possible to fully understand the increasing degrees of risk involved. Our laboratory experiment setup allowed us to isolate this effect, illustrating a case in which entrepreneurs and non-entrepreneurs behaved similarly and exhibited similar propensities. This perspective is helpful for resolving the difference between the theory and empirical evidence related to the risk propensities of entrepreneurs and non-entrepreneurs. Our study suggests that entrepreneurs may exhibit consistent behaviors across different degrees of risk, while non-entrepreneurs may change their behaviors at higher levels of risk.

These results have implications for entrepreneurs' stakeholders. Specifically, they offer insights that will help investors, business partners, employees, and other stakeholders better understand and manage the entrepreneurs with whom they do business. For one thing, the results provide evidence that there are certain circumstances in which entrepreneurs are more likely to make risky choices (e.g. immediately after an economic gain). In addition, the results suggest that entrepreneurs may pursue risky choices even when the possibility of an economic gain is low (i.e. cases of low risk). This information may be useful in helping stakeholders manage their entrepreneurial partners.

APPENDIX

Insert Table A1 about here

REFERENCES

- Andersen, S., Hanspal, T., & Nielsen, K. M. (2014). Once bitten, twice shy: Do personal experiences or wealth changes affect risk taking? (Working paper, Copenhagen Business School).
- Barber, B. M. & Odean, T. (2001). Boys will be boys: Gender, overconfidence, and common stock investment. *Quarterly Journal of Economics*, 116(1) 261–292.
- Baron, R. A. (1998). Cognitive mechanisms in entrepreneurship: Why and when enterpreneurs think differently than other people. *Journal of Business Venturing*, *13*(4), 275–294.
- Battalio, R. C., Kagel, J. H., & Jiranyakul, K. (1990). Testing between alternative models of choice under uncertainty: Some initial results. *Journal of Risk and Uncertainty*, 3(1), 25–50.
- Begley, T. M. & Boyd, D. P. (1988). Psychological characteristics associated with performence in entrepreneurial firms and smaller businesses. *Journal of Business Venturing*, 2(1), 79–93.
- Brockhaus, R. H. (1980). Risk taking propensity of entrepreneurs. *Academy of Management Journal*, 23(3), 509–520.
- Busenitz, L. W. & Barney, J. B. (1997). Differences between entrepreneurs and managers in large organizations: Biases and heuristics in strategic decision-making. *Journal of Business Venturing*, 12(1), 9–30.
- Cardon, M. S., Wincent, J., Singh, J., & Drnovsek, M. (2009). The nature and experience of entrepreneurial passion. *Academy of Management Review*, *34*(3), 511–532.
- Carland III, J. W., Carland Jr., J. W., Carland, J. A. C., & Pearce, J. W. (1995). Risk taking propensity among entrepreneurs, small business owners and managers. *Journal of Business and Entrepreneurship*, 7(1), 15-26.
- Cassar, G. & Craig, J. (2009). An investigation of hindsight bias in nascent venture activity. *Journal of Business Venturing*, 24(2), 149–164.

- Chen, C. C., Greene, P. G., & Crick, A. (1998). Does entrepreneurial self-efficacy distinguish entrepreneurs from managers? *Journal of Business Venturing*, *13*(4), 295–316.
- Cramer, J. S., Hartog, J., Jonker, N., & Van Praag, C. M. (2002). Low risk aversion encourages the choice for entrepreneurship: An empirical test of a truism. *Journal of Economic Behavior & Organization*, 48(1), 29–36.
- Croson, R. & Gneezy, U. (2009). Gender differences in preferences. *Journal of Economic Literature*, 47(2) 448–474.
- De Carolis, D. M., Litzky, B. E., & Eddleston, K. A. (2009). Why networks enhance the progress of new venture creation: The influence of social capital and cognition. *Entrepreneurship Theory and Practice*, 33(2), 527–545.
- Deligonul, Z. S., Hult, G. T. M., & Cavusgil, S. T. (2008). Entrepreneuring as a puzzle: An attempt to its explanation with truncation of subjective probability distribution of prospects. *Strategic Entrepreneurship Journal*, 2(2), 155–167.
- Denrell, J. (2007). Adaptive learning and risk taking. Psychological Review, 114(1), 177-187.
- Ellsberg, D. (1961). Risk, ambiguity, and the savage axioms. *The Quarterly Journal of Economics*, 75(4) 643–669.
- Fern, M. J., Cardinal, L. B., & O'Neill, H. M. (2012). The genesis of strategy in new ventures: Escaping the constraints of founder and team knowledge. *Strategic Management Journal*, *33*(4), 427–447.
- Fischbacher, U. (2007). z-Tree: Zurich toolbox for ready-made economic experiments. *Experimental Economics*, *10*(2), 171–178.
- Fischhoff, B., Slovic, P., & Lichtenstein, S. (1977). Knowing with certainty: The appropriateness of extreme confidence. *Journal of Experimental Psychology: Human Perception and Performance*, 3(4), 552-564.
- Fischhoff, B., Slovic, P., Lichtenstein, S., Read, S., & Combs, B. (1978). How safe is safe enough? A psychometric study of attitudes towards technological risks and benefits. *Policy Sciences*, 9(2), 127–152.
- Forbes, D. P. (2005). Are some entrepreneurs more overconfident than others? *Journal of Business Venturing*, 20(5), 623–640.
- Forlani, D. & Mullins, J. W. (2000). Perceived risks and choices in entrepreneurs' new venture decisions. *Journal of Business Venturing*, 15(4), 305–322.
- Fox, C. R., Erner, C., & Walters, D. J. (2015). Field to the laboratory and back. In *The Wiley Blackwell* handbook of judgment and decision making (pp. 43-88).
- Gneezy, U. & Rustichini, A. (2000). Pay enough or don't pay at all. *Quarterly Journal of Economics*, 115(3) 791–810.
- Heath, C. & Tversky, A. (1991). Preference and belief: Ambiguity and competence in choice under uncertainty. *Journal of Risk and Uncertainty*, 4(1), 5–28.

- Heath, C., Larrick, R. P., & Wu, G. (1999). Goals as reference points. *Cognitive Psychology*, 38(1), 79–109.
- Holm, H. J., Opper, S., & Nee, V. (2013). Entrepreneurs under uncertainty: An economic experiment in China. *Management Science*, 59(7), 1671–1687.
- Holt, C. A. & Laury, S. K. (2002). Risk aversion and incentive effects. American Economic Review, 92(5), 1644–1655.
- Hsu, M., Bhatt, M., Adolphs, R., Tranel, D., & Camerer, C. F. (2005). Neural systems responding to degrees of uncertainty in human decision-making. *Science*, *310*(5754), 1680–1683.
- Huettel, S. A., Stowe, C. J., Gordon, E. M., Warner, B. T., & Platt, M. L. (2006). Neural signatures of economic preferences for risk and ambiguity. *Neuron*, 49(5), 765–775.
- Jia, J., Dyer, J. S., & Butler, J. C. (1999). Measures of perceived risk. *Management Science*, 45(4), 519–532.
- John, O. P., Donahue, E. M., Kentle, R. L. (1991). *The big five inventory—Versions 4a and 54*. Berkeley, CA: University of California, Berkeley, Institute of Personality and Social Research.
- John, O. P., Naumann, L. P., & Soto, C. J. (2008). Paradigm shift to the integrative big five trait taxonomy. *Handbook of Personality: Theory and Research*, *3*, 114–158.
- Kachelmeier, S. J. & Shehata, M. (1992). Examining risk preferences under high monetary incentives: Experimental evidence from the People's Republic of China. *The American Economic Review*, 82(5) 1120–1141.
- Kahneman, D. & Lovallo, D. (1993). Timid choices and bold forecasts: A cognitive perspective on risk taking. *Management Science*, *39*(1), 17–31.
- Kahneman, D. & Tversky, A. (1972). Subjective probability: A judgment of representativeness. *Cognitive Psychology*, 3(3), 430–454.
- Kahneman, D. & Tversky, A. (1979). Prospect theory: An analysis of decision under risk. *Econometrica: Journal of the Econometric Society*, 47(2) 263–291.
- Keh, H. T., Foo, M. D., & Lim, B. C. (2002). Opportunity evaluation under risky conditions: The cognitive processes of entrepreneurs. *Entrepreneurship Theory and Practice*, 27(2), 125–148.
- Kihlstrom, R. E. & Laffont, J.-J. (1979). A general equilibrium entrepreneurial theory of firm formation based on risk aversion. *The Journal of Political Economy*, 87(4) 719–748.
- Knight, F. H. (1921). Risk, uncertainty and profit. New York, NY: Hart, Schaffner and Marx.
- Koellinger, P., Minniti, M., & Schade, C. (2007). "I think I can, I think I can": Overconfidence and entrepreneurial behavior. *Journal of Economic Psychology*, 28(4), 502–527.
- Koudstaal, M., Sloof, R., & Van Praag, M. (2015). Risk, uncertainty, and entrepreneurship: Evidence from a lab-in-the-field experiment. *Management Science*, 62(10), 2897–2915.

- Koudstaal, M., Sloof, R., & van Praag, M. (2015). Are entrepreneurs more optimistic and overconfident than managers and employees? (Tinbergen Institute Discussion Papers 15-124/VII). Tinbergen Institute.
- Kőszegi, B. & Rabin, M. (2006). A model of reference-dependent preferences. *The Quarterly Journal of Economics*, 121(4), 1133–1165.
- Krueger, N. F., Reilly, M. D., & Carsrud, A. L. (2000). Competing models of entrepreneurial intentions. *Journal of Business Venturing*, 15(5), 411–432.
- Langer, E. J. (1975). The illusion of control. *Journal of Personality and Social Psychology*, *32*(2), 311-328.
- Langowitz, N. & Minniti, M. (2007). The entrepreneurial propensity of women. *Entrepreneurship Theory and Practice*, *31*(3), 341–364.
- Levesque, M. & Minniti, M. (2006). The effect of aging on entrepreneurial behavior. *Journal of Business Venturing*, 21(2), 177–194.
- Liñán, F. & Chen, Y. W. (2009). Development and cross-cultural application of a specific instrument to measure entrepreneurial intentions. *Entrepreneurship Theory and Practice*, *33*(3), 593–617.
- March, J. G. (1994). *Primer on decision making: How decisions happen*. Simon and Schuster, New York, NY.
- Mellers, B. A., Schwartz, A., Ho, K., & Ritov, I. (1997). Decision affect theory: Emotional reactions to the outcomes of risky options. *Psychological Science*, 8(6) 423–429.
- Mellers, B., Schwartz, A., & Ritov, I. (1999). Emotion-based choice. *Journal of Experimental Psychology: General*, 128(3), 332-345.
- Moskowitz, T. J., & Vissing-Jørgensen, A. (2002). The returns to entrepreneurial investment: A private equity premium puzzle? *The American Economic Review*, 92(4), 745-778. Mullins, J. W. & Forlani, D. (2005). Missing the boat or sinking the boat: A study of new venture decision making. *Journal of Business Venturing*, 20(1), 47–69.
- Novemsky, N. & Dhar, R. (2005). Goal fulfillment and goal targets in sequential choice. *Journal of Consumer Research*, 32(3), 396–404.
- Oskarsson, A. T., Van Boven, L., McClelland, G. H., & Hastie, R. (2009). What's next? Judging sequences of binary events. *Psychological Bulletin*, 135(2), 262-285.
- Parker, S. C. (2006). Learning about the unknown: How fast do entrepreneurs adjust their beliefs? *Journal of Business Venturing*, 21(1), 1–26.
- Parker, S. C. (2009). Can cognitive biases explain venture team homophily? *Strategic Entrepreneurship Journal*, *3*(1), 67–83.
- Patel, P. C., & Fiet, J. O. (2009). Systematic search and its relationship to firm founding. *Entrepreneurship Theory and Practice*, *33*(2), 501–526.

- Post, T., Van den Assem, M. J., Baltussen, G., & Thaler, R. H. (2008). Deal or no deal? Decision making under risk in a large-payoff game show. *The American Economic Review*, 98(1), 38–71.
- Riley Jr., W. B. & Chow, K. V. (1992). Asset allocation and individual risk aversion. *Financial Analysts Journal*, 48(6), 32–37.
- Roach, M. & Sauermann, H. (2015). Founders or joiners? The role of preferences and context in shaping different entrepreneurial interests. *Management Science*, *61*(9), 2160–2184.
- Rothschild, M. & Stiglitz, J. E. (1970). Increasing risk: I. A definition. *Journal of Economic Theory*, 2(3), 225–243.
- Rottenstreich, Y. & Hsee, C. K. (2001). Money, kisses, and electric shocks: On the affective psychology of risk. *Psychological Science*, *12*(3), 185–190.
- Sarasvathy, D., Simon, H. A., & Lave, L. (1998) Perceiving and managing business risks: Differences between entrepreneurs and bankers. *Journal of Economic Behavior & Organization*, 33(2), 207– 225.
- Schade, C., Kunreuther, H., & Koellinger, P. (2012). Protecting against low-probability disasters: The role of worry. *Journal of Behavioral Decision Making*, 25(5), 534–543.
- Shadish, W. R., Cook, T. D., & Campbell, D. T. (2002). *Experimental and quasi-experimental designs* for generalized causal inference. Houghton, Mifflin and Company.
- Shane, S. & Venkataraman, S. (2000). The promise of entrepreneurship as a field of research. *Academy* of Management Review, 25(1), 217–226.
- Shepherd, D. A., Williams, T. A., & Patzelt, H. (2015). Thinking about entrepreneurial decision making review and research agenda. *Journal of Management*, *41*(1), 11–46.
- Simon, M., Houghton, S. M., & Aquino, K. (2000). Cognitive biases, risk perception, and venture formation: How individuals decide to start companies. *Journal of Business Venturing*, 15(2), 113–134.
- Stewart, W. H., Watson, W. E., Carland, J. C., & Carland, J. W. (1999). A proclivity for entrepreneurship: A comparison of entrepreneurs, small business owners, and corporate managers. *Journal of Business Venturing*, 14(2), 189–214.
- Stewart, N., Chater, N., Stott, H. P., & Reimers, S. (2003). Prospect relativity: How choice options influence decision under risk. *Journal of Experimental Psychology: General*, 132(1), 23-46.
- Stinchfield, R. (2000). Gambling and correlates of gambling among Minnesota public school students. *Journal of Gambling Studies*, *16*(2–3), 153–173.
- Stuart, R. W. & Abetti, P. A. (1990). Impact of entrepreneurial and management experience on early performance. *Journal of Business Venturing*, 5(3), 151–162.
- Taylor, S. E. & Brown, J. D. (1988). Illusion and well-being: A social psychological perspective on mental health. *Psychological Bulletin*, *103*(2), 193-210.

- Thaler, R. H. & Johnson, E. J. (1990). Gambling with the house money and trying to break even: The effects of prior outcomes on risky choice. *Management Science*, *36*(6), 643–660.
- Weber, E. U. & Johnson, E. J. (2009). Mindful judgment and decision making. Annual Review of Psychology, 60, 53–85.
- Weber, E. U. & Milliman, R. A. (1997). Perceived risk attitudes: Relating risk perception to risky choice. *Management Science*, 43(2), 123–144.
- Winters, K. C., Bengston, P., Door, D., & Stinchfield, R. (1998). Prevalence and risk factors of problem gambling among college students. *Psychology of Addictive Behaviors*, 12(2), 127-135.
- Wu, B. & Knott, A. M. (2006). Entrepreneurial risk and market entry. *Management Science*, 52(9), 1315–1330.
- Zhao, H. & Seibert, S. E. (2006). The big five personality dimensions and entrepreneurial status: a metaanalytical review. *Journal of Applied Psychology*, 91(2), 259-271.

TABLES

TABLE 1: Descriptive statistics and test of differences between individuals with and without entrepreneurial intentions (N=45)

	With entrepreneurial intention	Without entrepreneurial intention	χ^2	Sig.
Male	14 (31 1)	21 (46.7)		
Female	4(8.9)	6 (13.3)	0.000	
Danish	16(35.6)	26(57.8)		
International	2(4.4)	1(2.2)	0.952	
50.000 < income < 300.000	15(33.3)	24 (53.3)		
0 < income < 50.000	3 (6.7)	3 (6.7)	0.313	
Parent entrepreneur	5(11.1)	7(15.6)		
No parent entrepreneur	13(28.9)	$(20 \ (44.4)$	0.019	
University degree	3(6.7)	1(2.2)		
No University degree	15(33.3)	26 (57.8)	2.588	
Part-time employed	11(24.4)	23 (51.1)		
Not part-time employed	7 (15.6)	4 (8.9)	3.389	*
Established firm	3(6.7)	1(2.2)		
Not established firm	15(33.3)	26 (57.8)	2.241	
Establish firm within 3 years	12(26.7)	4 (8.9)		
Not establish firm within 3 years	6 (13.3)	23(51.1)	12.672	***

* p < 0.10, ** p < 0.05, *** p < 0.01, percentages of table total in parentheses

TABLE 2: Timeline of experiment: Tasks and compensation schemes

		Between-Subjects Design					
		With entrepreneurial intentions (N=18)	Without entrepreneurial intentions (N=27)				
	Part 1: Introduction (10 minutes)	Individuals are assigned to a computer and rules are explained. No communic allowed					
Tasks	Part 2: Money Games (15 minutes)	e gamble choices, presented one-by-one in a gambles have no monetary feedback. Four risk, certainty-uncertainty, risk-risk, risk- as an expected value of Dkr 14.					
	Part 3: Questionnaire (35 minutes)	Individuals answer questions on: demographics, risk attitudes, investme s) opportunities, cognitive biases (representativeness, overconfidence), psycho traits (Big Five, Gambling attitudes)					

Note: at the end of part 3, subjects were paid according to the rules (fixed show-up fee, and performance during money games) and dismissed.

TABLE 3: Considered four real monetary games between certainty (Option A) and risk (OptionB)

	Opt	ion A	Option B		
Decision 1	100%	Dkr14	$50\% \\ 50\%$	Dkr8 Dkr20	
Decision 2	100%	Dkr14	$50\% \\ 50\%$	Dkr4 Dkr24	
Decision 3	100%	Dkr14	$50\% \\ 50\%$	Dkr0 Dkr28	
Decision 4	100%	Dkr14	$50\% \\ 50\%$	Dkr-2 Dkr30	

TABLE 4: Descriptive statistics on key variables (N=180)

	With entrepreneurial intention	Without entrepreneurial intention	Total	χ^2	Sig.
Choose risk (option B) Choose certainty (option A)	$\begin{array}{c} 34 (18.9) \\ 38 (21.1) \end{array}$	$\begin{array}{c} 38 \ (21.1) \\ 70 \ (38.9) \end{array}$	$\begin{array}{c} 72 \ (40.0) \\ 108 \ (60.0) \end{array}$	2.608	
Prior gain greater than expected Prior gain not greater than expected Total	$\begin{array}{c} 33 \ (18.3) \\ 39 \ (21.7) \\ 72 \ (40.0) \end{array}$	$ \begin{array}{c} 11 \ (6.1) \\ 97 \ (53.9) \\ 108 \ (60 \ 0) \end{array} $	$\begin{array}{c} 44 \ (24.4) \\ 136 \ (75.6) \end{array}$	29.724	***

* p < 0.10, ** p < 0.05, *** p < 0.01, percentages of table total in parentheses

TABLE 5: Descriptive statistics and Pearson's correlation coefficients

		Mean	S.D.	Min	Max	[1]	[2]	[3]	[4]	[5]	[6]	[7]	[8]
[1]	XX7·41 4 · · · · 4 · ·	0.40	0.40	0.00	1.00								
[1]	With entrepreneurial intention	0.40	0.49	0.00	1.00	0.00							
[2]	Prior gain	0.22	0.42	10.00	1.00	0.06	0.07						
[3]	Degree of risk	20.07	8.70	12.00	48.00	0.00	0.07	0.00					
[4]	Accumulated winning	198.32	(5.51	08.00	388.00	0.07	0.39	0.90	0.10				
[5]	Extraversion	0.00	1.00	-3.07	1.46	0.13	0.12	0.00	0.13	0.00			
[6]	Openness	0.00	1.00	-2.14	2.52	0.16	0.01	0.00	-0.03	0.00			
[7]	Neuroticism	0.00	1.00	-1.90	1.86	-0.12	0.01	0.00	-0.07	0.00	0.00		
[8]	Conscientiousness	0.00	1.00	-2.87	1.60	0.35	0.06	0.00	-0.08	0.00	0.00	0.00	
[9]	Agreebleness	0.00	1.00	-1.98	1.80	-0.20	-0.14	0.00	-0.13	0.00	0.00	0.00	0.00
[10]	Overconfidence	101.25	21.72	67.00	148.00	-0.25	-0.11	0.00	-0.09	0.03	0.22	0.11	0.11
[11]	Risk neutral	0.27	0.44	0.00	1.00	0.02	-0.02	0.00	0.00	-0.03	0.05	-0.24	-0.06
[12]	Risk prone	0.20	0.40	0.00	1.00	-0.07	0.01	0.00	0.11	0.15	-0.19	-0.05	0.00
[13]	Pathological aggregate	1.52	0.56	1.00	3.20	-0.07	-0.02	0.00	0.00	0.07	-0.22	-0.27	-0.12
[14]	Age	21.51	0.78	20.00	23.00	-0.07	0.09	0.00	0.04	0.02	-0.07	0.06	-0.10
[15]	Female	0.22	0.42	0.00	1.00	0.00	0.04	0.00	0.04	0.35	0.30	0.39	-0.05
[16]	International	0.07	0.25	0.00	1.00	0.15	0.00	0.00	-0.05	0.16	-0.02	0.04	0.18
[17]	Parent entrepreneur	0.27	0.44	0.00	1.00	0.02	-0.02	0.00	-0.03	0.04	-0.01	0.12	0.04
[18]	University degree	0.09	0.28	0.00	1.00	0.22	0.14	0.00	0.05	-0.05	-0.02	0.16	-0.07
[19]	Establish firm within 3 years	0.36	0.48	0.00	1.00	0.53	0.23	-0.00	0.03	-0.01	0.38	-0.18	0.23
[20]	Income above 50.000	0.87	0.34	0.00	1.00	-0.08	-0.04	0.00	-0.04	0.13	-0.19	-0.09	0.03
		[9]	[10]	[11]	[12]	[13]	[14]	[15]	[16]	[17]	[18]	[19]	
		[0]	[10]	[11]	[12]	[10]	[11]	[10]	[10]	[*1]	[10]	[10]	
[10]	Overconfidence	0.18											
[11]	Risk neutral	-0.15	-0.32										
[12]	Risk prone	-0.20	-0.07	-0.3 0									
[13]	Pathological aggregate	-0.12	0.20	0.25	0.14								
[14]	Age	0.00	0.09	0.06	0.24	0.33							
[15]	Female	0.02	0.10	-0.20	0.00	-0.40	0.06						
[16]	International	-0.16	-0.03	0.24	-0.13	-0.06	-0.29	0.29					
[17]	Parent entrepreneur	-0.04	-0.09	0.09	-0.05	-0.14	-0.27	0.04	0.04				
[18]	University degree	0.00	-0.16	-0.19	-0.16	-0.23	0.00	0.02	-0.08	-0.19			
[19]	Establish firm within 3 years	-0.03	-0.04	0.08	-0.02	-0.09	0.05	-0.06	-0.01	-0.03	0.26		
[20]	Income above 50.000	0.08	0.04	-0.35	0.20	0.17	0.01	-0.26	-0.42	0.09	-0.11	0.02	

Note: Correlation coefficients above 0.06 are significant at a 5% level

	Standard Logit Models					
	(1)	(2)	(3)	(4)		
Explanatory Variables						
Entrepreneurial intention	0.160	0.144	0.680^{***}	* 0.800***		
	[0.123]	[0.097]	[0.156]	[0.133]		
Prior Gain		-0.063	-0.044	-0.169		
		[0.082]	[0.161]	[0.130]		
Degree of risk		0.002	0.009	-0.040***		
		[0.005]	[0.007]	[0.014]		
Entrepreneurial intention \times Prior gain			0.321	0.364		
Entropyon quick interation & Domas of risk			[0.205]	[0.242]		
Entrepreneurial intention \times Degree of risk			-0.055	-0.044		
Control variables			[0.012]	[0.017]		
Accumulated winning	0.001			0.006***		
Treedintatated withing	[0.001]			[0.002]		
Extraversion	0.018			0.027		
	[0.035]			[0.037]		
Openness	0.071			0.113		
	[0.057]			[0.069]		
Neuroticism	-0.050			-0.009		
	[0.033]			[0.035]		
Conscientiousness	-0.061*			-0.042		
	[0.035]			[0.038]		
Agreebleness	-0.074			-0.046		
	[0.049]			[0.048]		
Overconfidence	0.005*			0.005*		
	[0.003]			[0.003]		
Bish many risk attitude	0.169			0.164		
Risk prone	0.102			[0.104]		
Risk neutral	[0.143] 0.409**			0.142]		
TUSK neutral	[0 162]			[0.195]		
Risk averse	benchmark			benchmark		
Pathological Aggregate	-0.360***			-0.420**		
0 00 0	[0.131]			[0.165]		
Age	0.059			0.088		
	[0.063]			[0.069]		
Female	-0.309***			-0.463***		
	[0.118]			[0.153]		
International	0.394^{**}			0.699^{***}		
	[0.193]			[0.174]		
Parent entrepreneur	-0.046			-0.065		
TT 1	[0.076]			[0.084]		
University degree	0.094			0.049		
Establish from within 2 years	[0.111]			[0.132]		
Establish mini within 5 years	-0.071 [0 138]			-0.090		
Income above 50 000	0.130]			0.140		
meonie above 50.000	[0.142]			[0.151]		
	[0.1 12]			[0.101]		
Number of observations	180	180	180	180		
Log-likelihood	-103.144 -	119.621-	116.363	-92.856		
χ^2	89.430***	2.250	11.599^{**}	164.040^{***}		
$Pseudo - R^2$	0.149	0.013	0.040	0.234		

TABLE 6: Explaining subjects choice of risk compared to certainty; results of logit regressions

* p < 0.10, ** p < 0.05, *** p < 0.01 at a two sided test, Clustered standard errors in parentheses

TABLE 7: Hausman test for differences between a random and a fixed effects model

specification

	Coef	ficients				
	Fixed Random Difference in					
	Effects	Effects	Coefficients	error		
Prior gain	-0.966	-0 786	-0.180	0.462		
Entrepreneurial intention \times Prior gain	1.641	1.575	0.065	0.790		
Degree of risk	-0.251	-0.174	-0.077	0.074		
Entrepreneurial intention \times Degree of risk	-0.174	-0.191	0.017	0.026		
Accumulated Gains	0.035	0.027	0.009	0.009		

Hausman test for difference not systematic: $\chi^2(5)=0.020$ (Prob> $\chi^2=1.000$)

TABLE 8: Explaining subjects choice of risk compared to certainty; results when using continuous variable for prior gain (prior pay) and a restricted sample

	Random	Prior Pav	Restricted Sample
	(1)	(2)	(3)
Explanatory Variables			
Entrepreneurial intention	4.417^{**}	0.497^{***}	0.706^{***}
	[1.938]	[0.090]	[0.244]
Prior gain/pay	-0.786	-0.008	-0.015
	[0.985]	[0.005]	[0.011]
Degree of risk	-0.174	-0.03***	-0.025
	[0.113]	[0.008]	[0.016]
Entrepreneurial intention \times Prior gain/pay	1.575	0.009*	0.021**
	[1.344]	[0.005]	[0.010]
Entrepreneurial intention \times Degree of risk	-0.191**	-0.028**	-0.044*
	[0.089]	[0.014]	[0.024]
Control variables	0.007**	0.005***	0.00.4**
Accumulated winning	0.027^{++}	0.005	[0.004
E-turner in	[0.013]	[0.001]	[0.002]
Extraversion	[0.269]	0.025	0.075
Onenness	[0.308]	[0.028]	[0.034]
Openness	[0,474]	[0.041]	0.143
Nouvoticiam	[0.474]	[0.041]	[0.039]
Neuroticisii	-0.038	-0.005	[0.02
Consciontiousness	[0.293] 0.182	$\begin{bmatrix} 0.027 \end{bmatrix}$	0.004
Conscientiousness	-0.182 [0.326]	[0.020]	[0.038]
Agreebleness	[0.320]	$\begin{bmatrix} 0.029 \end{bmatrix}$	0.015
Agreebleness	[0.350]	[0.030]	[0.050]
Overconfidence	0.023	0.003	0.004
Overconfidence	[0.021]	[0.004	[0.004]
Holt & Laury risk attitude	[0:021]	[0.002]	[0.000]
Bisk prone	0.709	0.114	0.255^{*}
TOOL PLOTE	[0.889]	[0.089]	[0.139]
Risk neutral	1.582	0.268**	0.3
	[1.306]	[0.109]	[0.226]
Risk averse	benchmark	penchmark	benchmark
Pathological Aggregate	-1.811	-0.314***	-0.374**
0 00 0	[1.107]	[0.098]	[0.174]
Age	0.381	0.067	0.052
с С	[0.512]	[0.051]	[0.075]
Female	-1.999**	-0.343***	-0.433***
	[1.005]	[0.084]	[0.108]
International	3.023	0.505^{***}	0.265
	[2.033]	[0.134]	[0.217]
Parent entrepreneur	-0.28	-0.047	-0.05
	[0.652]	[0.057]	[0.077]
University degree	0.21	0.032	0.09
	[1.750]	[0.092]	[0.118]
Establish firm within 3 years	-0.415	-0.061	
	[1.095]	[0.099]	
Income above 50.000	0.627	0.095	0.032
	[1.455]	[0.104]	[0.143]
		100	1.40
Number of observations	02.055	180	140
Log-likelihood	-92.855	-92.765	-72.448
χ^{-}	•	189.250***	1013.810***
$F seudo = K^{-}$		0.234	0.216
LIN TEST (NE AS, IOZISTIC LEGLESSIOII)	0.000		

* p < 0.10, ** p < 0.05, *** p < 0.01 at a two sided test Clustered standard errors in parentheses

Measure	1st Percentile	25th Percentile	50th Percentile	75th Percentile	99th Percentile
Marginal prob.	0.012	0.180	0.348	0.611	0.955
Joint prob.	0.000	0.033	0.122	0.374	0.912
Odds ratio	1.010	1.010	1.010	1.010	1.010
Pearson's r	0.000	0.001	0.002	0.002	0.000
Yule's Q	0.005	0.005	0.005	0.005	0.005

TABLE 9: Measures of intra-class associations in random-effects logit

TABLE A1: Results of factor analysis generating Big 5 Personality Traits

I am someone who is talkative 0.7915 vertex to the functional knowledge to the set of the functional knowledge to the set of th	Question	Extraversion	Openness	Neroticism	Conscien- tiousness	Agreeable– ness	As expected
is talkative 0.7915 tends to find furths with others 0.4307 is depressed, blue -0.5399 is depressed, blue 0.4307 is reserved 0.6932 is reserved 0.6932 is reserved 0.6932 is reserved 0.5165 v can be somewhat careless 0.5165 v is helpful and unselfish with others 0.5064 is reliable worker 0.6015 v tarst quarrels with others 0.6498 v starts quarrels with others 0.6498 v ortice a lot of enthusiasm 0.8022 tends to be disorganized 0.7265 has an active imagination 0.4028 v tends to be disorganized 0.7265 has an active inagination 0.4028 v tends to be dupit 0.8135 is inventive 0.6131 v as service personality can be cold and aloof 0.5797 v generevers until the task is finished 0.7268 is sometimes shy, inhibited 0.7268 v sources and kind to almost everyone does things efficiently 0.6183 v tends to be haxy 0.6542 v values artistic, aesthetic experiences 0.4505 v sources and kind to almost everyone does things efficiently 0.6183 v prefers work that is routine 0.5409 v sources and kind to almost everyone does things efficiently 0.6183 v prefers work that is routine 0.5409 v sources and kind to almost everyone v does things efficiently 0.6486 v prefers work that is routine 0.5409 v sources and kind to almost everyone v does things efficiently 0.6486 v thes to cooperate with others 0.6456 v thes to cooperate with others 0.6470 v and keep base and follows through with them 0.6456 v thes to cooperate with others 0.6470 v and keep base and bind veryone 0.6456 v thes to cooperate with others 0	I am someone who						
tends to find faults with others 0.4307 v does a thorough job is depressed, blue -0.5399 is original, comes up with new ideas 0.4043 0.5208 v is helpful and unselfish with others 0.6932 0.5165 v can be somewhat careless 0.4063 0.6015 v is traited realess 0.7396 v is traited realess 0.6015 v starts quarrels with others 0.6015 v starts quarrels with others 0.6015 v starts quarrels with others 0.6015 v that a for energy 0.6498 0.6015 v starts quarrels with others 0.6384 v v starts quarrels with others 0.6022 0.5251 0.4328 v tends to be disorganized 0.7063 v worries a lot of enthusiasm 0.8022 0.5251 0.4328 v worries a lot to enthusiasm 0.8022 0.7063 v is generates a lot of enthusing 0.8022 0.7063 v worries a lot be disorganized 0.7265 v has an active imagination 0.4028 v tends to be disorganized 0.7265 v is senotionally stable, not easily upset 0.7654 v is is mentive inscribed 0.512 v has an assertive personality c can be cold and aloof 0.4512 v is sometimes shy, inhibited 0.7265 v v values artistic, aesthetic experiences 0.4505 v is sometimes shy, inhibited 0.7268 v is considerate and kind to almost everyone does things efficiently 0.6183 v values artistic, aesthetic experiences 0.4505 v is sometimes shy, inhibited 0.7978 v is sometimes shy with ideas 0.6463 v v makes plans and follow through with them 0.8054 v is sometimes shy with ideas 0.6456 v has few artistic interests likes to cooperate with others 0.4740 is easily distracted d	is talkative	0.7915					\checkmark
does a thorough job	tends to find faults with others					0.4307	, V
is depressed, blue -0.5399 is original, comes up with new ideas 0.4043 0.5208 $^{\rm v}$ is helpful and unselfish with others 0.6332 $^{\rm v}$ can be somewhat careless 0.4685 $^{\rm v}$ is relaxed, handles stress well 0.6498 0.6015 $^{\rm v}$ is carbou shout many different things 0.6015 $^{\rm v}$ is carbou shout many different things 0.6015 $^{\rm v}$ is a reliable worker 0.6384 $^{\rm v}$ starts quarrels with others 0.6498 0.6384 $^{\rm v}$ is a reliable worker 0.6384 $^{\rm v}$ can be tense 0.6384 $^{\rm v}$ generates a lot of enthusiasm 0.8022 0.5251 0.4328 $^{\rm v}$ tends to be disorganized 0.7265 $^{\rm v}$ is motionally stable, not easily upset 0.4028 $^{\rm v}$ tends to be lazy 0.6451 $^{\rm v}$ is is motionally stable, not easily upset 0.5654 $^{\rm v}$ is sometimes with the task is finished 0.4512 $^{\rm v}$ is sometimes shy, inhibited 0.7265 $^{\rm v}$ is sometimes theta is finished 0.5442 $^{\rm v}$ is sometimes with the task is finished 0.7265 $^{\rm v}$ is sometimes whethet is finished 0.5450 $^{\rm v}$ is sometimes whethet is finished 0.7265 $^{\rm v}$ is sometime 0.4512 $^{\rm v}$ is sometimes 0.6081 $^{\rm v}$ is sometimes 0.6081 $^{\rm v}$ is sometimes 0.6081 $^{\rm v}$ is sometimes 0.5442 $^{\rm v}$ is sometimes 0.6183 $^{\rm v}$ is sometimes 0.6183 $^{\rm v}$ is sometimes 0.5449 $^{\rm v}$ is considerate and kind to almoset everyone 0.5449 $^{\rm v}$ is sometimes 0.6463 $^{\rm v}$ is sometimes 10.5409 $^{\rm v}$ is sometimes 10.5409 $^{\rm v}$ is sometimes routine 0.5409 $^{\rm v}$ is sometimes routine 0.5409 $^{\rm v}$ is sometimes routine to others 0.6456 $^{\rm v}$ is sometimes thich exterests ilkes to cooperate with others 0.4740 is easily distracted 0.4740 is easily distracted 0.4740 is easily distracted 0.4400 is distracted 0.4400 is distracted 0.4400 is distracted 0.4400 is distracted 0.4400 is distracted 0.4400 is distracted 0.4400 i	does a thorough job						•
is original, comes up with new ideas 0.4043 0.5208 \vee is reserved 0.6932 \vee is helpful and unselfish with others 0.6932 \vee can be somewhat careless 0.7396 \vee is reliaxed, handles stress well 0.7396 \vee is curious about many different things 0.6015 \vee starts quarrels with others 0.6498 \vee starts quarrels with others 0.6498 \vee can be tense 0.6384 \vee generates a lot of enthusiasm 0.8022 \vee has a forgiving nature -0.5251 0.4328 vorries a lot 0.7265 \vee has an active imagination 0.4028 \vee tends to be quiet 0.8135 \vee is emetally trusting 0.7654 \vee tends to be lazy 0.5232 \vee is sometimes shutily personality 0.5232 \vee can be cold and alof 0.7654 \vee is considerate and kind to almost everyone 0.6463 \vee does things efficiently 0.6183 \vee values artistic, aesthetic experiences 0.46463 \vee values and to others 0.7978 \vee is sometimes rude to others 0.6463 \vee is sometimes rude to others 0.6456 \vee is considerate and kind to almost everyone 0.6463 \vee is sourging, sociable 0.7978 \vee is outgoing, sociable 0.7978 \vee is outgoing, sociable 0.6456	is depressed, blue	-0.5399					
is reserved 0.6932 0.5165 v is helpful and unselfish with others 0.5165 v can be somewhat careless 0.4685 0.4685 is relaxed, handles stress well 0.7396 0.4685 v is curious about many different things 0.6015 v is a reliable worker 0.6498 0.4676 v is a reliable worker 0.6384 v generates a lot of enthusiasm 0.8022 -0.5251 0.4328 v tends to be disorganized 0.7265 v has an active imagination 0.4028 v tends to be disorganized 0.7265 v has an astive imagination 0.4028 v tends to be quiet 0.8135 v is inventive 0.4512 v has an assertive personality 0.4512 v has an assertive personality 0.4512 v is sometimer shuthing 0.4028 v values artistic, aesthetic experiences 0.4505 v is sometimers shu, inhibited 0.7268 v is considerate and kind to almost everyone 0.4505 v does things efficiently 0.4505 v is sometimes shu, inhibited 0.7268 v is considerate and kind to almost everyone 0.4505 v does things efficiently 0.4505 v is sometimes work that is routine 0.5409 v is outgoing, sociable 0.7978 v makes plans and follows through with them 0.4088 v is sometimes rule to others 0.4740 v is to streak thetic therests 1.4500 v has few artistic interests 1.4500 v has few artis	is original, comes up with new ideas	0.4043	0.5208				\checkmark
is helpful and unselfish with others 0.5165 0.4685 0.4685 0.4685 0.4685 0.4685 0.4685 0.4685 0.4685 0.4685 0.4685 0.4685 0.4685 0.4685 0.4685 0.4685 0.4685 0.4685 0.595 0.4685 0.4685 0.595 0.4685 0.595 0.4676 0.595 0.4505 0.595 0.4505 0.595 0.4595 0.	is reserved	0.6932					
can be somewhat careless0.4685is relaxed, handles stress well0.7396Vis curious about many different things0.6015Vis full of energy0.6498Vstarts quarels with others0.4676Vis a reliable worker0.6384Vcan be tense0.6384Vgenerates a lot of enthusiasm0.8022Vhas a forgiving nature-0.52510.4328Vtends to be disorganized0.7063Vworries a lot0.7063Vhas an active imagination0.4028Vtends to be quiet0.8135Vis inventive0.7554Vhas an ascritive personality0.5232Vcan be cold and aloof0.7063Vvalues artistic, aesthetic experiences0.4505Vvalues artistic, aesthetic experiences0.4505Vvalues artistic, aesthetic to everyone0.5409Vdoes things efficiently0.5409Vvalues artistic, aesthetic to thers0.5409Vvalues artistic, aesthetic to thers0.5409Vvalues artistic, nerverse suitines0.6463Vvalues artistic, interests0.4740Vlikes to cooperate with others0.4740Vlikes to cooperate with others0.4740Vis sometimes rule to others0.4740Vis sometimes rule to others0.4740Vis sometimes rule to others0.4740Vis soure t	is helpful and unselfish with others					0.5165	\checkmark
is relaxed, handles stress well 0.7396 V is curious about many different things 0.6015 V starts quarrels with others 0.6498 0.4676 V starts quarrels with others 0.6498 0.4676 V is a reliable worker 0.5641 V generates a lot of enthusiasm 0.8022 V has a forgiving nature 0.5041 V generates a lot of enthusiasm 0.8022 V tends to be disorganized 0.7063 V worries a lot of enthusiasm 0.8022 V tends to be disorganized 0.7063 V is generally trusting 0.4028 V tends to be disorganized 0.5232 V tends to be quiet 0.8135 V is generally trusting 0.4028 V tends to be lazy 0.5232 V has an assertive personality 0.4512 V has an assertive personality 0.4512 V has an assertive personality 0.4512 V is inventive 0.4512 V has an assertive personality 0.5442 V values artistic, aesthetic experiences 0.4505 V is considerate and kind to almost everyone 0.5442 V values artistic, aesthetic experiences 0.4505 V is outgoing, sociable 0.7978 0.6183 V is outgoing, sociable 0.7978 0.6183 V is outgoing, sociable 0.7978 V is sometimes rule to others 0.4509 V has few artistic interests likes to reflect, play with ideas 0.6456 V has few artistic interests likes to cooperate with others 0.4740 is sometimes rule to others 0.	can be somewhat careless					0.4685	
is curious about many different things 0.6015 v is full of energy 0.6498 v starts quarrels with others 0.6498 v is a reliable worker 0.6384 v is ingenious, a deep thinker 0.5641 v generates a lot of enthusiasm 0.8022 v has a forgiving nature 0.5021 0.4328 v tends to be disorganized 0.7265 v has an active imagination 0.4028 v tends to be quist is generally trusting 0.8135 v is generally trusting 0.8135 v is inventive 0.5232 v has an assertive personality v can be cold and aloof 0.5797 v perseveres until the task is finished 0.5122 v has an assertive personality 0.5265 v is sometimes shy, inhibited 0.7268 v is considerate and kind to almost everyone 0.5402 v is sometimes shy, inhibited 0.7268 v is considerate and kind to almost everyone 0.5409 v is sometimes shy, inhibited 0.7978 v perfers work that is routine 0.5409 v is outgoing, sociable 0.7978 v gensentines rude to others 0.4740 is outgoing, sociable 0.7978 v makes plans and follows through with them 0.6456 v has few artistic interests likes to reflect, play with ideas 0.4740 is end source or 0.4740 is end source or 0.4740 is end source with others 0.4740 is end source with others 0.4740	is relaxed, handles stress well			0.7396			\checkmark
is full of energy starts quarrels with others 0.6498 0.4676 starts quarrels with others 0.6384 0.4676 is a reliable worker can be tense 0.6384 v generates a lot of enthusiasm 0.8022 -0.5251 0.4328 has a forgiving nature tends to be disorganized -0.5251 0.4328 v worrices a lot 0.7265 v v has an active imagination 0.4028 v v tends to be quiet 0.8135 v v is generally trusting tends to be lazy 0.5232 v tends to be lazy 0.5232 v is inventive 0.4512 v has an assertive personality can be cold and aloof tense shy, inhibited 0.7268 v values artistic, aesthetic experiences 0.4505 v values artistic, aesthetic experiences 0.4505 v is isometimes shy, inhibited 0.7268 v is sometimes shy, inhibited 0.7978 v is sometimes shy inhibited 0.7978 v is sometimes rule to others 0.6463 v is sometimes rule to others 0.6456 v is sometimes rule to others 0.6456 v ikes to reflect, play with ideas 0.6456 v has few artistic interests 0.4740 v	is curious about many different things		0.6015				
starts quarrels with others0.4676vis a reliable worker0.6384vcan be tense0.6384vgenerates a lot of enthusiasm0.8022vhas a forgiving nature0.50210.4328vworries a lot0.7063vvworries a lot0.7063vvhas an active imagination0.4028vvtends to be quiet0.8135vvis generally trusting0.5654vvtends to be lazy0.5654vvis inventive0.4512vvhas an assertive personality0.5122vvcan be cold and aloof0.5040vvperseveres until the task is finished0.6081vvvalues artistic, aesthetic experiences0.4505vvis sometimes shy, inhibited0.7268vvis sometimes rude to others0.6463vvperfers work that is routine0.5978vvis sometimes rude to others0.6463vvis sometimes rude to others0.6466vvis sometimes rude to others0.6456vvhas few artistic interests0.4740is easily distractedvis easily distracted0.4740is easily distractedv	is full of energy	0.6498					
is a reliable worker can be tense can deep thinker can be cold attemption can be disorganized can be cold and aloof can be cold and aloof can be moody can be moo	starts quarrels with others					0.4676	, V
can be tense 0.6384 vis ingenious, a deep thinker 0.5641 vgenerates a lot of enthusiasm 0.8022 vhas a forgiving nature -0.5251 0.4328 vtends to be disorganized 0.7063 vworries a lot 0.7265 vhas an active imagination 0.4028 vtends to be quiet 0.8135 vis generally trusting 0.5232 vtends to be lazy 0.5232 vis emotionally stable, not easily upset 0.4512 vhas an assertive personality 0.5797 vcan be cold and aloof 0.5797 vvalues artistic, aesthetic experiences 0.4505 vis sometimes shy, inhibited 0.7268 vis sometimes rule to almost everyone 0.6463 vdoes thing efficiently 0.6463 vprefers work that is routine 0.5409 vis sometimes rule to others 0.7978 vis sometimes rule to others 0.6456 vhas few artistic interests 0.4740 vlikes to reflect, play with ideas 0.6456 vhas few artistic interests 0.4740 vlikes to cooperate with others 0.4740 vis easily distracted 0.4740 v	is a reliable worker						
is ingenious, a deep thinker 0.5641 vgenerates a lot of enthusiasm 0.8022 -0.5251 0.4328 vhas a forgiving nature 0.7063 0.7063 vworries a lot 0.7265 0.7265 vhas an active imagination 0.4028 vtends to be quiet 0.8135 v is generally trusting 0.5232 v tends to be hazy 0.7654 v is generally stable, not easily upset 0.7654 v has an assertive personality 0.5232 v can be cold and aloof 0.6081 v can be moody 0.5442 v values artistic, asthetic experiences 0.4505 v is sometimes shy, inhibited 0.7268 v is outgoing, sociable 0.7978 v is outgoing, sociable 0.7978 v is sometimes rude to others 0.6463 v upset plans and follows through with them 0.6456 v has few artistic interests 0.4740 v likes to reflect, play with ideas 0.4740 v	can be tense			0.6384			\checkmark
generates a lot of enthusiasm0.8022Vhas a forgiving nature-0.52510.4328Vtends to be disorganized0.7063Vworries a lot0.7265Vhas an active imagination0.4028Vtends to be quiet0.8135Vis generally trusting0.7654Vtends to be lazy0.7654Vis emotionally stable, not easily upset0.7654Vhas an assertive personality0.4512Vcan be cold and aloof0.5797Vperseveres until the task is finished0.6081Vcan be mody0.5442Vvalues artistic, aesthetic experiences0.4505Vis considerate and kind to almost everyone0.5403Vdoes things efficiently0.6183Vremains calm in tense situations0.6463Vis outgoing, sociable0.7978Vis sometimes rude to others0.7978Vmakes plans and follows through with them0.8054Vgets nervous easily0.4498Vlikes to cooperate with others0.4740Vis easily distracted0.4740V	is ingenious, a deep thinker		0.5641				· V
has a forgiving nature -0.5251 0.4328 $\sqrt{1000}$ tends to be disorganized 0.7063 $\sqrt{1000}$ worries a lot 0.7265 $\sqrt{1000}$ has an active imagination 0.4028 $\sqrt{1000}$ tends to be quiet 0.8135 $\sqrt{1000}$ is generally trusting 0.5232 $\sqrt{1000}$ tends to be lazy 0.7654 $\sqrt{1000}$ is inventive 0.7654 $\sqrt{1000}$ has an assertive personality 0.7654 $\sqrt{1000}$ can be cold and aloof 0.5797 $\sqrt{1000}$ can be cold and aloof 0.5797 $\sqrt{1000}$ values artistic, aesthetic experiences 0.4505 $\sqrt{1000}$ is sometimes shy, inhibited 0.7268 $\sqrt{1000}$ is outgoing, sociable 0.7978 $\sqrt{1000}$ is sometimes rude to others 0.6463 $\sqrt{1000}$ is sometimes rude to others 0.6463 $\sqrt{1000}$ ikes to reflect, play with ideas 0.6456 $\sqrt{1000}$ has few artistic interests 0.4740 $\sqrt{1000}$ ikes to cooperate with others 0.4740 $\sqrt{1000}$ is easily distracted 0.4740 $\sqrt{1000}$	generates a lot of enthusiasm	0.8022					
tends to be disorganized0.7063worries a lot0.7265has an active imagination0.4028tends to be quiet0.8135is generally trusting0.5232tends to be lazy0.7654tends to be lazy0.7654is inventive0.4512has an assertive personality0.5797can be cold and aloof0.5797perseveres until the task is finished0.6081can be moody0.5442values artistic, aesthetic experiences0.4505is sometimes shy, inhibited0.7268is outgoing, sociable0.7978is outgoing, sociable0.7978is outgoing, sociable0.7978is outgoing, sociable0.7978is outgoing, sociable0.7978is outgoing, sociable0.6456has few artistic interests0.44740likes to coperate with others0.4740is easily distracted0.4740	has a forgiving nature				-0.5251	0.4328	, V
worries a lot0.7265has an active imagination0.4028tends to be quiet0.8135is generally trusting0.5232tends to be lazy0.5232is emotionally stable, not easily upset0.7654is inventive0.4512has an assertive personality0.6081can be cold and aloof0.5797perseveres until the task is finished0.6081can be cold and aloof0.5442values artistic, aesthetic experiences0.4505is considerate and kind to almost everyone0.6463does things efficiently0.6463perfers work that is routine0.7978is sometimes rude to others0.7978is sometimes rude to others0.7978is sometimes rude to others0.7978is outgoing, sociable0.7978is onterimes rude to others0.6456has few artistic interests0.4740likes to coperate with others0.4740is easily distracted0.4740	tends to be disorganized				0.7063		, V
has an active imagination0.4028tends to be quiet0.8135is generally trusting0.5232tends to be lazy0.5232tends to be lazy0.7654is inventive0.4512has an assertive personality0.6081can be cold and aloof0.6081perseveres until the task is finished0.6081can be moody0.5442values artistic, aesthetic experiences0.4505is sometimes sky, inhibited0.7268is considerate and kind to almost everyone0.6463does things efficiently0.6183vremains calm in tense situations0.6463prefers work that is routine0.7978is sometimes rule to others0.7978is sometimes and follows through with them0.8054gets nervous easily0.4546has few artistic interests0.4740likes to coperate with others0.4740is easily distracted0.4740	worries a lot			0.7265			, V
tends to be quiet0.8135is generally trusting0.8135tends to be lazy0.5232tends to be lazy0.7654is emotionally stable, not easily upset0.7654is inventive0.4512has an assertive personality0.4512can be cold and aloof0.5797perseveres until the task is finished0.6081can be mody0.5442values artistic, aesthetic experiences0.4505is sometimes shy, inhibited0.7268is considerate and kind to almost everyone0.6463does things efficiently0.6183remains calm in tense situations0.6463prefers work that is routine0.7978is sometimes rule to others0.7978is sometimes rule to others0.7978usts plans and follows through with them0.4898gets nervous easily0.4898likes to reflect, play with ideas0.4740is easily distracted0.4740	has an active imagination		0.4028				v
is generally trusting tends to be lazy 0.5232 v is emotionally stable, not easily upset 0.7654 v is inventive 0.4512 v has an assertive personality can be cold and aloof 0.5797 v perseveres until the task is finished 0.6081 v can be moody 0.5442 v values artistic, aesthetic experiences 0.4505 v is sometimes shy, inhibited 0.7268 v does things efficiently 0.7268 v is considerate and kind to almost everyone 0.5876 v does things efficiently 0.6183 v remains calm in tense situations 0.6463 v prefers work that is routine 0.5409 v is outgoing, sociable 0.7978 v is sometimes rude to others 0.7978 v gets nervous easily 0.4898 v likes to reflect, play with ideas 0.4740 is easily distracted v is easily dist	tends to be quiet	0.8135					, v
tends to be lazy0.5232vis emotionally stable, not easily upset0.7654vis inventive0.4512vhas an assertive personality0.4512vcan be cold and aloof0.6081vperseveres until the task is finished0.6081vcan be moody0.5442vvalues artistic, aesthetic experiences0.4505vis sometimes shy, inhibited0.7268vis considerate and kind to almost everyone0.6463vdoes things efficiently0.6463vprefers work that is routine0.5409vis outgoing, sociable0.7978vis sometimes rude to others0.7978vgets nervous easily0.6456vlikes to reflect, play with ideas0.6456vhas few artistic interests0.4740vlikes to cooperate with others0.4740is easily distracted0.4740	is generally trusting						•
is emotionally stable, not easily upset0.7654is inventive0.4512has an assertive personality0.4512can be cold and aloof0.5797perseveres until the task is finished0.6081can be moody0.5442values artistic, aesthetic experiences0.4505is sometimes shy, inhibited0.7268is considerate and kind to almost everyone0.66183does things efficiently0.6463prefers work that is routine0.5409is sometimes rude to others0.7978is sometimes rude to others0.7978gets nervous easily0.6456has few artistic interests0.4740likes to cooperate with others0.4740is easily distracted0.4740	tends to be lazy				0.5232		
is inventive0.4512has an assertive personality can be cold and aloof0.5797can be cold and aloof0.5797perseveres until the task is finished0.6081can be moody0.5442values artistic, aesthetic experiences0.4505is sometimes shy, inhibited0.7268is considerate and kind to almost everyone0.6183does things efficiently0.6463prefers work that is routine0.7978is sometimes rude to others0.7978is sometimes rude to others0.7978gets nervous easily0.4898likes to reflect, play with ideas0.6456has few artistic interests0.4740likes to cooperate with others0.4740is easily distracted0.4740	is emotionally stable, not easily upset			0.7654			, V
has an assertive personality can be cold and aloof0.5797perseveres until the task is finished0.6081can be moody0.5442values artistic, aesthetic experiences0.4505is sometimes shy, inhibited0.7268is considerate and kind to almost everyone0.6463does things efficiently0.6183vremains calm in tense situations0.5409is sometimes rude to others0.7978is sometimes rude to others0.7978makes plans and follows through with them0.6456likes to reflect, play with ideas0.6456has few artistic interests0.4740likes to cooperate with others0.4740	is inventive		0.4512				, V
can be cold and aloof0.5797vperseveres until the task is finished0.6081vcan be moody0.5442vvalues artistic, aesthetic experiences0.4505vis sometimes shy, inhibited0.7268vis considerate and kind to almost everyone0.6183vdoes things efficiently0.6183vremains calm in tense situations0.6463vprefers work that is routine0.5409vis sometimes rude to others0.7978vis sometimes rude to others0.7978vgets nervous easily0.6456vlikes to reflect, play with ideas0.6456vlikes to cooperate with others0.4740vis easily distracted0.4740v	has an assertive personality						•
perseveres until the task is finished0.6081can be moody0.5442values artistic, aesthetic experiences0.4505is sometimes shy, inhibited0.7268is considerate and kind to almost everyone0.6183does things efficiently0.6183remains calm in tense situations0.6463prefers work that is routine0.5409is outgoing, sociable0.7978is sometimes rude to others0.7978gets nervous easily0.6456has few artistic interestslikes to cooperate with others0.4740is easily distracted0.4740	can be cold and aloof					0.5797	
can be moody0.5442values artistic, aesthetic experiences0.4505is sometimes shy, inhibited0.7268is considerate and kind to almost everyone0.5876does things efficiently0.6183remains calm in tense situations0.6463prefers work that is routine0.5409is outgoing, sociable0.7978is sometimes rude to others0.7978gets nervous easily0.6456has few artistic interestslikes to cooperate with others0.4740is easily distracted0.4740	perseveres until the task is finished				0.6081		, V
values artistic, aesthetic experiences0.4505vis sometimes shy, inhibited0.7268vis considerate and kind to almost everyone0.5876vdoes things efficiently0.6183vremains calm in tense situations0.6463vprefers work that is routine0.5409vis outgoing, sociable0.7978vis sometimes rude to others0.7978vgets nervous easily0.6456vlikes to reflect, play with ideas0.6456vlikes to cooperate with others0.4740vis easily distracted0.4740v	can be moody			0.5442			v V
is sometimes shy, inhibited 0.7268 v is considerate and kind to almost everyone 0.5876 v does things efficiently 0.6183 v remains calm in tense situations 0.6463 v prefers work that is routine 0.5409 v is outgoing, sociable 0.7978 v is sometimes rude to others 0.7978 v makes plans and follows through with them 0.8054 v gets nervous easily 0.4898 v likes to reflect, play with ideas 0.6456 v has few artistic interests likes to cooperate with others 0.4740 is easily distracted v	values artistic, aesthetic experiences		0.4505				v V
is considerate and kind to almost everyone 0.5876 v does things efficiently 0.6183 v remains calm in tense situations 0.6463 v prefers work that is routine 0.5409 v is outgoing, sociable 0.7978 v is sometimes rude to others 0.7959 v makes plans and follows through with them 0.8054 v gets nervous easily 0.4898 v likes to reflect, play with ideas 0.6456 v has few artistic interests likes to cooperate with others 0.4740 is easily distracted v	is sometimes shy, inhibited	0.7268					v
does things efficiently0.6183Vremains calm in tense situations0.6463Vprefers work that is routine0.5409Vis outgoing, sociable0.7978Vis sometimes rude to others0.7978Vmakes plans and follows through with them0.8054Vgets nervous easily0.4898Vlikes to reflect, play with ideas0.6456Vhas few artistic interests0.4740Vis easily distracted0.4740V	is considerate and kind to almost everyone					0.5876	v V
remains calm in tense situations 0.6463 v prefers work that is routine 0.5409 v is outgoing, sociable 0.7978 v is sometimes rude to others 0.7959 v makes plans and follows through with them 0.8054 v gets nervous easily 0.4898 v likes to reflect, play with ideas 0.6456 v has few artistic interests likes to cooperate with others 0.4740 is easily distracted v	does things efficiently				0.6183		v V
prefers work that is routine 0.5409 v is outgoing, sociable 0.7978 v is sometimes rude to others 0.7959 v makes plans and follows through with them 0.8054 v gets nervous easily 0.4898 v likes to reflect, play with ideas 0.6456 v has few artistic interests 0.4740 v is easily distracted 0.4740 v	remains calm in tense situations			0.6463			v V
is outgoing, sociable 0.7978 \vee is sometimes rule to others 0.7959 \vee makes plans and follows through with them 0.8054 \vee gets nervous easily 0.4898 \vee likes to reflect, play with ideas 0.6456 \vee has few artistic interests likes to cooperate with others 0.4740 is easily distracted	prefers work that is routine		0.5409				v V
is sometimes rude to others 0.7959 v makes plans and follows through with them 0.8054 v gets nervous easily 0.4898 v likes to reflect, play with ideas 0.6456 v has few artistic interests likes to cooperate with others 0.4740 is easily distracted 0.4740	is outgoing, sociable	0.7978					v
makes plans and follows through with them 0.8054 V gets nervous easily 0.4898 V likes to reflect, play with ideas 0.6456 V has few artistic interests 0.4740 V is easily distracted 0.4740 V	is sometimes rude to others					0.7959	v V
gets nervous easily 0.4898 v likes to reflect, play with ideas 0.6456 v has few artistic interests 0.4740 v is easily distracted 0.4740 v	makes plans and follows through with them				0.8054		v v
likes to reflect, play with ideas 0.6456 has few artistic interests 0.4740 is easily distracted 0.4740	gets nervous easily			0.4898			v v
has few artistic interests V likes to cooperate with others 0.4740 is easily distracted 0.4740	likes to reflect, play with ideas		0.6456				v v
likes to cooperate with others 0.4740 is easily distracted	has few artistic interests						v
is easily distracted	likes to cooperate with others	0.4740					
	is easily distracted						
is sophisticated in art, music, or literature 0.6223	is sophisticated in art. music. or literature		0.6223				./

ote: Factor loading above/below 0.4/-0.4 reported – 35 out of 44 load as expected
FIGURES



(a) (b) **FIGURE 1:** Differences in gains between subjects with entrepreneurial intentions and subjects without entrepreneurial intentions, considering (a) all gains games and (b) all games considered in the analysis



FIGURE 2: (a) Average marginal effects and (b) contrasting marginal effects of subjects with and without entrepreneurial intentions with respect to prior gains



FIGURE 3: (a) Average marginal effects and (b) contrasting marginal effects of subjects with and without entrepreneurial intentions with respect to degree of risk



FIGURE 4: (a) Average marginal effects and (b) contrasting marginal effects of subjects with and without entrepreneurial intentions with respect to prior pay

CHAPTER 3. ENTREPRENEURS' FOCUS OF ATTENTION AND PERCEPTION OF FINANCIAL RISK

Giulio Zichella – Copenhagen Business School, gz.ino@cbs.dk

Abstract

Content analysis of verbal answers and behavioral data from a laboratory experiment shows that individuals with and without entrepreneurial intentions differ in their perception of financial risk due to their focus of attention. Individuals with entrepreneurial intentions tend to choose risky investment prospects by consistently focusing on the size of possible monetary returns, while individuals without entrepreneurial intentions are inclined to concentrate on the probabilities attached to them. Furthermore, individuals with entrepreneurial intentions who pay more attention to monetary returns exhibit a higher perception of risk compared to individuals in the same group who focus more on probabilities, indicating an overall lower risk perception of individuals with entrepreneurial intentions. My contribution to extant entrepreneurship literature is twofold. First, I show that differences in behavior within and across groups are driven by the tight link between individuals' focus of attention and perception of financial risk. Second, I argue for the ex-ante differences in risk perception between entrepreneurs and non-entrepreneurs by showing how such differences already exist when comparing individuals with different entrepreneurial intentions. The results have practical implications for both entrepreneurs and stakeholders, particularly investors, as they need to align their assessment of financial risk with that of entrepreneurs.

1. Introduction

Defined as the subjective assessment of risk in a given opportunity, risk perception (Weber, Blais, & Betz, 2002) is an important cognitive mechanism that influences choices. Investment opportunities that are perceived as less risky are chosen more often than those that are regarded as more risky (Keh, Foo, & Lim, 2002; Simon, Houghton, & Aquino, 2000). Entrepreneurs are believed to differ from non-entrepreneurs in risk perception; they pursue certain business opportunities involving high risk and uncertainty, while others shy away (Mitchell et al., 2007; Shepherd, Williams, & Patzelt, 2015). However, there is relatively little empirical knowledge of how entrepreneurs and non-entrepreneurs differ in risk perception. From a cognitive perspective, entrepreneurs and non-entrepreneurs vary in the way they think about risky opportunities (e.g., *effectual logic*, ¹⁴ see Dew, Read, Sarasvathy, & Wiltbank, 2009; Sarasvathy, 2001) and evaluate these (Wood, McKelvie, & Haynie, 2014; Wood & Williams, 2014). Entrepreneurs accept risk as a given and focus on their personal commitment to cope with it. In contrast, non-entrepreneurs do not accept risk as a given but concentrate on minimizing it (Sarasvathy, Simon, & Lave, 1998). It has been hypothesized that these differences also affect risk perception, as entrepreneurs and non-entrepreneurs focus their attention on controlling different elements of risk (Sarasvathy et al., 1998). In this chapter, these hypotheses are tested, particularly whether entrepreneurs' focus of attention-defined as individuals' selection of available information through attention (March & Shapira, 1987)drives the differences in risk perception between entrepreneurs and non-entrepreneurs.

Accordingly, the following research question is addressed: how do individuals with and without entrepreneurial intentions differ in their perception of financial risk? This question

¹⁴ Effectual logic is defined as the logic of thinking that uniquely serves entrepreneurs when starting businesses (Sarasvathy, 2001).

is important for several reasons. First, perception plays a fundamental role in guiding decision making in environments characterized by high risk and uncertainty, such as entrepreneurship (Slovic, 1987, 2000). Hence, by studying entrepreneurs' perception of financial risk, I help explain the cognitive mechanisms leading individuals to self-select for entrepreneurship. Second, in the few studies that have measured risk perception, it has been treated primarily as a fixed personality trait (Simon et al., 2000; Weber et al., 2002; Weber & Milliman, 1997). In even fewer studies, a difference has been hypothesized between the risk perceptions of entrepreneurs versus non-entrepreneurs (Podoynitsyna, Van der Bij, & Song, 2012; Sarasvathy et al., 1998). Hence, by measuring risk perception as a dynamic aspect of decision making under risk, it is possible to test a more direct link between choice behavior and cognition, a relevant issue in entrepreneurship research (Mitchell et al., 2007). Third, entrepreneurs' focus of attention affects their choices through aspirations (typically two reference points,¹⁵ see March & Shapira, 1987, 1992). However, the question of whether or not these aspirations affect individuals' risk perception remains untested. Finally, by investigating the research question, I answer Shepherd and colleagues' (2015) recent call to test the relationship between individuals' perceptions and decision-making tools in order to understand new venture creation.

In a laboratory setting, I compared the risk choices of two groups of individuals (N = 72) who differed in entrepreneurial intentions—a sample often used in entrepreneurship literature to study the choices of entrepreneurial individuals with limited experience (Krueger, Reilly, & Carsrud, 2000). Each individual was confronted with a total of three choices between two possible investment opportunities—low-risk Option A and high-risk Option B. I

¹⁵ The literature on reference point-dependent choices is fundamental for understanding individuals' risk preferences as these depend on current states (status quo, e.g., owned assets, see Tversky & Kahneman, 1991). Individuals focus their attention on elements that are linked to reference points. Particularly, March and Shapira (1987, 1992) argue that focus of attention is linked to two reference points: a fixed survival point and an aspiration point. An individual's choice is affected by the distance between such a survival and the aspiration point. A test of March and Shapira's model in organizations supports such a view, with organizational risk preferences changing due to financial performance (Miller & Chen, 2004).

exogenously manipulated the risk of Option B. For each individual, I measured (a) his/her choices between the two options, (b) the reasons for such choices, and (c) his/her risk perception.

Several features of my experiment made the approach to the research question unique. First, I used the data from a laboratory experiment. The tasks were designed to identify the causal link between cognition and behavior in a controlled environment (McMullen & Shepherd, 2006). Second, I used a sample of individuals with and without entrepreneurial intentions, who shared similar backgrounds in terms of experience and education but differed in their entrepreneurial intentions. This sample choice had a twofold advantage. It limited the confounding effect of individuals' prior work experience, and it was useful for testing the differences in the focus of attention between individuals with very limited entrepreneurial experience.¹⁶ Third, perception was contextually measured for each choice. This represented an additional feature compared to prior studies (e.g., Sarasvathy et al., 1998; Simon et al., 2000), which allowed for testing the link between focus of attention and risk perception. Fourth, the analysis of risk perception was restricted to financial risk, an important component of the entrepreneurial environment. In essence, financial risk could be decomposed into two elements-the probability distribution of obtaining monetary outcomes and the sizes of such outcomes (Fox, Erner, & Walters, 2016). Finally, focus of attention was measured by codifying the verbal reasoning behind risk choices as suggested by Shepherd and colleagues (2015).

The results provide new insights into the differences within and across groups of entrepreneurs and non-entrepreneurs, both in their perception of financial risk and focus of attention. Across groups, individuals with and without entrepreneurial intentions differ in their

¹⁶ As entrepreneurs have already been found to have a different focus of attention compared to non-entrepreneurs (Sarasvathy et al., 1998), I contribute to extant research by showing how such differences exist before entrepreneurs acquire entrepreneurial experience. Furthermore, I show how such differences in the focus of attention result in different risk perceptions between groups.

focus of attention when faced with risky investment prospects. Particularly, when choosing between investments, the first group makes choices by consistently focusing on the size of possible monetary outcomes. In contrast, the second group concentrates on the probabilities attached to them. Such differences in focus of attention are robust to different manipulations of probabilities and outcomes. Individuals with entrepreneurial intentions also perceive relatively less risk compared to their counterparts without entrepreneurial intentions. Particularly, the first group perceives risk only when paying attention to monetary outcomes, while the second group perceives risk consistently among different foci of attention.

My contribution is twofold. First, the results add to entrepreneurship research by advancing the understanding of cognition under risk before entrepreneurial experience. This is achieved by showing that individuals with and without entrepreneurial intentions differ in their focus of attention when making choices. Second, particularly for individuals with entrepreneurial intentions, the perception of financial risk is tightly linked to their focus on the size of possible outcomes. These findings represent a step ahead in the understanding of entrepreneurs' risk perception and its direct, opportunity-specific importance in making decisions under risk.

The remainder of this chapter is organized as follows. Section 2 reviews the relevant literature and presents my arguments pertaining to risk perception. Section 3 describes the data, the sample construction, the details of the experiment, and the method for testing the proposed arguments. Section 4 presents the results. Section 5 draws the conclusions and discusses the findings' implications and limitations.

75

2. Theory

2.1. Entrepreneurial cognition

Entrepreneurial cognition is defined as the "knowledge structures that people use to make assessments, judgments, or decisions involving opportunity evaluation, venture creation, and growth" (Mitchell et al., 2002, p. 97). Research on entrepreneurial cognition focuses on understanding how entrepreneurs use mental models to simplify and piece together previously unconnected information. These mental models help entrepreneurs identify opportunities, create new products or services, and assemble the necessary resources to start and grow their businesses (Mitchell et al., 2002, 2007). Cognitive mental models ultimately help entrepreneurs cope with business risks and uncertainties.

Cognitive research on entrepreneurs' risk taking has been especially fruitful. Entrepreneurs use decision heuristics (Busenitz & Barney, 1997; Deligonul, Hult, & Cavusgil, 2008) and cognitive biases (e.g., overconfidence [Simon et al., 2000] or status quo bias [Burmeister & Schade, 2007]) in making decisions under risk. Furthermore, they are overoptimistic both in evaluating opportunities (Cassar, 2010; Parker, 2009) and predicting the likelihood of success (Cooper, Woo, & Dunkelberg, 1988; Dushnitsky, 2010). Cognitive biases and heuristics are fundamental elements behind risk taking. Many entrepreneurial decisions would never be made without the use of biases and heuristics, as they help entrepreneurs make timely decisions and deal with multiple problems connected with successfully starting a new business (Busenitz & Barney, 1997). Even so, there is no direct relationship between cognitive biases and risk taking. Instead, risk perception acts as a mediator between cognitive heuristics and behavior under risk (Keh et al., 2002; Simon et al., 2000). Particularly, heuristics foster risk taking by lowering individuals' risk perception. Despite the importance of risk perception in individuals' behavior under risk, relatively little is known about how risk perception differs among groups of individuals, particularly between entrepreneurs and non-entrepreneurs.

2.2. Entrepreneurs' risk perception

As mentioned, risk perception is defined as the subjective assessment of risk in a given opportunity (Weber et al., 2002). Risk assessment assumes an important role in entrepreneurship because individuals self-select for entrepreneurship by positively assessing opportunities (Choi & Shepherd, 2004; Haynie, Shepherd, & McMullen, 2009). Specifically, the lower the predicted financial risk is in investment opportunities, the more positive entrepreneurs' evaluations of such investments will be (Sarasvathy et al., 1998). Conversely, entrepreneurs' negative assessments of opportunities discourage such new investments and delay decisions to disinvest for existing investments (Shepherd, Wiklund, & Haynie, 2009).

While entrepreneurs are often considered risk takers, empirical evidence shows that their decisions regarding exploiting opportunities are influenced by systematic differences in cognitive processes, not necessarily by a desire to pursue risky ventures (Forlani & Mullins, 2000). Furthermore, individuals' risk perception depends on the risk characteristics of opportunities—for example, the probability of obtaining a certain financial return on investments—and cannot be treated as an individual stable trait, such as risk propensity (Weber et al., 2002). Finally, in managing risky investments, entrepreneurs accept risk as a given, while non-entrepreneurs attempt to control and minimize it (Sarasvathy et al., 1998). My contribution lies in showing that entrepreneurs select and evaluate available information differently from non-entrepreneurs and that such a difference between the two groups exists before entrepreneurial experience. Particularly, they focus their attention to guide the process of information selection. Hence, I argue that differences in financial risk perception between individuals with and without entrepreneurial intentions are driven by investment-specific risk characteristics and that the focus of attention guides such a perception.

77

2.3. Focus on outcomes versus probabilities

The illusion of control—a bias referring to individuals' overestimation of control in obtaining uncertain outcomes (Simon et al., 2000)—drives individuals to focus their attention on opportunities that are perceived as more manageable or controllable (Sitkin & Pablo, 1992; Weber et al., 2002). This influence is widely recognized in both the fields of cognitive psychology (Kahneman & Tversky, 1982; Lerner & Keltner, 2001; Loewenstein, Weber, Hsee, & Welch, 2001) and entrepreneurship (Baron, 1998; De Carolis & Saparito, 2006; McGrath & MacMillan, 1992). The entrepreneurship literature shows that at the individual level of analysis, the illusion of control positively affects both opportunity evaluations and decisions to start new ventures (Keh et al., 2002; Simon et al., 2000). Risk perception acts as a mediator in the relationship between the evaluation of opportunities and the illusion of control. The overall result is that investment opportunities that are perceived as more manageable or controllable are chosen more often than those that are viewed as less controllable (De Carolis & Saparito, 2006; Keh et al., 2002).

Under financial risk, entrepreneurs and non-entrepreneurs try to control different elements of risky investments. In contrast to managers—who try to reduce the probabilities of obtaining negative outcomes—entrepreneurs tend to accept such a set of probabilities as a given and focus on personal commitment to reach target outcomes. This argument is consistent with both the effectuation theory¹⁷ (Dew et al., 2009; Sarasvathy, 2001; Sarasvathy et al., 1998) and cognitive psychology, with the bias of the illusion of control over monetary outcomes pushing entrepreneurs to accept high levels of risk in the pursuit of greater outcomes (Forlani & Mullins, 2000). Furthermore, entrepreneurs exhibit a lower risk aversion to the size of possible monetary losses compared to managers (Koudstaal, Sloof, & van Praag, 2015), indicating that they pay

¹⁷ The theoretical reason behind such a finding is effectuation (Sarasvathy, 2001). Instead of focusing on an unpredictable future, entrepreneurs focus on minimizing risk by managing their available monetary resources (e.g., affordable loss).

greater attention to the size of monetary outcomes and that the size matters in their risk choices (Zichella & Reichstein, 2016). Finally, the effectuation logic-defined as the logic of thinking that uniquely serves entrepreneurs in starting businesses (Sarasvathy, 2001)—also helps explain why entrepreneurs differ in their feelings and management of control over risky investments. Effectuation assumes that the future is unpredictable, but entrepreneurs can still control the value-creating part of it through the use of a given set of means available to them. In this sense, entrepreneurs can utilize the means at their disposal to influence their future, without the need to predict it or to consider the expected chance probabilities. Prior research has not tested whether such a difference already exists at the early stage of entrepreneurship (e.g., among novice entrepreneurs or entrepreneurship students), when it is crucial to support efficient decision making in startups (e.g., through incubation, see Amezcua et al., 2013). Based on the preceding discussion, I argue that individuals with and without entrepreneurial intentions perceive financial risk differently due to differences in their focus of attention. Specifically, when comparing risky investment opportunities, I argue that individuals with entrepreneurial intentions choose investments by mainly considering the size of possible outcomes. In contrast, individuals without entrepreneurial intentions primarily consider the probabilities of obtaining possible outcomes. Thus, the following hypotheses are presented.

H1a. Individuals with entrepreneurial intentions choose investments by mainly focusing their attention on the size of monetary outcomes instead of on the probabilities of obtaining them.

H1b. Individuals without entrepreneurial intentions choose investments by mainly focusing their attention on the probabilities of obtaining monetary outcomes instead of on the size of such outcomes.

79

2.4. Focus of attention and risk perception

Very little is known about how pieces of information affect entrepreneurs' risk perception. Nonetheless, it is important to understand whether or not entrepreneurs can perceive the riskiness of their own actions or whether they perceive risk differently in such a way that what is considered risky by one person is not viewed as such by another (Shane, Locke, & Collins, 2003).

The link between individuals' focus of attention and risk perception is found in both the logic (or illusion) of control and individuals' aspirations. Individuals subjectively assess (and perceive) the risk of opportunities by selecting pieces of available information. On one hand, the illusion of control bias helps individuals select such pieces of information by focusing their attention on elements that are perceived as more controllable. Entrepreneurs are particularly affected by the illusion of control bias and perceive less risk in opportunities that are assessed as more controllable (De Carolis & Saparito, 2006; Keh et al., 2002; Simon et al., 2000). On the other hand, entrepreneurs' ambitions to reach a target objective (e.g., a greater but riskier monetary outcome) positively affect their risk perception by increasing their awareness of the distance between a safer survival level and their aspiration level (March & Shapira, 1992; Sarasvathy et al., 1998). I argue that the net effect of the illusion of control and individual aspirations may result in conflicting emotions, leading to an overall increase in risk perception for individuals who focus their attention on specific financial risk features. The role of conflicting emotions in increasing risk perception has been shown empirically (Podoynitsyna et al., 2012).

I argue that when individuals face two alternative risky investments, they significantly differ in their financial risk perceptions, depending on whether they focus on the outcomes or on the probabilities attached to the investments. Thus, the following hypotheses are presented.

H2a. Among individuals with entrepreneurial intentions, risk perception is higher for those who focus their attention on the size of monetary outcomes instead of on the probabilities of obtaining these.

H2b. Among individuals without entrepreneurial intentions, risk perception is higher for those who focus their attention on the probabilities of obtaining monetary outcomes instead of on the size of such outcomes.

3. Data and Method

3.1. Sample

To test the hypotheses, I used data from a computer-based quasi-laboratory experiment designed and conducted in the fall of 2014. A total of 72 students¹⁸ of business economics at a major European business school participated in the experiment. The students in the sample all shared similar backgrounds in terms of experience and education¹⁹ but differed in their entrepreneurial intentions. The purpose of this choice was to test differences between groups of individuals without entrepreneurial experience in both risk perception and focus of attention as possible confounding factors.

Among the programs with an entrepreneurship focus, 11 subjects were sampled from the local student incubator for novice entrepreneurs. The individuals in this group actively applied to enter the local incubator. After handing in their applications with detailed descriptions of the respective business ideas they wished to pursue, they followed a nine-month entrepreneurship program. Another group of 18 subjects was sampled from an entrepreneurship-

¹⁸ To determine my sample size, I have used a goodness-to-fit approach to the contingency table's chi-square tests. Based on the hypothesized large difference in the focus of attention between entrepreneurs and non-entrepreneurs, I use Cohen's *w* effect size coefficient of 0.5, and with 72 individuals, I reach a power of approximately 96% (Critical $\chi^2 = 7.81473$). This sample size is consistent with those of prior studies on entrepreneurship, which have used money games (with and without real incentives, e.g., Forlani & Mullins, 2000 [*N* = 78]; Keh et al., 2002 [*N* = 77]; Weber & Milliman, 1997 [N = 54]).

¹⁹ Admission criteria—particularly with respect to language, mathematical, and logical abilities—are shared among different study programs, making students relatively comparable in terms of abilities.

focused bachelor program. Similar to the students from the local incubator, these entrepreneurship students had to actively apply for the program by submitting individual motivation letters explaining why each of them wanted to study entrepreneurship. This bachelor program had a standard duration of three years. No barriers to entering this program existed, as all the students who applied were accepted.

Among the programs with a focus other than entrepreneurship, 17 subjects were sampled from a two-year business administration program designed for future managers. Individuals in this group did not show a current interest in starting a firm and had not been entrepreneurs previously. Another set of 26 subjects was sampled from a non-entrepreneurshipfocused bachelor program. These students were not entrepreneurs prior to starting their program, either, and did not show a current active interest in starting a firm.

I divided these 72 individuals into two groups—entrepreneurial-oriented and nonentrepreneurial oriented individuals²⁰. I assigned each person according to three criteria: first, his/her study concentration (entrepreneurship program versus normal business program); second, his/her intention to start a firm over the next three years;²¹ and third, his/her current active role as a founder in a student incubator. Active interest in entrepreneurship was valued more heavily than future interest in it. I considered the subjects actively enrolled in either the entrepreneurship program or the incubator to be individuals with entrepreneurial intentions. The students who did not apply for the entrepreneurship program represented the pool of individuals without entrepreneurial intentions since they actively chose not to specialize in entrepreneurship and instead continued in their general business economics studies. I checked that both samples

²⁰ In my main analysis, to further test the genuine nature of the group dichotomy (individuals with and without entrepreneurial intentions), I checked for differences across study concentrations, finding no statistical significance in both focus of attention and risk perception. See Table 8 in Appendix 1 for the results.

²¹ In 2016, I contacted the experiment participants and found that more than 80% of those who intended to start a firm actually did.

were representative of their respective populations in general characteristics (age, gender, and nationality); the samples reflected the demographics of the populations.

After defining the two groups of interest, I sent the participants an email invitation to join the experiment. The purpose of the experiment was not revealed in terms of both group differences and risk perception. A dedicated website was set up for individuals to register and provide their initial demographics (age and sex). All the contacted subjects agreed to participate, providing a total of 72 individuals for the investigation.

The preliminary analysis of the sampled subjects revealed that 75.4% (22) of the entrepreneurship students intended to start a firm over the next three years, while only 16.3% (7) of the control sample had such intentions ($\chi^2 = 25.561$, p < 0.000). These findings supported the choice of the group sampling, as it suggested that not only current but also future entrepreneurial intentions were greater among the entrepreneurship students than among the general business economics students. I then checked the match between entrepreneurial traits (Big Five Personality Traits, see John, Donahue, & Kentle, 1991; John, Naumann, & Soto, 2008; Zhao & Seibert, 2006) and group composition. The results confirmed the distinction between the two groups and their compositions.²²

Insert Table 1 about here

Table 1 compares the descriptive statistics between the two groups. Overall, nuanced differences were found in the demographics. Most notably, differences in part-time jobs ($\chi^2 = 8.815$, p < 0.000) and education (possessing a bachelor's degree, $\chi^2 = 3.9355$, p < 0.05) were the

²² The two groups significantly differed in their scores on *openness to experience*, one of the factors correlated with self-selection in entrepreneurship (Zhao & Seibert, 2006). Particularly, individuals with entrepreneurial intentions had positive scores (mean = 0.2321), whereas those without such intentions had negative scores (mean = -0.1565). The difference was statistically significant (*t* = -1.63, *p* = 0.05).

only probable reasons for concern. Regarding the former, only small differences were observed in the overall annual income between the groups, thereby limiting the impact of monetary considerations on choice behavior. Regarding the latter, I controlled for the differences between study concentrations in the key variables of interest, finding no difference (Appendix 1).

Age differences were also tested between the two groups by using a t-test, and no significance was found. Overall, these data findings indicated that the samples were comparable and provided some justification for considering the participants to belong to the two respective groups of individuals with and without entrepreneurial intentions.

3.2. The experimental task

A quasi-experimental design was employed to compare individuals with and without entrepreneurial intentions in their choices of investment opportunities. Programmed on the z-tree (Fischbacher, 2007), the experiment consisted of three unique choices (per individual) between two risky investments.²³ Table 2 presents the full experimental task.

Insert Table 2 about here

The subjects were asked to imagine that they were about to undertake a new investment. After being presented with descriptions of two new potential investments, they were told that both investment prospects were fully comparable in all aspects but two—the sizes of their predicted outcomes (net present values [NPVs]) and the probabilities of such outcomes. The investments were presented one at a time, and no information was given about the number of decisions to be made during the experiment. For every decision, the investments' descriptions

²³ The experiment lasted approximately 40 minutes. Although I could not completely rule out fatigue, I found no reasons for concern when considering the time spent across the different experiment subtasks.

were repeated. The subjects were asked precise questions, requiring them to measure the two key variables of interest—focus of attention and risk perception. The three pairs of investments had equal expected values but differed in their risk characteristics (probabilities and outcome sizes). Particularly, two risk characteristics of Option B were exogenously manipulated one at a time during the experiment to present a higher risk when compared to Option A.

For every decision, three questions were asked. First, the subjects were asked to choose between two alternative investments (Option A or Option B). Second, they were instructed to explain the reasons behind their choice. As the experiment was computer based,²⁴ the verbal answers were recorded as written explanations limited to 180 characters.²⁵ Verbal content analysis is a standard procedure used to analyze qualitative data, which has been recently suggested as one of the possible future directions for research exploring entrepreneurial decision making (Shepherd et al., 2015). Each answer was coded independently by three scholars (two PhD students and one assistant professor, whose research focused on entrepreneurship), who were informed about the terminology and the concepts used in this chapter. The intercoder reliability in this experiment was consistently higher than 80% when considering the participants' explanations for their decisions. The focus of attention, codified as the focus on either outcomes or probabilities, recorded the reasons why the subjects decided to choose a particular investment over another. Lastly, the subjects were asked to indicate how risky they perceived their choice to be in comparison to the alternative investment option.

²⁴ In contrast to the method of Sarasvathy and colleagues (1998), who performed content analysis on the data collected from interviews (think-aloud verbal protocol), my data was collected via a computer-based experiment, and verbal content analysis was performed based on written answers.

²⁵ The length of the answer was chosen based on the nature of the choice (simplified investment opportunities) and the objective (to identify the main reason behind the choice). The participants in both groups did not express concerns about the limited answering space during or after the experiment.

3.3. Main variables

Focus of attention

The variable used to test the cognition differences *across* groups was the individuals' focus of attention. This variable was codified with a two-step process. The first step involved defining two dummy variables that captured the individuals' focus under financial risk; the investments were chosen mainly due to the focus of attention on NPVs (the sizes of possible monetary outcomes) or on the probability distribution attached to the outcomes. We labeled these dummy variables *FocusOutcome* and *FocusProbabilities*, respectively. The second step entailed performing a verbal content analysis of the qualitative responses by using the two variables. The codification of the variables *FocusOutcome* and *FocusProbabilities* (1998), who originally codified the focus of attention on entrepreneurship.²⁶ First, I did not use a think-aloud protocol due to the nature of the experiment (computer based) but a verbal content analysis of the written data. Second, I used three independent coders instead of one. Finally, only the focus on financial risk was tested here. Table 3 presents some quotes from the sample of verbal responses.

Insert Table 3 about here

To give an example of the codification procedure, the variable *FocusOutcome* was codified as 1, with the following explanation: "I choose Option B as it presents higher potential gains compared to Option A." As for the variable *FocusProbabilities*, the following explanation (also codified as 1) was given: "A 20% probability of losing the investment is too much for me." These two variables were mutually exclusive. In limited cases, the coders did not recognize any

²⁶ Particularly, *FocusOutcome* and *FocusProbability* were codified as the variables *Contret* and *Contrisk*, respectively, in Sarasvathy and colleagues' (1998) study. Both reflected individuals' feeling of control over returns and probabilities, respectively.

main factor between the two and thus codified both as zeros. In Decision 1, both the explanations scoring zeros represented around 5% of the cases (4 out of 72 individuals), while in both Decision 2 and Decision 3, less than 3% of the explanations scored zeros (2 out of 72 individuals). Only in one case did a subject use an ambiguous verbal explanation, citing only a number to explain his/her decision. For the robustness check, these nine observations (the eight explanations scoring zeros, plus the ambiguous one) were removed; at the end, the results held at a 5% level of significance.

Financial risk perception

The variable used to test the differences *within* groups was financial risk perception, specifically, the extent to which an individual perceived his/her investment decision as risky compared to the alternative investment. Risk perception was operationalized as a dummy variable, assuming a *high* or a *low* value according to the score given on a 5-point Likert scale. Specifically, *low* was assigned to values lower than or equal to 3, and *high* was ascribed to values greater than 3.²⁷ This measurement of risk perception followed that of prior research in the field (Forlani & Mullins, 2000; Podoynitsyna et al., 2012).

Descriptively, investment Option B was consistently perceived as significantly more risky compared to Option A across the participants' decisions (Decision 1, $\chi^2 = 36.9780$, p < 0.000; Decision 2, $\chi^2 = 44.8000$, p < 0.000; Decision 3, $\chi^2 = 38.3312$, p < 0.000). However, the differences in risk perception across groups did not appear stark at first glance. The individuals with entrepreneurial intentions perceived their choices as scoring *high* for risk 34% of the time (30 out of 87 decisions), while the individuals without entrepreneurial intentions regarded their choices as risky about 32% of the time (41 out of 129 decisions). Across decisions, these

²⁷ The cutoff point of 3 between high and low values followed that of Forlani and Mullins (2000).

numbers indicated a significant difference only in Decision 3, when the entrepreneurs perceived a significantly higher risk compared to the non-entrepreneurs. Table 4 shows the results.

Insert Table 4 about here

4. Results

Descriptively, the riskier investment Option B was chosen 25% of the time, both by individuals with entrepreneurial intentions, who chose it on average about 25.28% of the time (22 out of 87 decisions), and individuals without entrepreneurial intentions, who chose it on average about 25.58% of the time (33 out of 129 decisions). These differences were not statistically significant, as shown in Table 5.

Insert Table 5 about here

4.1. Focus of attention and financial risk perception across groups

The results fully confirmed both Hypotheses 1a and 1b. The entrepreneurs differed from the non-entrepreneurs in how they focused their attention when choosing between investments. Table 6 provides a test comparing the two groups (with and without entrepreneurial intentions) in their focus on NPVs as opposed to probabilities across decisions.

Insert Table 6 about here

88

As predicted and in line with the effectuation theory, the individuals with entrepreneurial intentions consistently focused their attention on the sizes of the possible outcomes (NPVs) of the investments. In both Decision 1 and Decision 3, more than 2 out of 3 entrepreneurs (versus 1 out of 4 non-entrepreneurs) in the sample justified their investment choice by specifically mentioning the size of possible outcomes (Decision 1 difference of 38%, $\chi^2 = 9.9923$, p < 0.001; Decision 3 difference of 44%, $\chi^2 = 13.8189$, p < 0.000). In Decision 2, the difference between the two groups was still significant (difference of 34%, $\chi^2 = 8.3160$, p < 0.01).

The individuals without entrepreneurial intentions consistently focused their attention on the probabilities attached to possible outcomes. In both Decision 1 and Decision 3, more than 2 out of 3 non-entrepreneurs (versus approximately 1 out of 4 entrepreneurs) in the sample justified their investment choice by mainly mentioning the risk connected to the probability distribution of possible outcomes (Decision 1 and Decision 3 difference of 46%, $\chi^2 = 14.4351$, p < 0.000). In Decision 2, the difference between the two groups was also significant (difference of 38%, $\chi^2 = 9.9923$, p < 0.001).²⁸

4.2. Focus of attention and financial risk perception within groups

The results confirmed Hypothesis 2a, while Hypothesis 2b was not verified. Only in the group with entrepreneurial intentions did the individuals' focus of attention affect their financial risk perception in a significantly different way. Table 7 provides a test for the dependence between focus of attention and financial risk perception within groups.

 $^{^{28}}$ Note that the percentages do not total one hundred because in a few cases (<10%), the three coders did not code focus of attention as driven by either outcomes or probabilities but by other factors.

Insert Table 7 about here

In the group of individuals with entrepreneurial intentions, the subjects who focused their attention on the sizes of possible outcomes (NPVs) perceived more financial risks than the individuals who focused on probabilities. However, in the group of non-entrepreneurs, the difference between the subjects focusing on outcomes and on probabilities was not statistically significant. While this evidence should be interpreted cautiously, the results do suggest that individuals with entrepreneurial intentions (who perceive high risk only when focusing on monetary outcomes) perceive on average *less* risk compared to individuals without entrepreneurial intentions (who perceive risk consistently when focusing either on monetary outcomes or probabilities). These findings are in line with those of prior research hypothesizing a lower risk perception for entrepreneurs compared to non-entrepreneurs (Keh et al., 2002; Podoynitsyna et al., 2012; Sarasvathy et al., 1998; Simon et al., 2000).

Given the nature of the experiment, two robustness checks were performed. First, to further validate the results of the grouping dichotomy (with and without entrepreneurial intentions), I checked for differences between the subjects assigned to the two groups. The results (presented in Appendix 1) indicated no reason for concern about the grouping decision with respect to both *focus of attention* and *risk perception*. Second, I checked if the results were influenced by ambiguous responses, defined as the verbal answers that the coders interpreted differently. The choice to remove such observations and results from the analysis did hold. Finally, the findings were robust in comparison to alternative explanations, such as risk propensity and loss aversion, as the individuals belonging to the two groups did not significantly differ in the former, and the hypothesized predictions were verified in the choices where losses were possible (Decision 3).

5. Conclusions

In this chapter, the following research question has been addressed: how do individuals with and without entrepreneurial intentions differ in their perception of financial risk? Several studies have identified risk perception as a primary driver of self-selection in entrepreneurship. The laboratory experiment performed in this study, comparing individuals with and without entrepreneurial intentions in a controlled environment, contributes to the entrepreneurship literature by finding significant differences in financial risk perception. In this chapter, I have tested two specific sets of hypotheses.

First, I have hypothesized that individuals with and without entrepreneurial intentions would differ in their focus of attention when choosing between risky investments. Focus of attention has been considered a factor influencing entrepreneurs' risk perception due to two important characteristics of investments-NPVs and probabilities of success-having different subjective values. The content analysis of the verbal answers reveal that when making decisions between investments, the individuals with entrepreneurial intentions focus on the sizes of possible investment outcomes, while the individuals without entrepreneurial intentions focus on the probabilities of obtaining such outcomes. The results are robust in comparison to several alternative explanations, including individual risk and loss aversion. As I have tested the hypotheses on the sample of individuals with entrepreneurial intentions, the results show that differences in focus of attention and perception exist before the individuals become entrepreneurs, which are particularly relevant for understanding the cognition of individuals with limited entrepreneurial experience (e.g., novice entrepreneurs). Furthermore, the study's results have practical implications for both private and public stakeholders. One such implication is that private investors and entrepreneurs should align their assessments of risky opportunities by comparing their focus on key elements of financial risk. Differences in focus of attention between individuals with and without entrepreneurial intentions reveal different

91

motives behind their actions under risk that need to be aligned between entrepreneurs and stakeholders because entrepreneurs tend to accept risk as a given and leverage personal commitment to reach target objectives. Another conflicting implication is that focusing on probabilities of obtaining monetary values pushes individuals without entrepreneurial intentions to adopt a more conservative view on investment opportunities. In this regard, the results imply that public institutions investing in entrepreneurship (e.g., universities, public incubators, etc.) have the potential to tailor programs that are better designed to nurture novice entrepreneurs. Such programs should consider the impact of cognitive elements, such as focus of attention, on starting new business ventures, which could ultimately aid in reducing entrepreneurs' underestimation of risk.

Second, I have hypothesized that differences in focus of attention across groups would directly influence individuals' risk perception. While prior literature has identified risk perception as a factor affecting entrepreneurial action, few papers have attempted to measure risk perception. This study has tested the link between risk perception and focus of attention within groups. As individuals focus on different elements when making decisions, I argue that such differences would be reflected among various individuals' and groups' risk perceptions. Specifically, I have hypothesized that individuals with entrepreneurial intentions would perceive a high-level risk when focusing on monetary outcomes, while individuals without entrepreneurial intentions would perceive a high-level risk when focusing on probabilities. The results partly confirm the hypotheses, with the link between focus of attention and risk perception only significant in the group of individuals with entrepreneurial intentions. Although these results are interpreted with caution, some research and practical implications are evident. From the research perspective, the exploratory findings suggest more nuanced differences in risk perception between entrepreneurs and non-entrepreneurs than expected (Sarasvathy et al., 1998). Entrepreneurs appear to be more affected in their risk perception when focusing on outcomes

than non-entrepreneurs when focusing on probabilities. Overall, these differences make entrepreneurs perceive risk *less* than non-entrepreneurs. These findings indicate a different subjective value of available information when evaluating investment opportunities. As information translates into a very different risk perception for individuals with versus without entrepreneurial intentions, further research on entrepreneurship is needed to assess the impacts of information on both behavior and risk perception.

From a private stakeholder's perspective, the results indicate that entrepreneurs are not blind risk takers but perceive risk in a significantly different way as a function of their focus of attention. Particularly for entrepreneurs, predictive information does not drive risk perception as much as monetary information. As the importance of monetary feedback has already been explored (Zichella & Reichstein, 2016), future research on entrepreneurship is required to further understand (a) how focus of attention affects risk perception beyond financial risk²⁹ and (b) how the limited available predictive information on investment opportunities is used by entrepreneurs versus non-entrepreneurs. While leaving the former challenge to future studies, the latter will be explored in Chapter 4 of this dissertation.

5.1. Limitations and avenues for future research

I acknowledge several limitations in the current study, the most important being the narrow definition of risk, experimental shortcomings, external validity, and possible confounding factors due to unobservable variables (e.g., learning). Financial risk is recognized as only one aspect of evaluating opportunities. However, this study has used a precise and quantifiable definition of risk by narrowing down the scope of the analysis to financial measures. Financial risk is a good indicator of business risk due to its strong connection with the

²⁹ Particularly, eye tracking and functional magnetic resonance could be appropriate tools for further studies on focus of attention, with the former measuring attention through overt eye orienting and the latter linking attention to brain activity (Duchowski, 2007).

idea that one value, such as profit or the probability of failure, can indicate firms' success and risk, respectively (March & Shapira, 1987).

The laboratory setting limits how much and to how many I can generalize the results. My laboratory setting was designed to study two aspects of individual cognition—focus of attention and risk perception. While cognition is essential in explaining behavior, solely studying cognitive mechanisms is acknowledged as a simplification. Nonetheless, little is known about entrepreneurs' ways of connecting the dots in financial risk assessment (Baron & Ensley, 2006; Mitchell et al., 2007).

Even when using laboratory experiment data, it is recognized that a researcher cannot identify all motivational differences among individuals in their actual risk taking in realworld investment opportunities. Motivational differences indeed influence the entrepreneurial process, and providing real monetary incentives makes it possible to observe the motives behind financial risk taking. With this in mind, it is beyond the scope of this chapter to measure the differences in risk by using its broader definition, but future research in this direction is considered quite promising.

With respect to external validity and the sample used, it is granted that there are advantages and disadvantages of using individuals with and without entrepreneurial intentions. Using a sample of real, experienced entrepreneurs (compared to students) would have required careful management of the endogenous nature of choice by such individuals, with substantial heterogeneity in entrepreneurial experience. In this regard, individuals with entrepreneurial intentions are good proxies for novice entrepreneurs and are easier to access. Although the two groups sampled do not show differences in experience and learning, I acknowledge that I cannot identify possible alternative explanations due to unobservable variables, such as classroom socialization and learning connected to risk. Moreover, due to the low probability of success at the entrepreneurial level, it is reasonable to believe that people who are willing to start a new venture despite these odds might be more optimistic or have higher self-efficacy than people deterred by such drawbacks. Incorporating this individual-level variation in motivation is considered important to the entrepreneurial process. It is believed that future research will exceed these limitations and further advance our understanding of risk perception in entrepreneurship, for example, by exploring the neurological motives behind risk taking among entrepreneurs (Tom, Fox, Trepel, & Poldrack, 2007).

References

Amezcua, A. S., Grimes, M. G., Bradley, S. W., & Wiklund, J. (2013). Organizational sponsorship and founding environments: a contingency view on the survival of business-incubated firms, 1994–2007. *Academy of Management Journal*, *56*(6), 1628-1654.

Baron, R. A. (1998). Cognitive mechanisms in entrepreneurship: Why and when entrepreneurs think differently than other people. *Journal of Business Venturing*, *13*(4), 275–294.

Baron, R. A., & Ensley, M. D. (2006). Opportunity recognition as the detection of meaningful patterns: Evidence from comparisons of novice and experienced entrepreneurs. *Management Science*, *52*(9), 1331–1344.

Burmeister, K., & Schade, C. (2007). Are entrepreneurs' decisions more biased? An experimental investigation of the susceptibility to status quo bias. *Journal of Business Venturing*, 22(3), 340–362.

Busenitz, L. W., & Barney, J. B. (1997). Differences between entrepreneurs and managers in large organizations: Biases and heuristics in strategic decision-making. *Journal of Business Venturing*, *12*(1), 9–30.

Cassar, G. (2010). Are individuals entering self-employment overly optimistic? An empirical test of plans and projections on nascent entrepreneur expectations. *Strategic Management Journal*, *31*(8), 822–840.

Choi, Y. R., & Shepherd, D. A. (2004). Entrepreneurs' decisions to exploit opportunities. *Journal of Management*, *30*(3), 377–395.

Cooper, A. C., Woo, C. Y., & Dunkelberg, W. C. (1988). Entrepreneurs' perceived chances for success. *Journal of Business Venturing*, *3*(2), 97–108.

De Carolis, D. M., & Saparito, P. (2006). Social capital, cognition, and entrepreneurial opportunities: A theoretical framework. *Entrepreneurship Theory and Practice*, *30*(1), 41–56.

Deligonul, Z. S., Hult, G. T. M., & Cavusgil, S. T. (2008). Entrepreneuring as a puzzle: An attempt to its explanation with truncation of subjective probability distribution of prospects. *Strategic Entrepreneurship Journal*, *2*(2), 155–167.

Dew, N., Read, S., Sarasvathy, S. D., & Wiltbank, R. (2009). Effectual versus predictive logics in entrepreneurial decision-making: Differences between experts and novices. *Journal of Business Venturing*, 24(4), 287–309.

Duchowski, A. (2007). *Eye tracking methodology: Theory and practice* (Vol. 373). Springer Science & Business Media.

Dushnitsky, G. (2010). Entrepreneurial optimism in the market for technological inventions. *Organization Science*, 21(1), 150–167.

Fischbacher, U. (2007). z-Tree: Zurich toolbox for ready-made economic experiments. *Experimental economics*, 10(2), 171-178.

Forlani, D., & Mullins, J. W. (2000). Perceived risks and choices in entrepreneurs' new venture decisions. *Journal of Business Venturing*, 15(4), 305–322.

Fox, C. R., Erner, C., & Walters, D. J. (2015). Field to the laboratory and back. In *The Wiley Blackwell handbook of judgment and decision making* (pp. 43-88).

Haynie, J. M., Shepherd, D. A., & McMullen, J. S. (2009). An opportunity for me? The role of resources in opportunity evaluation decisions. *Journal of Management Studies*, 46(3), 337–361.

John, O. P., Donahue, E. M., & Kentle, R. L. (1991). *The Big Five inventory—versions 4a and 54*. Berkeley, CA: University of California, Berkeley, Institute of Personality and Social Research.

John, O. P., Naumann, L. P., & Soto, C. J. (2008). Paradigm shift to the integrative Big Five trait taxonomy: History, measurement, and conceptual issues. In O. P. John, R. W. Robins, & L. A. Pervin (Eds.), *Handbook of personality: Theory and research* (pp. 114–158). New York: Guilford Press.

Kahneman, D., & Tversky, A. (1982). The psychology of preferences. *Scientific American 135-42*.

Keh, H. T., Foo, M. D., & Lim, B. C. (2002). Opportunity evaluation under risky conditions: The cognitive processes of entrepreneurs. *Entrepreneurship Theory and Practice*, 27(2), 125–148.

Koudstaal, M., Sloof, R., & Van Praag, M. (2015). Risk, uncertainty, and entrepreneurship: Evidence from a lab-in-the-field experiment. *Management Science*, 62(10), 2897–2915.

Krueger, N. F., Reilly, M. D., & Carsrud, A. L. (2000). Competing models of entrepreneurial intentions. *Journal of Business Venturing*, 15(5), 411–432.

Lerner, J. S., & Keltner, D. (2001). Fear, anger, and risk. *Journal of Personality and Social Psychology*, *81*(1), 146-159.

Loewenstein, G. F., Weber, E. U., Hsee, C. K., & Welch, N. (2001). Risk as feelings. *Psychological Bulletin*, 127(2), 267-286.

March, J. G., & Shapira, Z. (1987). Managerial perspectives on risk and risk taking. *Management Science*, 33(11), 1404–1418.

March, J. G., & Shapira, Z. (1992). Variable risk preferences and the focus of attention. *Psychological Review*, 99(1), 172-183.

McGrath, R. G., & MacMillan, I. C. (1992). More like each other than anyone else? A cross-cultural study of entrepreneurial perceptions. *Journal of Business Venturing*, 7(5), 419–429.

McMullen, J. S., & Shepherd, D. A. (2006). Entrepreneurial action and the role of uncertainty in the theory of the entrepreneur. *Academy of Management Review*, *31*(1), 132–152.

Miller, K. D., & Chen, W. R. (2004). Variable organizational risk preferences: Tests of the March–Shapira model. *Academy of Management Journal*, 47(1), 105–115.

Mitchell, R. K., Busenitz, L. W., Bird, B., Marie Gaglio, C., McMullen, J. S., Morse, E. A., & Smith, J. B. (2007). The central question in entrepreneurial cognition research 2007. *Entrepreneurship Theory and Practice*, *31*(1), 1–27.

Mitchell, R. K., Busenitz, L., Lant, T., McDougall, P. P., Morse, E. A., & Smith, J. B. (2002). Toward a theory of entrepreneurial cognition: Rethinking the people side of entrepreneurship research. *Entrepreneurship Theory and Practice*, 27(2), 93–104.

Parker, S. C. (2009). Can cognitive biases explain venture team homophily? *Strategic Entrepreneurship Journal*, 3(1), 67–83.

Podoynitsyna, K., Van der Bij, H., & Song, M. (2012). The role of mixed emotions in the risk perception of novice and serial entrepreneurs. *Entrepreneurship Theory and Practice*, *36*(1), 115–140.

Sarasvathy, S. D. (2001). Causation and effectuation: Toward a theoretical shift from economic inevitability to entrepreneurial contingency. *Academy of Management Review*, *26*(2), 243–263.

Sarasvathy, D. K., Simon, H. A., & Lave, L. (1998). Perceiving and managing business risks: Differences between entrepreneurs and bankers. *Journal of Economic Behavior & Organization*, 33(2), 207–225.

Shane, S., Locke, E. A., & Collins, C. J. (2003). Entrepreneurial motivation. *Human Resource Management Review*, 13(2), 257–279.

Shepherd, D. A., Wiklund, J., & Haynie, J. M. (2009). Moving forward: Balancing the financial and emotional costs of business failure. *Journal of Business Venturing*, 24(2), 134–148.

Shepherd, D. A., Williams, T. A., & Patzelt, H. (2015). Thinking about entrepreneurial decision making review and research agenda. *Journal of Management*, *41*(1), 11–46.

Simon, M., Houghton, S. M., & Aquino, K. (2000). Cognitive biases, risk perception, and venture formation: How individuals decide to start companies. *Journal of Business Venturing*, *15*(2), 113–134.

Sitkin, S. B., & Pablo, A. L. (1992). Reconceptualizing the determinants of risk behavior. *Academy of Management Review*, 17(1), 9–38.

Slovic, P. (1987). Perception of risk. *Science*, 236(4799), 280–285.

Slovic, P. E. (2000). *The perception of risk*. Earthscan Publications.

Tom, S. M., Fox, C. R., Trepel, C., & Poldrack, R. A. (2007). The neural basis of loss aversion in decision-making under risk. *Science*, *315*(5811), 515–518.

Tversky, A., & Kahneman, D. (1991). Loss aversion in riskless choice: A reference-dependent model. *The quarterly journal of economics*, *106*(4), 1039-1061.

Weber, E. U., Blais, A. R., & Betz, N. E. (2002). A domain-specific risk-attitude scale: Measuring risk perceptions and risk behaviors. *Journal of Behavioral Decision Making*, 15, 263–290.

Weber, E. U., & Milliman, R. A. (1997). Perceived risk attitudes: Relating risk perception to risky choice. *Management Science*, 43(2), 123–144.

Wood, M. S., McKelvie, A., & Haynie, J. M. (2014). Making it personal: Opportunity individuation and the shaping of opportunity beliefs. *Journal of Business Venturing*, 29(2), 252–272.

Wood, M. S., & Williams, D. W. (2014). Opportunity evaluation as rule-based decision making. *Journal of Management Studies*, *51*(4), 573–602.

Zhao, H., & Seibert, S. E. (2006). The Big Five personality dimensions and entrepreneurial status: A meta-analytical review. *Journal of Applied Psychology*, 91(2), 259-271.

Zichella, G., & Reichstein, T. (2016). "A risk worth taking." Risk as entrepreneurs' conditional choice. [Mimeo]. Copenhagen Business School.

6. Tables

	With entrepreneurial intentions (<i>n</i> = 29)	Without entrepreneurial intentions (<i>n</i> = 43)	χ ²	t-test	Significance
Risk averse	13 (44.8)	20 (46.5)	0.0832		
Risk neutral	7 (24.1)	11 (25.6)			
Risk prone	9 (31.1)	12 (27.9)			
Male	22 (75.8)	32 (74.4)	0.0192		
Female	7 (24.2)	11 (25.6)			
Danish	23 (79.3)	41 (95.1)	4.8078		*
Foreign	6 (20.7)	2 (4.9)			
50k DKR < income < 300k					
DKR	23 (79.3)	40 (93.0)	2.9777		*
0 < income < 50k DKR	6 (20.7)	3 (7.0)			
Parent entrepreneur	7 (24.2)	10 (23.3)	0.0075		
No parent entrepreneur	22 (75.8)	33 (76.7)			
University degree	15 (51.7)	32 (74.4)	3.9355		**
No university degree	14 (48.3)	11 (25.6)			
Employed part time	17 (58.6)	38 (88.4)	8.4997		***
Not employed part time	12 (41.4)	5 (11.6)			
Entrepreneur in 3 years Not entrepreneur in 3	22 (75.9)	7 (16.3)	25.5610		***
years	7 (24.1)	36 (83.7)			
Overconfidence	70.37 (2.131)	72.06 (1.268)		0.4734	
Age	23.68 (0.726)	24.67 (0.938)		0.4478	

Table 1: Descriptive statistics and test of differences between individuals with and without entrepreneurial intentions (N = 72)

***p < 0.01, **p < 0.05, *p < 0.1; percentages of table total in parentheses

	Option A	A	Option B	
	Probability (%)	NPV (US\$)	Probability (%)	NPV (US\$)
Decision 1	30	300,000	30	0
	40	400,000	40	400,000
	30	300,000	30	800,000
Decision 2	30	300,000	20	0
	40	400,000	60	400,000
	30	300,000	20	800,000
Decision 3	30	300,000	20	-10,000
	40	400,000	60	400,000
	30	300,000	20	810,000

Table 2: Choice tasks (adapted from Sarasvathy et al., 1998): Three decisions between a less risky investment (Option A) and a riskier investment (Option B)

Questions for every decision:

- 1) Which project would you invest in? (Option A or Option B)
- 2) Why would you invest as such? (Verbal answer, 180-character limit)
- 3) Compared to the other investment, how risky is your choice? (5-point Likert scale)

 Table 3: Sample answers and codification of two main variables (keywords for codification in bold). Groups with and without entrepreneurial intentions

Variable	Group	Verbal explanation
FocusOutcome	With Without With	"Relatively the same NPV, but there is a potential for <i>higher gains</i> " "At a minimum, I would rather <i>get \$300,000</i> than \$0" " <i>Higher payout</i> possibility is reasonably leveraged"
Focus Probability	With Without Without	"I choose Project 1 because 30% probability of \$0 is too high for me" "To make sure that I have the highest probability for cash flow" "There is no chance to lose all the money"

 Table 4: Differences in *high* financial risk perception between entrepreneurs and nonentrepreneurs, across decisions

	With entrepreneurial intentions (<i>n</i> = 29)	Without entrepreneurial intentions (<i>n</i> = 43)	χ ²	Significance
Decision 1	17.24%	27.91%	1.0923	
Decision 2	48.28%	51.16%	0.0577	
Decision 3	37.93%	16.23%	4.3304	**

***p < 0.01, **p < 0.05, *p < 0.1; percentages of group total

	With entrepreneurial	Without entrepreneurial		
	intentions (<i>n</i> = 29)	intentions (<i>n</i> = 43)	χ2	Significance
Decision 1, Option A	26 (0.90)	34 (0.79)	1.3973	
Decision 1, Option B	3 (0.10)	9 (0.21)		
Decision 2, Option A	17 (0.59)	25 (0.58)	0.0016	
Decision 2, Option B	12 (0.41)	18 (0.42)		
Decision 3, Option A	22 (0.76)	37 (0.86)	1.2142	
Decision 3, Option B	7 (0.23)	6 (0.14)		
-				

 Table 5: Differences in choices between individuals with and without entrepreneurial intentions, across decisions

***p < 0.01, **p < 0.05, *p < 0.1; percentages of group total in parentheses

1 abic 0 , individuals inclusions decisions (verbal response analysis	Tab	ole 6:	Individuals'	focus across	decisions	(verbal re	esponse analys	sis)
--	-----	--------	--------------	--------------	-----------	------------	----------------	------

	With entrepreneurial intentions (<i>n</i> = 29)	Without entrepreneurial intentions (<i>n</i> = 43)	Difference	χ²	Significance
Focus on outcomes					
Decision 1	66%	28%	38%	9.9923	***
Decision 2	62%	28%	34%	8.3160	***
Decision 3	72%	28%	44%	13.8189	***
Focus on probabilities					
Decision 1	24%	70%	46%	14.4351	* * *
Decision 2	34%	72%	38%	9.9923	* * *
Decision 3	24%	70%	46%	14.4351	* * *

***p < 0.01, **p < 0.05, *p < 0.1; percentages of group total

	With entrepreneurial intent	tions (<i>n</i> = 29)	Fisher exact test	Significance
	Focus outcomes	Focus probabilities		
Decision 1	26%	0%	0.098	*
Decision 2	45%	50%	0.558	
Decision 3	50%	11%	0.053	*
	Without entrepreneurial (<i>n</i> = 43)	intentions		
	Focus outcomes	Focus probabilities		
Decision 1	23%	30%	0.471	
Decision 2	42%	55%	0.332	
Decision 3	15%	17%	0.648	

Table 7: Differences within groups in their *high* financial risk perception, across decisions

***p < 0.01, **p < 0.05, *p < 0.1; percentages of group total
7. Appendix 1

			Difference	χ ²	Significance
	Incubator (n = 11)	Entrepreneurship bachelor program (<i>n</i> = 18)			
Decision 1	72%	61%	11%	0.4078	
Decision 2	63%	61%	2%	0.0185	
Decision 3	72%	66%	6%	0.1172	
	Business administration (n = 17)	Non- entrepreneurship bachelor program (n = 26)			
Decision 1	23%	31%	8%	0.2678	
Decision 2	23%	31%	8%	0.2678	
Decision 3	18%	35%	17%	1.4710	

Table 8: Shares of individuals who focus on NPVs as opposed to probabilities acrossdecisions (by study concentrations)

****p* < 0.01, ***p* < 0.05, **p* < 0.1; percentages of group total

CHAPTER 4. IT'S ALL OR NOTHING: ENTREPRENEURS' WILLINGNESS TO BEAR UNCERTAINTY

Giulio Zichella – Copenhagen Business School – gz.ino@cbs.dk

ABSTRACT

Entrepreneurs are often considered to be more willing to bear and manage uncertainty than non-entrepreneurs. However, the available empirical evidence regarding this topic is limited and mixed. This chapter outlines the performance of a quasilaboratory experiment that investigated whether a lack of predictive information affects individuals with entrepreneurial intentions in their likelihood to choose uncertainty vis-àvis certainty. Drawing on prospect theory, it is argued that individuals with entrepreneurial intentions are less sensitive than individuals without entrepreneurial intentions due to a framing effect that occurs when information about probabilities is manipulated. In particular, it is argued that individuals with entrepreneurial intentions are more likely than their comparable counterparts to choose consistently among monetary opportunities that share everything in common but information about probabilities of success. This study finds support for the argument, as individuals with entrepreneurial intentions have preferences pertaining to uncertainty that are more stable compared to those of individuals without such intentions. The results hold after controlling for a number of alternative explanations, including status quo bias, prior gain effect, and risk propensity effect. These results have implications for the understanding of factors that guide entrepreneurial choices under uncertainty and, ultimately, entrepreneurial action.

Keywords: risk, uncertainty, entrepreneur, money games, lab experiment, framing effect, odds of success

INTRODUCTION

Uncertainty-defined as the lack of predictive information (Knight, 1921)-is a fundamental variable that entrepreneurs have to manage. Entrepreneurs are often considered more willing than non-entrepreneurs to bear and manage uncertainty because they self-select in an environment where choices have to be taken despite a lack of critical information (e.g., on expected returns). However, prior research provides mixed evidence on entrepreneurs' willingness to bear uncertainty. Some contributions suggest that entrepreneurs are not more willing to bear uncertainty compared to nonentrepreneurs (e.g., McKelvie et al., 2011; O'Brien et al., 2003), while others suggest that differences between these groups are driven by circumstantial factors such as strategic competition (Holm et al., 2013), monetary losses (Koudstaal et al., 2015), and decision framing (Dew et al., 2009). This latter stream of research aims at understanding whether or not entrepreneurs are unique in their decision making under uncertainty, and new contributions in this direction have been encouraged (for a review, see Shane & Ulrich, 2004; Shepherd et al., 2015). This chapter acts as a follow-up to prior studies that have argued in favor of a different cognitive approach to uncertainty between entrepreneurs and non-entrepreneurs. In particular, it has been argued that entrepreneurs are less likely to predict an unknowable future (Sarasvathy, 2001), and accordingly tend to focus less on predicting information (Zichella, 2017). However, the link between willingness to bear uncertainty and the lack of predictive information has not yet been tested within entrepreneurship research. Individuals with different entrepreneurial intentions provide a relevant group of analysis to test whether differences in willingness to bear uncertainty exist before entrepreneurial experience (Krueger et al., 2000).

The research question follows this call to action: How do individuals with and without entrepreneurial intentions differ in their willingness to bear uncertainty?

107

Answering this research question is important for several reasons. First, willingness to bear uncertainty is positively associated with self-selection into entrepreneurship, a relevant phenomenon to public welfare (Baron & Ensley, 2006; McMullen & Shepherd, 2006; Shane & Venkataraman, 2000). Understanding under which conditions individuals are more willing to bear uncertainty may help public and private stakeholders to support entrepreneurial action. Second, research on entrepreneurs' behavior under uncertainty is timely, as public and private institutions (e.g., new venture *incubators* and *accelerators*; Amezcua et al., 2013) currently aim at reducing entrepreneurs' chances of failure by focusing on uncertainty management. Furthermore, contributions that compare individuals with different entrepreneurial intentions in their behavior under uncertainty are scarce, particularly those that use an experimental methodology (Shepherd et al., 2015). Third, a better understanding of contextual factors influencing entrepreneurs' willingness to bear uncertainty will help stakeholders to align their objectives with their entrepreneurial partners. Finally, as I specifically explore how information on probabilities of success influences willingness to bear uncertainty, I shed light on a key factor that can help entrepreneurs to assess both the feasibility of an investment and how it affects individuals with different entrepreneurial intentions differently (McMullen & Shepherd, 2006).

In this chapter, I make two assumptions. First, I assume that decision makers suffer from cognitive biases and use heuristics to select among available information under uncertainty (Gigerenzer et al., 1999; Kahneman & Tversky, 1979; March, 1994; Shah & Oppenheimer, 2008). In particular, individuals show inconsistent preferences due to different presentations of the same piece of information (Kahneman & Tversky, 1974; Tversky & Kahneman, 1989). This bias is known as the *framing effect*, a violation of the principle of invariance that underlies the rational theory of choice. Individuals suffer

from the framing effect in a variety of situations, such as when ambiguity or vagueness are involved (Ellsberg, 1961; Tversky & Kahneman, 1981). As an entrepreneur is someone who exercises business judgment in the face of uncertainty, I explore here how a lack of information on probabilities of success affects choices of individuals with and without entrepreneurial intentions differently. I argue that individuals with entrepreneurial intentions are less subject than individuals without entrepreneurial intentions are less subject than individuals without entrepreneurial intentions to the framing effect when information about probabilities is manipulated. In particular, individuals with entrepreneurial intentions exhibit a bias toward opting for an uncertain higher monetary gain (vs. a certain lower monetary gain) regardless of the availability of predictive information.³⁰ This argument resonates well with the finding that entrepreneurs frame decisions while paying less attention to predictive information (Dew et al., 2009).

The second assumption I make is that both the *locus and logic of control*³¹ affect entrepreneurs' discovery and exploitation of opportunities (Nordgren et al., 2007; Sarasvathy, 2001). In particular, while non-entrepreneurs focus more on prediction, entrepreneurs make choices by focusing more on opportunities that they subjectively feel in control of. Furthermore, entrepreneurs focus on controlling possible outcomes instead of their odds of success. These findings hold true for individuals with different entrepreneurial experience (serial entrepreneurs, novice entrepreneurs, and individuals with entrepreneurial intentions; see Sarasvathy et al., 1998; Zichella, 2016). Consequently, I argue that individuals with entrepreneurial intentions are more likely

 $^{^{30}}$ Due to the principle of indifference, individuals can assign the probability 1/2 to the two possible monetary outcomes in the uncertain lottery.

³¹ Locus of control refers to the extent to which individuals believe they can control events affecting them (Rotter, 1966). The logic of control refers to the individuals' preference to focus on controllable aspects of an unpredictable future (Sarasvathy, 2001)

than individuals without entrepreneurial intentions to choose consistently between two prospects that share everything in common except for information about probabilities.

I approach the research question methodologically by using real money games in a laboratory quasi-experiment.³² Scholars have recently started testing entrepreneurs' preferences when faced with real monetary incentives, contributing to our understanding of the role of loss aversion and strategic dynamics under risk and uncertainty (Holm et al., 2014; Koudstaal et al., 2015). I aim at uncovering the role of information pertaining to probabilities of success when real monetary outcomes are at stake. The sample in this study includes students with different entrepreneurial intentions. Some of these individuals exhibit an active entrepreneurial intention as they have selected into a designated entrepreneurship program after their first semester of attending classes. These individuals are motivated by a desire to pursue a career as an entrepreneur, and are similar in many respects (e.g., age, gender, education) to other students except for their entrepreneurial intention. The choice to select a sample of students rather than a random sample from the entire population was motivated by two main reasons. First, using entrepreneurial intentions-defined as the cognitive state that precedes the decision to form a new venture—as a proxy for entrepreneurship is consistent with prior research (Krueger et al., 2000; Lee et al., 2011; Zellweger et al., 2011). Second, by using individuals possessing entrepreneurial intentions, I limit alternative explanations due to heterogeneity in professional experience. It is worth noting that the samples used in Chapter 2 and this chapter are identical. This choice is deliberate, as it makes it possible to compare results under two different conditions-risk and uncertainty, respectively.

³² Compared with a traditional laboratory experiment, a quasi-laboratory experiment lacks the element of random assignment to treatment group or control group, in this case with or without entrepreneurial intentions.

The experiment consisted of asking participants to choose between two possible combinations—certainty vs. risk and certainty vs. uncertainty—with real monetary incentives. Monetary combinations were presented in a non-random order and in proximity to each other for the purpose of being able to directly compare preferences between the two groups. Monetary rewards were given to each subject at the end of the experiment. The results revealed that instead of pursuing a certain monetary gain, individuals with entrepreneurial intentions consistently chose the lottery option regardless of whether information about probabilities was given (risk) or not (uncertainty). Such an effect is robust to alternative explanations such as status quo bias, prior gain effect, and risk propensity effect.³³ Overall, these results suggest that differences between individuals with and without entrepreneurial intentions under uncertainty are due to a different level of sensitivity to the presence of predictive information.

The remainder of this chapter is organized as follows. Section 2 briefly reviews the relevant literature linking predictive information with uncertainty taking. Section 3 describes data, sample construction, the details of the experiment, and the method for testing the research question. Section 4 presents the results. Section 5 concludes and discusses the implications of the findings.

THEORY AND HYPOTHESES

Uncertainty—defined as immeasurable risk (Knight, 1921)—constitutes a conceptual cornerstone in entrepreneurship literature as entrepreneurs face an unknowable future. Entrepreneurship requires judgments to be made about whether to pursue an opportunity or not and, at the individual level of analysis, an entrepreneur is

³³ By controlling for the two mechanisms tested in Chapter 2—prior gain effect and risk propensity effect—it is possible for me to exclude alternative explanations of results under uncertainty.

someone who exercises business judgment in the face of uncertainty (Hebert & Link, 1988). Therefore, it has been suggested that a higher willingness to bear uncertainty is a distinctive characteristic of entrepreneurs, especially when compared to nonentrepreneurs (McMullen & Shepherd, 2006). However, empirical evidence is mixed. On the one hand, recent available findings do not support an overall greater entrepreneurial willingness to bear uncertainty (McKelvie et al., 2011; O'Brien et al., 2003). On the other hand, it has been recently suggested that there is a need to explore how certain specific factors (e.g., a monetary gain or loss; Koudstaal et al., 2015; Zichella, 2016) can trigger entrepreneurs' willingness to bear uncertainty in a way that exceeds the willingness of non-entrepreneurs (Shepherd et al., 2015). Following this research direction, this chapter explores how availability of information affects individuals' willingness to bear uncertainty is a product of a learning process due to experience: To control for such a potential explanation, I compared individuals with and without entrepreneurial intentions.

The concept of uncertainty in entrepreneurship finds its roots in the seminal work of Knight (1921). He posited that profit is the reward for those willing to bear uncertainty because, unlike risk, uncertainty is defined as inestimable and therefore uninsurable. Uncertainty has been under theoretical examination both in economics and psychology. Whereas economic theories of entrepreneurship focus on explaining what must occur (e.g., uncertainty bearing) for the economy to function, psychological theories try to explain why entrepreneurs are more willing than their counterparts to bear uncertainty. A multi-level definition of uncertainty follows from both of these theoretical perspectives.

The first multi-level definition of uncertainty distinguishes between three distinct types: state, effect, and response (Milliken, 1987). *State uncertainty* is defined as the inability to assign probabilities to the likelihood of events; *effect uncertainty* is defined as

the lack of information about cause–effect relationships; and finally, *response uncertainty* is defined as the inability to predict accurately what the outcomes of a decision might be. Milliken's framework implies that these three types of uncertainty influence individuals in the context of action and should be treated separately.

Uncertainty impacts entrepreneurial action in different ways depending on the type of uncertainty faced by the individual. In a recent empirical test, state uncertainty was, surprisingly, found to be a relatively low impactful hindering factor of entrepreneurial action (McKelvie et al., 2011).³⁴ It is argued here that state uncertainty might not impede entrepreneurial action because entrepreneurs accept it as a given variable in the environment. This also resonates well with the arguments advanced by Sarasvathy et al. (2001) and Dew et al. (2009), as entrepreneurs are seen as individuals who use an effectual logic.

State uncertainty and entrepreneurial action

State uncertainty refers to the "perception by an individual that a particular component of the environment is unpredictable; more specifically, that one does not understand how the components of the environment are changing" (Milliken, 1987, p. 137). As state uncertainty increases, it becomes increasingly difficult to understand and predict the future state of the external environment. This ultimately translates into an aversion toward this type of uncertainty (Ellsberg, 1961) and an impediment to entrepreneurial action (McKelvie et al., 2011). State uncertainty takes the form of doubt, which prevents action by undermining the prospective actor's beliefs. It is detrimental to entrepreneurial action because the individual-level properties that it fuels, such as

³⁴ McKelvie et al. (2011) specifically use the rate of technological change and the rate of demand change as proxies for environmental (state) uncertainty.

hesitancy, indecisiveness, and procrastination, are thought to lead to missed opportunities (McMullen & Shepherd, 2006).

Cognition helps individuals selecting from among available information, ultimately preventing doubts and encouraging action (Mitchell et al., 2007). These cognitive mechanisms include, in particular, biases and heuristics. While cognitive biases refer to "thought processes that involve erroneous inferences or assumptions" (Forbes, 2005, p. 624), heuristics are "rule-of-thumb" decision-making "toolsets" that are "frugal." An individual using such means is able to select pieces of available information and ignore others (Gigerenzer & Goldstein, 1996).

Entrepreneurs are more biased in their decision making than non-entrepreneurs. Specifically, compared to non-founders, entrepreneurs tend to evaluate equivocal business situations more optimistically (Palich & Bagby, 1995), overestimate their ability to make correct predictions (Cooper et al., 1988), overgeneralize from limited information at hand (Busenitz & Barney, 1997; Forbes, 2005; Simon et al., 2000), focus more on their own competencies while neglecting the competitive environment (Moore et al., 2007), select previously chosen alternatives disproportionally more often (i.e., status quo bias; Burmeister & Schade, 2007), and expand their firms despite negative market feedback (i.e., escalation bias; McCarthy et al., 1993).

Information selection and individuals' willingness to bear uncertainty are tightly linked. In particular, individuals' willingness to bear uncertainty—and, consequently, action-driven behavior—is positively influenced by both individual knowledge and motivation (McMullen & Shepherd, 2006). While motivation pertains to the desirability of obtaining possible outcomes, knowledge pertains to the assessment of the feasibility of obtaining such outcomes. As both motivation and knowledge decrease, willingness to bear uncertainty decreases as well. Even so, the question of whether a lack of knowledge about probabilities of obtaining outcomes decreases entrepreneurs' willingness to bear uncertainty remains an open one. This is because entrepreneurs often use effectual reasoning and do not attempt to predict an unknowable future, but actually create their own future through their own actions, knowledge, skills, and available means (Sarasvathy, 2001). Uncertainty may not meaningfully impede entrepreneurial action, because such uncertainty is assumed a priori by entrepreneurs.

Entrepreneurs facing state uncertainty: The framing effect

Entrepreneurs combine desires (utilities, personal values, etc.) and beliefs (expectations, knowledge, etc.) to choose a course of action (Hastie, 2001). Individuals do not behave as choice statistical optimizers (for example, finding the best solution), but rather choose the first option that exceeds an aspiration level (March & Shapira, 1992). Given the uncertainty associated with entrepreneurship, founders must make decisions when they frequently lack adequate information. In particular, this is the case for individuals such as novice entrepreneurs or individuals with entrepreneurial intentions.

I argue that when individuals with entrepreneurial intentions lack information about probabilities, they are less likely than individuals without entrepreneurial intentions to change their aspirations and, ultimately, their behavior. To test this argument, I draw on Tversky and Kahneman's (1981) framing effect—a cognitive bias. The authors showed that individuals exhibit inconsistent preferences depending on how the same opportunity is presented; e.g., in a loss scenario vs. a gain scenario. In this chapter, I extend Tversky and Kahneman's definition of the framing effect in the context of uncertainty. This is done assuming that, for the principle of indifference, a decision maker can assume, given n > 1 possible events that are mutually exclusive and collectively exhaustive, a probability of 1/n to each event. As individuals are uncertainty averse (Ellsberg, 1961), it is expected that they shy away from a prospect that does not contain information about probabilities (state uncertainty) as compared to a prospect within which probabilities are explicitly stated (Knightian risk). However, as entrepreneurs focus their attention on outcomes over probabilities, they differ in this aspect when compared to non-entrepreneurs (Sarasvathy et al., 1998; Zichella, 2016). The hypothesis is that individuals with entrepreneurial intentions will exhibit a greater willingness to bear state uncertainty compared to individuals without entrepreneurial intentions due to their lower sensitivity to information regarding probabilities of success.

H1: The willingness to bear state uncertainty is higher for individuals with entrepreneurial intentions than individuals without such intentions due to a lower sensitivity to the lack of information about probabilities.

DATA AND METHOD

Sample

The sample size and composition is identical between Chapter 2 and Chapter 4, and consists of students with different entrepreneurial intentions. Besides limiting the impact of entrepreneurial experience in explaining the results, using the same sample makes it possible to compare results between the two chapters, ultimately giving a richer perspective on cognition and behavior under risk (Chapter 2) and uncertainty (Chapter 4). The students were enrolled in a general business economics undergraduate program at a major European business school. Although they were enrolled in the same study line, the students presented different entrepreneurial intentions. All students in the study line of general business were offered, at the end of their first semester, the possibility to enter a specialized program designed to address entrepreneurship topics in detail. Interested students were required to apply by handing in a motivation letter. Forty-nine students applied for the program and all were accepted. Each of the 49 students had specified within their motivation letters that they had an active interest in starting a firm, and thereby showed entrepreneurial intentions. Students not applying for the entrepreneurship program represented the population of individuals without entrepreneurial intentions. Overall, the sample allowed for comparing individuals who were similar in several demographics and yet different in their entrepreneurial intentions, thus limiting possible alternative explanations due to the factor of prior entrepreneurial experience.

Subjects from the two populations were invited to sign up for a laboratory experiment. The specific purpose of the experiment—namely, to test differences in cognition and behavior under uncertainty between individuals with different entrepreneurial intentions—was not specified in the sign-up call. Individual emails and a specific website were used for this purpose. Eighteen subjects from the population of individuals with entrepreneurial intentions and 27 from the population of individuals without entrepreneurial intentions signed up, making up a sample of 45 individuals in total. I controlled whether both samples were representative of the respective populations in their demographics (age, gender, nationality) and found no overall differences.³⁵

Table 1 compares the two sample groups in some of their key demographics, providing evidence of small differences between the groups in all but one dimension: willingness to start a new venture within three years. Among entrepreneurship students,

³⁵ Among entrepreneurship students: Age (sample mean 21.5; population mean 21), Female student (sample proportion 22%, population proportion 18%), Nationality/International student (sample proportion 11%, population proportion 15%). Among non-entrepreneurship students: Age (sample mean 21.5; population mean 21), Female student (sample proportion 22%, population proportion 31%), Nationality/International student (sample proportion 6%, population proportion 9%). All tests for differences were non-significant at the 5% level.

66.7 percent (12) intended to start a firm within the next three years, while only 14.8 percent (4) of the non-entrepreneurship student sample had such intentions.³⁶

Insert Table 1 about here

Overall, the data suggest that the sample groups are both comparable and representative of their respective population.

Experiment

In this study, I made use of an experiment based on a real money games experiment. The experiment's general features (e.g., timeline, laboratory rules, and payment structure) are the same as the ones used in Zichella and Reichstein (2016), described in Chapter 2 of this dissertation. In particular, individuals were assigned to computers randomly, communication among students was strictly forbidden, and individuals were individually paid at the end of the experiment.

The subjects were confronted with binary gamble decisions, for a total of 24 decisions. Subjects were not informed about the number of decisions to be made. Gambles presenting different combinations of certainty, risk, and uncertainty were presented to the individuals. In this chapter, only two types are considered (certainty vs. risk and certainty vs. uncertainty), providing a total of 10 unique decisions for individuals to choose from. This provided a total of 450 observations³⁷ (10 choices for 45 individuals) available to the investigation. In the "certainty vs. risk" type, the subjects

³⁶ As a robustness check, we have excluded from the empirical analysis subjects who have mixed entrepreneurial intentions (e.g., not in the entrepreneurial concentration but willing to start a firm in the next three years). The results held overall.

³⁷ These observations are not independent. Standard errors are therefore clustered in the regression analyses.

were presented with options between a certain gain and a risky choice, with an equal chance (50%) of either a greater gain or a smaller gain. The value of the certain gain and the expected value of the risky gamble were identical and kept constant throughout the experiment (at 14 Danish Krone, or \$1.89). The "certainty vs. uncertainty" type was identical to the "certainty vs. risk" type, except that it did not include information about probabilities of obtaining monetary gains, thereby resulting in an uncertain option vis-à-vis a certain one. The combined use of risk and uncertain gambles is a necessary feature for testing the effect of availability of information and makes the findings of Chapter 2 and Chapter 4 complementary.³⁸ Despite the difference in availability of information about probabilities, choices in the "certainty vs. risk" and "certainty vs. uncertainty" gambles can be compared in the experiment, as in both cases a 50% chance probability distribution could be assumed. The 10 different gambles are specified as depicted in Table 2.

Insert Table 2 about here

Throughout the different decision rounds, I controlled for both the prior gain effect and the risk propensity effect (Zichella & Reichstein, 2017) An individual's payoff is attributable to random draws and does not reflect their abilities.

³⁸ As specified in the dissertation introduction, the two chapters are complementary with respect to decisions under risk (Chapter 2) and uncertainty (Chapter 4). The reason behind using risk gambles (together with uncertain gambles) in this chapter lies in the very purpose of the answering the research question: to test differences between groups under uncertainty by manipulating the availability of information.

Main variables

The main dependent variable was a dummy indicating whether the individual in each gamble chose uncertainty as opposed to certainty. Descriptively, the uncertain choice was chosen in 26 percent of the gambles. Individuals with entrepreneurial intentions chose uncertainty for about 27 percent (24 out of 90 decisions) of the gambles. Similarly, individuals without entrepreneurial intentions chose it for about 25 percent (35 out of 135 decisions). These numbers were not significantly different, as shown in Table 3.

Insert Table 3 about here

The main independent variable to test the hypothesis was a dummy indicating group belonging (individuals with and without entrepreneurial intentions). As I compared choices with and without predictive information, I added a second dummy indicating whether individuals chose an option where probabilities were specified (risk) before choosing an option where such probabilities were not specified (uncertainty). The two groups did differ in their choice of risk ($\chi^2 = 4,1446$, *p*-value = 0.042), with individuals with entrepreneurial intentions choosing it for about 44 percent of the gambles (40 out of 90 decisions) versus approximately 31 percent (42 out of 135) for individuals without entrepreneurial intentions.

Controls

To test whether observed differences in choice under uncertainty are due to the same mechanisms presented in Chapter 2, I controlled for prior gains effect by adding a dummy that indicated whether the individual experienced a monetary gain greater than the certain option in the previous gamble. Individuals with entrepreneurial intentions experienced a greater gain than expected about 66 percent (60 out of 90) of the time, while the corresponding number for individuals without entrepreneurial intentions was only 51 percent (69 out of 135). This was significant at a 5% level using a Chi-square test. Individuals with entrepreneurial intentions were "luckier" than individuals without such intentions in their immediate initial risk gambles. Furthermore, I created a variable to control for the degree of risk faced by the subjects in a given choice. As choices were compared in pairs of gambles (certainty vs. risk and certainty vs. uncertainty), and such pairs had different degrees of risk, I controlled for pair number.³⁹ This was strictly exogenously given by the gamble design.

Several personality and demographic factors were used as controls. First, I controlled for the big-five personality traits (John et al., 1991; John et al., 2008; Zhao & Seibert, 2006), which characterize entrepreneurs and are important in choice behavior (e.g., entrepreneurs' higher degree of "openness to experience"). The results of the factor analysis can be found in Table A1 in the Appendix. Second, I controlled for overconfidence, a cognitive bias that encourages risk taking, particularly in entrepreneurs (Busenitz & Barney, 1997; Koellinger et al., 2007). I followed Fischhoff et al.'s (1977) operationalization of overconfidence by checking individuals' level of confidence in answering a series of two-choice questions about health statistics in Denmark (based on World Health Organization 2010 data). Third, pathological gambling was controlled for, as this might have increased individuals' willingness to choose uncertainty (Stinchfield 2000; Winters et al., 1998). A series of five questions were asked about how frequently

³⁹ As shown in Table 2, pair number 3 has the lowest degree of risk, while pair number 1 and 4 present a medium degree of risk. Finally, pair number 2 and 5 present the highest degree of risk.

the subjects gambled (e.g., "gambling in casinos" and "buying lottery tickets"). Finally, a series of demographic variables were controlled for (age, gender, nationality, parental entrepreneurship, income, part-time job) that were proven to be significant in explaining entrepreneurship.

Method

The experiment was designed to predict the binary choices of two groups. For this reason, a logistic regression technique was chosen. Clustered standard errors were used to account for repeated choices by the same subject. The main exogenously inflicted manipulation was the presence of information about probabilities—separating risk from uncertainty. Prior risk choice, prior gains, and degree of risk were included as controls for taking into account their main effects on subsequent choices (as shown in Chapter 2). For prior risk choice, I checked also for a possible interaction effect, as I argue that individuals with entrepreneurial intentions are more likely to bear uncertainty due to their lower sensitivity to the lack of predictive information. To account for the potential bias due to unobservable factors and non-random assignment of individuals, a random effect specification of the logistic regression was included.

RESULTS

Table 4 contains descriptive statistics and correlation coefficients between all considered variables. None of the correlations in Table 4 are of a magnitude causing concern in terms of potential multicollinearity.

Insert Table 4 about here

122

Table 5 illustrates the results of the first regression analysis, where the dependent variable was a dummy indicating the choice between certainty and uncertainty. The first column only includes the control variables and the dummy for group belonging (with and without entrepreneurial intentions). Consistent with prior literature, this regression confirms that individuals with entrepreneurial intentions do not exhibit a lower overall uncertainty aversion compared to individuals without entrepreneurial intentions. The second column reports the results when only including the group dummy and the variable indicating individuals' prior choice in an identical gamble with information about probabilities (prior risk). Column 3 introduces the interaction between the entrepreneurship dummy and prior risk. As this was included, the entrepreneurial intentions dummy became significant at a 10% level, indicating that individuals with entrepreneurial intentions are less likely to choose uncertainty as opposed to certainty in general. Yet Column 3 also reveals that individuals with entrepreneurial intentions are much more likely than individuals without such intentions to choose the uncertain lottery after they have chosen the risk lottery (estimate = 1.675, *p*-value = 0.013). An initial interpretation of this result follows the arguments for the hypothesis; that is, individuals with entrepreneurial intentions do not seem to be affected in their choices by the absence of probability information as much as individuals without such intentions, but rather by monetary outcomes. This result gives early support for the hypothesis: *Entrepreneurs are* less sensitive to the lack of information about probabilities compared to nonentrepreneurs. The results hold in Column 4, which presents a full model with controls. It is noteworthy to mention that while the main negative effect of the "entrepreneurial intentions dummy" was insignificant, the interaction term kept its strength and significance. Furthermore, results were robust after performing a log-likelihood test comparing the interaction model with the full model, as a significant improvement was found in the full model's predictive power (Chi2 = 20.259, *p*-value < 0.000). In the random effect specification of the logistic regression⁴⁰ (Column 5), the interaction coefficient decreased in significance but kept both sign and magnitude.

Insert Table 5 about here

The interaction term reported in Column 4 was investigated further by considering the marginal effect of prior risk on willingness to choose uncertainty. This is depicted in Figure 1. The marginal effect was contrasted between the groups of individuals with and without entrepreneurial intentions. Figure 1 provides further support for the hypothesis, as the marginal effect of prior risk on entrepreneurs' willingness to choose uncertainty was significantly higher for individuals with entrepreneurial intentions. After expressing a preference toward a risky gamble (vis-à-vis a certain gain), individuals with entrepreneurial intentions in their choice preference when dealing with a lack of information on probabilities.

Insert Figure 1 about here

In terms of the control variables, the results tend to be consistent with the predicted associations. Most notably, the results held even when controlling for the two mechanisms identified in Chapter 2 (degree of risk and prior gain). The increasing degree of risk was significant at a 10% level. International students in our sample seemed to be

⁴⁰ We chose to use a random effect specification after controlling for fit with a Hausman test (vs. fixed effects, Chi2 = 6.03, *p*-value = 0.05).

more likely to choose uncertainty than nationals. Also consistent with previous research, it was found that individuals who scored higher on openness to experience and lower on conscientiousness were more prone to choosing uncertainty (Hodson & Sorrentino, 1999). The "pathological gambling" aggregate was negative, which might reflect two possible mechanisms at play. On the one hand, one possibility is that individuals who are habitual gamblers might have an aversion toward gambles where expected values cannot be calculated easily. On the other hand, another possibility is that as habitual gamblers show increased impulsiveness in decision making (Tom et al., 2008), they might reflect a preference for a certain monetary gain.

Overall, the results were confirmed when using a random effects specification, although this was with weakened significances for some of the estimates (Column 5 in Table 5). The random effects specification allowed for a brief look at the inner group correlations in order to assess the nature of the consistency of choices made by the subjects. The rho of the random effect specification was 0.16, indicating a relatively low correlation between the choices of the subjects after controlling for the observables. Table 6 reports some intra-class statistics at the 1st, 25th, 50th, 75th, and 99th percentile of propensity toward choosing uncertainty as opposed to certainty. At the median propensity, the marginal probability of choosing uncertainty was 0.210, while the corresponding number for joint probability of any two given choices was 0.060. These numbers vary greatly across the percentiles, with only a 0.001 joint probability at the 1st percentile of observed propensity, while the corresponding number was 0.631 for observations at the 99th percentile of propensity toward choosing uncertainty. At the median, these numbers suggest that there is a 1.697 times greater chance of someone choosing uncertainty, given that they chose risk previously, than someone who did not choose risk in the previous round.

125

Insert Table 6 about here

Robustness checks

In the first robustness check, the results presented in Table 5 were expanded by redefining the dependent variable. The new dependent variable takes into account all the possible choice combinations between gamble pairs. In each pair of gamble combinations (certainty vs. risk and certainty vs. uncertainty), individuals can exhibit four distinct patterns of choice (certainty, certainty; certainty, uncertainty; risk, certainty; risk, uncertainty). What is of interest is to see how individuals differ in their choice patterns depending on the group they belong to. In order to investigate this issue, a multinomial logistic regression was used with a dependent variable that proxied individual pair choice pattern with four possible values. I assumed that under risk and uncertainty aversion, individuals would prefer a certain option as compared to a risky or uncertain gamble with the same expected value. Therefore, the baseline outcome was the pattern "certainty, certainty." Before interpreting the results, I checked if the IIA assumption was violated by looking at the Suest-based Hausmann Test: The results confirmed the null hypothesis of independent odds for alternatives in the logistic regression. The results are presented in Table 7.

Insert Table 7 about here

At the start is a comparison of mixed preferences—a combination of a certain option with either a risky or uncertain option—with the baseline. Columns 1 and 2 contain results for the combination "certainty, uncertainty," while Columns 3 and 4 contain results for the combination "risk, certainty." The entrepreneurial intentions dummies' estimates were insignificant in both combinations, leaving the researcher uninformed about the existence of differences between groups in their preferences toward mixed strategies as compared to the baseline, "certainty, certainty." The control variable estimates were in line with the findings presented in Table 5, and their interpretations did not change.

In Columns 5 and 6, I compared the combination of choices "risk, uncertainty" with the baseline "certainty, certainty." The results confirmed the findings in Table 5. In particular, entrepreneurs were found to be more consistent in their choices among a risk gamble (with information about probabilities) and an uncertain gamble (without information about probabilities). These results provide further evidence that *individuals with entrepreneurial intentions are less sensitive to the lack of information about probabilities compared to individuals without entrepreneurial intentions*, as they do not shy away from uncertainty but rather choose it more consistently than individuals without entrepreneurial intentions. Finally, control variables' estimates suggest that when choosing risk and uncertainty, other personal factors become important. In particular, lower conscientiousness and higher agreeableness are positively associated with risk taking and acceptance of uncertainty (but are usually associated with a non-entrepreneurial personality; Zhao & Seibert, 2006), and females are less likely to choose consistently between risk and uncertainty.

A possible alternative explanation—the status quo bias—that could possibly have been driving the results was controlled for. People suffering from this bias select previously chosen alternatives disproportionally more often than others (Burmeister & Schade, 2007). To control for this, the presence of streaks was checked for; namely, a pattern of seven or more repeated identical choices out of a total of 10 possible choices.

127

The results within and across groups proved insignificant (Fisher Exact Test, p-value = 0.395).

CONCLUSION

In this chapter, the following research question has been addressed: How do individuals with entrepreneurial intentions differ from individuals without entrepreneurial intentions in their willingness to bear uncertainty? Despite entrepreneurs perceiving themselves as better able to cope with and take decisions under uncertainty than non-entrepreneurs, empirical evidence has been mixed. Motivated by recent calls for understanding the role of contextual factors in uncertainty taking by entrepreneurs (e.g., McKelvie et al., 2011; Shepherd et al., 2015), I explored the role of information about probabilities of success in driving differences between individuals with and without entrepreneurial intentions under state uncertainty.

Cognitive biases and heuristics are an essential part of entrepreneurial decision making under uncertainty. Entrepreneurs face uncertainty in terms of absence of relevant pieces of information in nearly every stage of their business action. Prior literature has shown that entrepreneurs use biases and heuristics to select relevant pieces of information and discard others. However, it is necessary to study the contextual mechanisms that drive decisions in order to address which pieces of information matter the most to entrepreneurs facing uncertainty, thereby permitting controlling for endogeneity, as choices are often influenced by unobservable elements.

A quasi-laboratory experiment was performed that shed light on this matter. The experiment had a number of unique characteristics. First, the results were based on repeated individual decisions with real monetary incentives. This feature is particularly unique, as each choice per individual had a real monetary gain. Second, the presence of

information about probabilities of success was manipulated, leaving other characteristics constant. Third, I compared individuals with different entrepreneurial intentions, thereby controlling for entrepreneurial experience as a possible explanation of the results. Both groups were enrolled in the same study program but had contrasting entrepreneurial intentions. The two groups were very similar in terms of demographics and experience, but were clearly distinct when considering their career focus. Lastly, information on a variety of additional background characteristics—both psychological and attitudinal—was collected. This made it possible to control for a variety of otherwise unobservable characteristics, such as cognitive biases, that affect decision making.

In line with previous studies, individuals with and without entrepreneurial intentions were not found to be significantly different in their general propensity to bear uncertainty (Holm et al., 2013; Koudstaal et al., 2014). However, individuals with entrepreneurial intentions did exhibit a lower sensitivity to the presence of predictive information in comparison to the group without such intentions. I argue that the *framing effect* does not influence the two groups in the same way, as the propensity to choose uncertainty is higher for individuals with entrepreneurial intentions. These results are risky option compared to individuals without entrepreneurial intentions. These results are in tandem with the work by Sarasvathy et al. (2001) and McMullen and Shepherd (2006), which looked at the role of, respectively, information on probabilities and motivation in guiding entrepreneurs' decision making. At the same time, the results of this study contribute to the extant literature by experimentally demonstrating the nature of entrepreneurs' sensitivity to information about probabilities. The results hence provide further specific evidence of the circumstantial nature of entrepreneurs' propensities to choose uncertain monetary opportunities.

Limitations and implications

This chapter presents two important limitations—namely, external validity and a narrow definition of uncertainty. With respect to external validity, I acknowledge that using a sample of individuals with different entrepreneurial intentions provided advantages in terms of comparability of considered groups, but also limited the potential generalizability to entrepreneurs with limited experience (e.g., novice entrepreneurs). With respect to uncertainty and its definition, my study looked only at state uncertainty, one of the three possible types of uncertainty according to Milliken's (1987) categorization. Future research may explore also the role of effect and response uncertainty between entrepreneurs and non-entrepreneurs, giving a comprehensive overview of entrepreneurial behavior under uncertainty.

In this chapter, I tested the impact of information on probabilities on individuals' decision making, ultimately contributing to the understanding of cognitive mechanisms guiding entrepreneurial decision making. This has two direct implications for entrepreneurship research. First, when making decisions, entrepreneurs weigh available information about probabilities less than non-entrepreneurs. In real life, entrepreneurs often face environments and investments that do not provide information regarding their chances of obtaining the monetary objectives they aspire to achieve. The absence of information about their odds of success may lead them to tolerate the uncertainty of starting a new entrepreneurial venture in the pursuit of a monetary return.

The second direct implication of the study's results is related to the stakeholders involved with entrepreneurs. The insights that the findings offer allow investors, business partners, employees, and others to better understand and manage the entrepreneurs with whom they do business. This chapter offers evidence that suggests certain circumstances under which these stakeholders may be alerted to an elevated tendency on the part of the entrepreneur to make choices under uncertainty (i.e., any time the monetary objective is desirable and within reach). This may indeed prove useful to any stakeholder who looks to manage the relationship with their entrepreneurial partner.

APPENDIX

Insert Table A1 about here

REFERENCES

- Amezcua, A. S., Grimes, M. G., Bradley, S. W., & Wiklund, J. (2013). Organizational sponsorship and founding environments: A contingency view on the survival of business-incubated firms, 1994–2007. Academy of Management Journal, 56(6), 1628– 1654.
- Baron, R. A., & Ensley, M. D. (2006). Opportunity recognition as the detection of meaningful patterns: Evidence from comparisons of novice and experienced entrepreneurs. *Management Science*, 52(9), 1331–1344.
- Burmeister, K., & Schade, C. (2007). Are entrepreneurs' decisions more biased? An experimental investigation of the susceptibility to status quo bias. *Journal of Business Venturing*, 22(3), 340–362.
- Busenitz, L. W., & Barney, J. B. (1997). Differences between entrepreneurs and managers in large organizations: Biases and heuristics in strategic decision-making. *Journal of Business Venturing*, 12(1), 9–30.
- Cooper, A. C., Woo, C. Y., & Dunkelberg, W. C. (1988). Entrepreneurs' perceived chances for success. *Journal of Business Venturing*, *3*(2), 97–108.
- Cramer, J. S., Hartog, J., Jonker, N., & Van Praag, C. M. (2002). Low risk aversion encourages the choice for entrepreneurship: an empirical test of a truism. *Journal of Economic Behavior & Organization*, 48(1), 29–36.
- Dew, N., Read, S., Sarasvathy, S. D., & Wiltbank, R. (2009). Effectual versus predictive logics in entrepreneurial decision-making: Differences between experts and novices. *Journal of Business Venturing*, 24(4), 287–309.

- Ellsberg, D. (1961). Risk, ambiguity, and the savage axioms. *The Quarterly Journal of Economics*, 643–669.
- Fischhoff, B., Slovic, P., Lichtenstein, S. (1977). Knowing with certainty: The appropriateness of extreme confidence. *Journal of Experimental Psychology: Human Perception and Performance*, *3*(4), 552.
- Forbes, D. P. (2005). Are some entrepreneurs more overconfident than others? *Journal of Business Venturing*, 20(5), 623–640.
- Gigerenzer, G., & Goldstein, D. G. (1996). Reasoning the fast and frugal way: Models of bounded rationality. *Psychological Review*, 103(4), 650.
- Gneezy, U., & Rustichini, A. (2000). Pay enough or don't pay at all. *Quarterly Journal of Economics*, 791–810.
- Hastie, R. (2001). Problems for judgment and decision making. *Annual Review of Psychology*, 52(1), 653–683.
- Hébert, R. F., & Link, A. N. (1988). *The entrepreneur: Mainstream views & radical critiques*. Praeger Publishers.
- Hodson, G., & Sorrentino, R. M. (1999). Uncertainty orientation and the Big Five personality structure. *Journal of Research in Personality*, *33*(2), 253–261.
- Holm, H. J., Opper, S., & Nee, V. (2013). Entrepreneurs under uncertainty: An economic experiment in China. *Management Science*, 59(7), 1671–1687.
- John, O. P., Donahue, E. M., & Kentle, R. L. (1991). The big five inventory—versions 4a and 54. Berkeley, CA: University of California, Berkeley, Institute of Personality and Social Research.
- John, O. P., Naumann, L. P., & Soto, C. J. (2008). Paradigm shift to the integrative big five trait taxonomy. *Handbook of Personality: Theory and Research*, *3*, 114–158.
- Kahneman, D., & Tversky, A. (1979). Prospect theory: An analysis of decision under risk. *Econometrica: Journal of the Econometric Society*, 263–291.
- Knight, F. H. (1921). Risk, uncertainty and profit. New York: Hart, Schaffner and Marx.
- Koellinger, P., Minniti, M., & Schade, C. (2007). "I think I can, I think I can": Overconfidence and entrepreneurial behavior. *Journal of Economic Psychology*, 28(4), 502–527.
- Koudstaal, M., Sloof, R., & Van Praag, M. (2015). Risk, uncertainty, and entrepreneurship: Evidence from a lab-in-the-field experiment. *Management Science*, 62(10), 2897–2915.
- Krueger, N. F., Reilly, M. D., & Carsrud, A. L. (2000). Competing models of entrepreneurial intentions. *Journal of Business Venturing*, 15(5), 411–432.

Lee, L., Wong, P. K., Der Foo, M., & Leung, A. (2011). Entrepreneurial intentions: The influence of organizational and individual factors. *Journal of Business Venturing*, 26(1), 124–136.

March, J. G. (1994). *Primer on decision making: How decisions happen*. New York (NY) Simon and Schuster.

- March, J. G., & Shapira, Z. (1992). Variable risk preferences and the focus of attention. *Psychological Review*, 99(1), 172.
- Mellers, B., Schwartz, A., & Ritov, I. (1999). Emotion-based choice. *Journal of Experimental Psychology: General*, *128*(3), 332.
- McCarthy, A. M., Schoorman, F. D., & Cooper, A. C. (1993). Reinvestment decisions by entrepreneurs: Rational decision-making or escalation of commitment? *Journal of Business Venturing*, 8(1), 9–24.
- McKelvie, A., Haynie, J. M., & Gustavsson, V. (2011). Unpacking the uncertainty construct: Implications for entrepreneurial action. *Journal of Business Venturing*, 26(3), 273–292.
- McMullen, J. S., & Shepherd, D. A. (2006). Entrepreneurial action and the role of uncertainty in the theory of the entrepreneur. *Academy of Management Review*, *31*(1), 132–152.
- Milliken, F. J. (1987). Three types of perceived uncertainty about the environment: State, effect, and response uncertainty. *Academy of Management Review*, *12*(1), 133–143.
- Mitchell, R. K., Busenitz, L. W., Bird, B., Marie Gaglio, C., McMullen, J. S., Morse, E. A., & Smith, J. B. (2007). The central question in entrepreneurial cognition research 2007. *Entrepreneurship Theory and Practice*, 31(1), 1–27.
- Moore, D. A., Oesch, J. M., & Zietsma, C. (2007). What competition? Myopic self-focus in market-entry decisions. *Organization Science*, *18*(3), 440–454.
- Nordgren, L. F., Van Der Pligt, J., & Van Harreveld, F. (2007). Unpacking perceived control in risk perception: The mediating role of anticipated regret. *Journal of Behavioral Decision Making*, 20(5), 533–544.
- O'Brien, J. P., Folta, T. B., & Johnson, D. R. (2003). A real options perspective on entrepreneurial entry in the face of uncertainty. *Managerial and Decision Economics*, 24(8), 515–533.
- Palich, L. E., & Bagby, D. R. (1995). Using cognitive theory to explain entrepreneurial risktaking: Challenging conventional wisdom. *Journal of Business Venturing*, 10(6), 425– 438.
- Rottenstreich, Y., & Hsee, C. K. (2001). Money, kisses, and electric shocks: On the affective psychology of risk. *Psychological Science*, *12*(3), 185–190.

- Rotter, J. B. (1966). Generalized expectancies for internal versus external control of reinforcement. *Psychological Monographs: General and Applied*, 80(1), 1.
- Sarasvathy, S. D. (2001). Causation and effectuation: Toward a theoretical shift from economic inevitability to entrepreneurial contingency. Academy of Management Review, 26(2), 243–263.
- Sarasvathy, D., Simon, H. A., & Lave, L. (1998). Perceiving and managing business risks: Differences between entrepreneurs and bankers. *Journal of Economic Behavior & Organization*, 33(2), 207–225.
- Shah, A. K., & Oppenheimer, D. M. (2008). Heuristics made easy: An effort-reduction framework. *Psychological Bulletin*, 134(2), 207.
- Shane S., & Venkataraman, S. (2000). The promise of entrepreneurship as a field of research. *Academy of Management Review*, 25(1), 217–226.
- Shane, S. A., & Ulrich, K. T. (2004). 50th anniversary article: Technological innovation, product development, and entrepreneurship in Management Science. *Management Science*, 50(2), 133–144.
- Shepherd, D. A., Williams, T. A., & Patzelt, H. (2015). Thinking about entrepreneurial decision making review and research agenda. *Journal of Management*, *41*(1), 11–46.
- Simon, M., Houghton, S. M., & Aquino, K. (2000). Cognitive biases, risk perception, and venture formation: How individuals decide to start companies. *Journal of Business Venturing*, 15(2), 113–134.
- Stinchfield, R. (2000). Gambling and correlates of gambling among Minnesota public school students. *Journal of Gambling Studies*, *16*(2-3), 153–173.
- Thaler, R. H., & Johnson, E. J. (1990). Gambling with the house money and trying to break even: The effects of prior outcomes on risky choice. *Management Science*, *36*(6), 643–660.
- Tom, S. M., Fox, C. R., Trepel, C., & Poldrack, R. A. (2007). The neural basis of loss aversion in decision-making under risk. *Science*, *315*(5811), 515–518.
- Tversky, A., & Kahneman, D. (1981). The framing of decisions and the psychology of choice. *Science*, *211*(4481), 453–458.
- Tversky, A., & Kahneman, D. (1989). *Rational choice and the framing of decisions* (pp. 81–126). Berlin, Heidelberg: Springer.
- Winters, K. C., Bengston, P., Door, D., & Stinchfield, R. (1998). Prevalence and risk factors of problem gambling among college students. *Psychology of Addictive Behaviors*, 12(2), 127.

- Zellweger, T., Sieger, P., & Halter, F. (2011). Should I stay or should I go? Career choice intentions of students with family business background. *Journal of Business Venturing*, 26(5), 521–536.
- Zichella, G., & Reichstein, T. (2016). "A risk worth taking." Risk as entrepreneurs' conditional choice. Mimeo, CBS.
- Zichella, G. (2016). Entrepreneurs' focus of attention and perception of financial risk. Mimeo, CBS.
- Zhao, H., & Seibert, S. E. (2006). The big five personality dimensions and entrepreneurial status: A meta-analytical review. *Journal of Applied Psychology*, *91*(2), 259.

TABLES

Table 1:Descriptive Statistics and Test of Differences Between Individuals with and without Entrepreneurial Intentions (N = 45)

	With entrepreneurial	Without entrepreneurial	2	
	intention	intention	χ^2	Sig.
Male	14(31.1)	21 (46.7)		
Female	4 (8.9)	6 (13.3)	0.000	
Danish	16(35.6)	26(57.8)		
International	2(4.4)	1(2.2)	0.952	
50.000 < income < 300.000	15(33.3)	24(53.3)		
0 < income < 50.000	3 (6.7)	3 (6.7)	0.313	
Parent entrepreneur	5(11.1)	7(15.6)		
No parent entrepreneur	13(28.9)	$(20 \ (44.4)$	0.019	
University degree	3(6.7)	1(2.2)		
No University degree	15 (33.3)	26 (57.8)	2.588	
Part-time employed	11 (24.4)	23(51.1)		
Not part-time employed	7 (15.6)	4 (8.9)	3.389	*
Established firm	3(6.7)	1(2.2)		
Not established firm	15 (33.3)	26 (57.8)	2.241	
Establish firm within 3 years	12(26.7)	4(8.9)		
Not establish firm within 3 years	6 (13.3)	23 (51.1)	12.672	***
* $p < 0.10$, ** $p < 0.05$, *** $p < 0.01$, percent	ntages of table total in parentheses	3		

		Optio	on A	Option B	Feedback Prior Gain
	Decision 1	100%	14 DKr	50% 4 DKr 50% 24 DKr	No
Pair 1	Decision 2	100%	14 DKr	Unknown chance of getting 4 DKr Unknown chance of getting 24 DKr	No
Pair 2	Decision 3	100%	14 DKr	50%0 DKr50%28 DKr	No
Pair 2	Decision 4	100%	14 DKr	Unknown chance of getting 0 DKr Unknown chance of getting 28 DKr	No
Pair 3	Decision 5	100%	14 DKr	50%8 DKr50%20 DKr	Yes
	Decision 6	100%	14 DKr	Unknown chance of getting 8 DKr Unknown chance of getting 20 DKr	Yes
Pair /	Decision 7	100%	14 DKr	50%4 DKr50%24 DKr	Yes
Pair 4	Decision 8	100%	14 DKr	Unknown chance of getting 4 DKr Unknown chance of getting 24 DKr	Yes
Pair 5	Decision 9	100%	14 DKr	50%0 DKr50%28 DKr	Yes
	Decision 10	100%	14 DKr	Unknown chance of getting 0 DKr Unknown chance of getting 28 DKr	Yes

Table 2: Considering 10 Real Monetary Games (in Six Pairs) Between Certainty (OptionA) and Risk or Uncertainty (Option B). 1Dkr = \$0.16, Expected Gain = \$2.2 of Decision

Table 3: Descriptive Statistics on Key Variables (N = 225)

	With entrepreneurial intentions	Without entrepreneurial intentions	Total	χ²	Significance
	24 (40 7)		50 (26 2)		
Choose Uncertainty (Option B)	24 (10.7)	35 (15.6)	59 (26.3)		
Choose Certainty (Option A)	66 (29.3)	100 (44.4)	166 (73.7)	0.0153	
Prior gain greater than expected	60 (26.6)	69 (30.6)	129 (57.4)		
Prior gain not greater than expected	30 (13.4)	66 (29.4)	96 (42.6)	5.3416	**
Total	90 (40.0)	135 (60.0)	225 (100.0)		

p < 0.10, ** p < 0.05, *** p < 0.01, percentages of table total in parentheses

Table 4: Descriptive Statistics and Pearson's Correlation Coefficients

		Mean	S.D.	Min	Max	[1]	[2]	[3]	[4]	[5]	[6]
						[-]	<u> </u>	[*]		[-]	[*]
[1]	Entrepreneurial Intention	0.40	0.49	0	1						
[2]	Prior gain (monetary)	0.13	0.34	0	1	0.1712					
[3]	Prior gain (presence)	0.66	0.47	0	1	-0.0000	-0.7055				
[4]	Degree of Risk (pair)	1.2	0.75	0	2	0.0000	0.2551	-0.3273			
[5]	Extraversion	0.06	0.92	-3.07	1.46	0.1333	0.0074	0.0000	-0.0000		
[6]	Openness	0.08	0.88	-2.14	2.52	0.1556	0.0137	0.0000	0.0000	-0.0049	
[7]	Neuroticism	-0.03	0.87	-1.90	1.86	-0.1205	-0.0320	0.0000	-0.0000	-0.0754	-0.0334
[8]	Conscientiousness	0.06	0.91	-2.87	1.60	0.3524	0.0032	-0.0000	-0.0000	0.0806	-0.0236
[9]	Agreebleness	0	0.93	-1.98	1.80	0.2007	0.0602	-0.0000	0.0000	0.0291	-0.0315
[10]	Overconfidence	101.25	21.72	67	148	-0.1368	-0.0043	0.0000	-0.0000	0.1037	0.0816
[11]	Pathological Aggregate	1.52	0.56	1	3.20	-0.0715	-0.0208	0.0000	-0.0000	0.0007	-0.2302
[12]	Age	21.51	0.78	20	23	-0.0700	-0.0299	0.0000	-0.0000	-0.0170	0.1560
[13]	Female	0.22	0.42	0	1	-0.0000	0.0038	0.0000	0.0000	0.2078	0.2075
[14]	International	0.07	0.25	0	1	0.1455	-0.0081	0.0000	0.0000	0.0674	0.0236
[15]	Parent Entrepreneur	0.27	0.44	0	1	0.0205	-0.0121	-0.0000	0.0000	0.0499	-0.0411
[16]	Part time employed	0.76	0.42	0	1	-0.2744	-0.0258	0.0000	-0.0000	0.1221	-0.0462
[17]	Income above 50.000	0.87	0.34	0	1	-0.0801	-0.0200	0.0000	0.0000	-0.0015	-0.1855
		[7]	[8]	[9]	[10]	[11]	[12]	[13]	[14]	[15]	[16]
101	Constitution	0.0072									
[0] [0]	Agraeblenass	0.0072	0.0206								
[7]	Overconfidence	-0.0160	0.0200	0.0570							
[10]	Pathological Aggregate	-0.2210	-0.1521	0.0570	0.2690						
[11]	A op	-0.1316	-0.0403	-0 1299	-0.0314	-0 0140					
[12]	Female	0 2817	0.0120	-0.1299	-0.0294	-0.3749	0 1968				
[13]	International	0.1133	0.2159	0.2743	0.0354	-0.1172	0.0417	0.1917			
[15]	Parent Entrepreneur	0.0916	0.0189	0.0680	-0.0303	-0.0641	-0.1361	-0.0314	-0.0895		
[16]	Part time employed	-0.0551	-0.1901	-0.2687	0.1940	0.1219	-0.0601	0.0314	-0.2217	-0.0798	
[17]	Income above 50.000	-0.0600	-0.0025	-0.1411	0.0856	0.2237	0.0254	-0.2524	-0.4020	0.1080	0.2863
			-		-				-	-	-

Note: Correlation coefficients above 0.065 are significant at a 5% level

		RE Logit			
	Baseline	Baseline Explanatory Interaction			
	Model	Model	Model	Full Model	
Finlanatory Variables					
Entrepreneurial intention	0.310	-0.035	-0.815*	-0.454	-0.454
	[0.436]	-0.035 [0 304]	-0.015 [0.471]	[0 592]	-0. 4 54
Risk Prior	[0.450]	0.537**	-0.162	-0.276	-0.0305
KISK I HOI		[0.305]	-0.102 [0.432]	[0,522]	-0.0303
Entropropeurial intention v Disk Prior		[0.505]	1 675**	[0.322]	[0.570]
Entrepreneurial intention x Kisk i nor			1.075	[0 791]	I.703** [0.884]
Control Variables			[0.234]	[0.771]	[0.004]
Prior Gain (Monetery)	-0.105			-0.447	-0.614
Thor Gain (Monetary)	-0.105			-0.447	-0.014
Drive Coin (Dresserve)	[0.398]			[0.794]	[0.691]
Prior Gain (Presence)	0.933***			-0.435	-0.308
	[0.464]			[0.424]	[0.439]
Degree of Risk (pair)	0.321			0.400*	0.470*
	[0.220]			[0.238]	[0.262]
Extraversion	-0.086			-0.0371	-0.0201
0	[0.221]			[0.229]	[0.270]
Openness	0.414**			0.378**	0.445*
Numericia	[0.190]			[0.190]	[0.262]
Neuroticism	-0.134			-0.0901	-0.0816
	[0.158]			[0.160]	[0.278]
Conscentiousness	-0.673***			-0.576***	-0.645**
	[0.192]			[0.189]	[0.264]
Agreebleness	0.195			0.185	0.192
	[0.190]			[0.187]	[0.239]
Overconfidence	0.018			0.0186	0.0199
	[0.023]			[0.0247]	[0.0269]
Pathological Aggregate	-0.893*			-0.873*	-0.939
	[0.463]			[0.467]	[0.574]
Age	0.243			0.193	0.240
	[0.270]			[0.283]	[0.367]
Female	-0.812			-0.825	-0.967
	[0.622]			[0.649]	[0.880]
International	2.245**			1.867**	2.126*
	[0.928]			[0.867]	-1204
Parent Entrepreneur	0.236			0.178	0.112
	[0.383]			[0.380]	[0.537]
Part time employed	0.445			0.472	0.556
	[0.514]			[0.504]	[0.665]
Income above 50.000	0.507			0.427	0.471
_	[0.689]			[0.713]	[0.833]
Constant	-8.081	-1.231***	-1.000	-6.374	-7.630
	[5.851]	[0./51]	[0.234]	[6.319]	[8.009]
Number of observations	225	225	225	225	225
Log Likelihood	-115.9824	-127.9790	-124.7084	-109.5551	-108.3070
γ^2	34,91***	3.13	8.91**	37.95***	22.89
LR test (RE vs. Logistic regression)					2.50*

Table 5: Explaining Subjects' Choices Under Uncertainty After Risk, Logit Regressions

Robust standard errors in parentheses. *** p<0.01, ** p<0.05, * p<0.1

Measure	1st Percentile	25th Percentile	50th Percentile	75th Percentile	99th Percentile
Marginal Prob.	.02425	.126013	.210552	.355698	.783894
Joint Prob.	.001046	.02363	.060115	.153578	.63079
Odds ratio	1.85103	1.73944	1.69741	1.66229	1.69539
Pearson's r	.019365	.070378	.094954	.118059	.096222
Yule's Q	.298498	.269925	.258549	.248766	.257991

Table 6: Measures of Intra-class Associations in Random-effects Logit Regression
Multinomial Logit Model (Dep. Var: Choice Pairs. Base Pair: Certainty, Certainty)									
	Pair 1 Certainty, Uncertainty		Pai Risk, Ce	Pair 2 Risk, Certainty		Pair 3 Risk, Uncertainty			
	Explanatory Model	Full Model	Explanatory Model	Full Model	Explanator y Model	Full Model			
Explanatory Variables									
Entrepreneurial intentions	-0.815	-0.689	0.128	0.269	0.989*	1.311**			
I I I I I I I I I I I I I I I I I I I	[0.579]	[0.751]	[0.490]	[0.557]	[0.543]	[0.636]			
Control Variables									
Prior Gain (Monetary)		0.444		0.138		0.0171			
		[0.298]		[0.279]		[0.371]			
Prior Gain (Presence)		0.0635		0.977		2.184*			
		[0.915]		[0.919]		[1.258]			
Degree of Risk (pair)		0.433		-0.460*		-0.380			
-		[0.297]		[0.268]		[0.381]			
Extraversion		-0.227		0.107		0.340			
		[0.270]		[0.252]		[0.367]			
Openness		0.578*		0.173		0.542**			
		[0.309]		[0.271]		[0.226]			
Neuroticism		-0.150		-0.520**		-0.630**			
		[0.296]		[0.262]		[0.274]			
Conscentiousness		-0.529**		-0.306		- 0.979***			
		[0.256]		[0.302]		[0.332]			
Agreebleness		-0.126		0.537**		1.082***			
-		[0.294]		[0.276]		[0.304]			
Overconfidence		0.00886		0.0675** *		0.0714*			
		[0.0346]		[0.0241]		[0.0440]			
Pathological Aggregate		-0.542		- 1.734***		-2.573**			
		[0.581]		[0.612]		-1.178			
Age		0.542		0.478		0.254			
-		[0.363]		[0.313]		[0.325]			
Female		-0.622		-1.330		-2.649**			
		[0.868]		[0.838]		-1.332			
International		0.994		0.381		3.160***			
		-1.641		-1.471		[0.834]			
Parent Entrepreneur		1.017		0.756		0.311			
		[0.656]		[0.512]		[0.455]			
Part time employed		0.186		0.492		1.282**			
		[0.684]		[0.737]		[0.637]			
Income above 50.000		0.415		-0.747		-0.328			
~		[0.838]		[0.669]		[0.744]			
Constant	-1.001***	-15.23**	-0.754***	-12.35*	-1.917***	-9.982			
	[0.318]	[-7.621]	[0.286]	[-6.507]	[0.335]	[-7.160]			
Number of observations	225	225	225	225	225	225			
Loglikelihood	-270.233	-223.410	-270.233	-223.410	-270.233	-223.410			
γ ²	7 046*	100 72***	7 046*	100.72**	7 046*	100.72**			
~			/.010	*	7.010	*			

Table 7: Explaining Individuals' Choice Combinations Across Pairs, Results ofMultinomial Logit Regressions

Clustered standard errors (unique ID for individuals) in parentheses. *** p<0.01, ** p<0.05, * p<0.1

FIGURES



Figure 1: Contrasting the marginal effect of individuals with entrepreneurial intentions (entrepreneurs) against individuals without entrepreneurial intentions (non-entrepreneurs) with respect to uncertainty by prior risk choice.

Table A1

Results of Factor Analysis Generating Big-5 Personality Traits

Question	Extraversion	Openness	Neroticism	Conscien– tiousness	Agreeable– ness	As expected
I am someone who						
is talkative	0.7915					\checkmark
tends to find faults with others					0.4307	, V
does a thorough job						•
is depressed, blue	-0.5399					
is original, comes up with new ideas	0.4043	0.5208				
is reserved	0.6932					
is helpful and unselfish with others					0.5165	, V
can be somewhat careless					0.4685	•
is relaxed, handles stress well			0.7396			\checkmark
is curious about many different things		0.6015				, V
is full of energy	0.6498					
starts quarrels with others					0.4676	v V
is a reliable worker						v
can be tense			0.6384			
is ingenious, a deep thinker		0.5641				, V
generates a lot of enthusiasm	0.8022					
has a forgiving nature				-0.5251	0.4328	v V
tends to be disorganized				0.7063		, V
worries a lot			0.7265			, V
has an active imagination		0.4028				, v
tends to be quiet	0.8135					
is generally trusting						v
tends to be lazy				0.5232		\checkmark
is emotionally stable, not easily upset			0.7654			v V
is inventive		0.4512				, v
has an assertive personality						•
can be cold and aloof					0.5797	
perseveres until the task is finished				0.6081		v V
can be moody			0.5442			v V
values artistic, aesthetic experiences		0.4505				, v
is sometimes shy, inhibited	0.7268					v
is considerate and kind to almost everyone					0.5876	v V
does things efficiently				0.6183		v.
remains calm in tense situations			0.6463			v V
prefers work that is routine		0.5409				v V
is outgoing, sociable	0.7978					v
is sometimes rude to others					0.7959	v
makes plans and follows through with them				0.8054		v
gets nervous easily			0.4898	010000		v
likes to reflect, play with ideas		0.6456	0.2000			v V
has few artistic interests		0				v
likes to cooperate with others	0.4740					
is easily distracted						
is sophisticated in art, music, or literature		0.6223				\checkmark

Note: Factor loading above/below 0.4/-0.4 reported -35 out of 44 load as expected

CHAPTER 5, CONCLUSIONS

This PhD dissertation addresses the themes of entrepreneurial cognition and behavior in situations of risk and uncertainty using an experimental methodology. This approach advances our understanding of the causal link between cognition and behavior and its implications for individual entrepreneurs and their stakeholders. Despite simplifying reality, an experimental approach exposes individuals to features typical of a business environment. My experiments focused on financial risk by incorporating two essential features that describe an investment opportunity: namely, monetary returns and the probabilities of obtaining them.

This dissertation has explicitly considered that individuals with and without entrepreneurial intentions do not differ in their general preferences towards risk and uncertainty. On the contrary, the dissertation has argued for more nuanced differences between the two groups, looking at both the contextual and the subjective mechanisms that positively affect individuals' willingness to bear risk and uncertainty.

The three essays focus on different mechanisms of risk and uncertainty, giving a complementary overview of entrepreneurial cognition. Chapter 2 explores individuals' sensitivity to two heuristics—the prior gain effect and the risk propensity effect—when choosing between a monetary lottery and a gain. The results of this essay show that individuals with entrepreneurial intentions choose risk more often than individuals without entrepreneurial intentions immediately after a positive gain, thereby illustrating a greater sensitivity to the prior gain effect. Yet, entrepreneurs do not take risk at all costs, since they shy away from high degrees of risk.

Chapter 3 examines how individuals with and without entrepreneurial intentions differ in their perceptions of risk, arguing that the focus of attention plays a prominent role in explaining such differences. The essay compares the two groups' choices, foci, and perceptions of risky

investment opportunities. The results reveal that individuals with entrepreneurial intentions are more likely than comparable others to focus on monetary outcomes rather than probabilities. Furthermore, individuals with entrepreneurial intentions are less likely than comparable others to perceive risk consistently with their focus of attention.

Finally, Chapter 4 explores individuals' choices under uncertainty by investigating how predictive information affects the behaviors groups of individuals with different entrepreneurial intentions differently. Individuals with entrepreneurial intentions are found to choose consistently between two prospects that share everything in common but information about probabilities, thereby exhibiting a lower sensitivity to a lack of predictive information than individuals without entrepreneurial intentions.

This dissertation develops around two growing streams of research in entrepreneurship. On one hand, the research on entrepreneurial decision-making is useful for exploring actual behavior in situations of risk and uncertainty. Our findings, in line with prior research, do not confirm that entrepreneurs have a greater overall propensity for either risk or uncertainty. On the other hand, the research stream on entrepreneurial cognition is useful for exploring the antecedents of entrepreneurs' behaviors. My three essays contribute to the extant literature by providing empirical evidence of the more nuanced differences between entrepreneurs' and nonentrepreneurs' behaviors and cognition in situations of risk and uncertainty.

My results also speak to practice by showing how individuals with different entrepreneurial intentions exhibit unique cognition and behaviors in situations of risk and uncertainty. Stakeholders working with novice or prospective entrepreneurs (e.g. incubators, accelerators) must consider entrepreneurs' greater sensitivity to monetary gains (Chapter 2) and lower sensitivity to the presence of predictive information (Chapter 3 and 4) in order to carefully align their assessments of business opportunities with the entrepreneurs' assessments.

TITLER I PH.D.SERIEN:

2004

- 1. Martin Grieger Internet-based Electronic Marketplaces and Supply Chain Management
- 2. Thomas Basbøll LIKENESS A Philosophical Investigation
- 3. Morten Knudsen Beslutningens vaklen En systemteoretisk analyse of moderniseringen af et amtskommunalt sundhedsvæsen 1980-2000
- 4. Lars Bo Jeppesen Organizing Consumer Innovation A product development strategy that is based on online communities and allows some firms to benefit from a distributed process of innovation by consumers
- 5. Barbara Dragsted SEGMENTATION IN TRANSLATION AND TRANSLATION MEMORY SYSTEMS An empirical investigation of cognitive segmentation and effects of integrating a TM system into the translation process
- 6. Jeanet Hardis Sociale partnerskaber Et socialkonstruktivistisk casestudie af partnerskabsaktørers virkelighedsopfattelse mellem identitet og legitimitet
- 7. Henriette Hallberg Thygesen System Dynamics in Action
- 8. Carsten Mejer Plath Strategisk Økonomistyring
- 9. Annemette Kjærgaard Knowledge Management as Internal Corporate Venturing

– a Field Study of the Rise and Fall of a Bottom-Up Process

- 10. Knut Arne Hovdal De profesjonelle i endring Norsk ph.d., ej til salg gennem Samfundslitteratur
- Søren Jeppesen Environmental Practices and Greening Strategies in Small Manufacturing Enterprises in South Africa – A Critical Realist Approach
- 12. Lars Frode Frederiksen Industriel forskningsledelse – på sporet af mønstre og samarbejde i danske forskningsintensive virksomheder
- 13. Martin Jes Iversen
 The Governance of GN Great Nordic
 in an age of strategic and structural transitions 1939-1988
- 14. Lars Pynt Andersen The Rhetorical Strategies of Danish TV Advertising A study of the first fifteen years with special emphasis on genre and irony
- 15. Jakob Rasmussen Business Perspectives on E-learning
- Sof Thrane
 The Social and Economic Dynamics of Networks
 – a Weberian Analysis of Three
 Formalised Horizontal Networks
- 17. Lene Nielsen Engaging Personas and Narrative Scenarios – a study on how a usercentered approach influenced the perception of the design process in the e-business group at AstraZeneca
- S.J Valstad
 Organisationsidentitet
 Norsk ph.d., ej til salg gennem
 Samfundslitteratur

- 19. Thomas Lyse Hansen Six Essays on Pricing and Weather risk in Energy Markets
- 20. Sabine Madsen Emerging Methods – An Interpretive Study of ISD Methods in Practice
- 21. Evis Sinani The Impact of Foreign Direct Investment on Efficiency, Productivity Growth and Trade: An Empirical Investigation
- 22. Bent Meier Sørensen Making Events Work Or, How to Multiply Your Crisis
- 23. Pernille Schnoor Brand Ethos Om troværdige brand- og virksomhedsidentiteter i et retorisk og diskursteoretisk perspektiv
- 24. Sidsel Fabech Von welchem Österreich ist hier die Rede? Diskursive forhandlinger og magtkampe mellem rivaliserende nationale identitetskonstruktioner i østrigske pressediskurser
- 25. Klavs Odgaard Christensen Sprogpolitik og identitetsdannelse i flersprogede forbundsstater Et komparativt studie af Schweiz og Canada
- 26. Dana B. Minbaeva Human Resource Practices and Knowledge Transfer in Multinational Corporations
- 27. Holger Højlund Markedets politiske fornuft Et studie af velfærdens organisering i perioden 1990-2003
- 28. Christine Mølgaard Frandsen A.s erfaring Om mellemværendets praktik i en

transformation af mennesket og subjektiviteten

29. Sine Nørholm Just
The Constitution of Meaning
– A Meaningful Constitution?
Legitimacy, identity, and public opinion in the debate on the future of Europe

- 1. Claus J. Varnes Managing product innovation through rules – The role of formal and structured methods in product development
- Helle Hedegaard Hein Mellem konflikt og konsensus

 Dialogudvikling på hospitalsklinikker
- 3. Axel Rosenø Customer Value Driven Product Innovation – A Study of Market Learning in New Product Development
- 4. Søren Buhl Pedersen Making space An outline of place branding
- 5. Camilla Funck Ellehave Differences that Matter An analysis of practices of gender and organizing in contemporary workplaces
- 6. Rigmor Madeleine Lond Styring af kommunale forvaltninger
- 7. Mette Aagaard Andreassen Supply Chain versus Supply Chain Benchmarking as a Means to Managing Supply Chains
- 8. Caroline Aggestam-Pontoppidan From an idea to a standard The UN and the global governance of accountants' competence
- 9. Norsk ph.d.
- 10. Vivienne Heng Ker-ni An Experimental Field Study on the

Effectiveness of Grocer Media Advertising Measuring Ad Recall and Recognition, Purchase Intentions and Short-Term Sales

- 11. Allan Mortensen Essays on the Pricing of Corporate Bonds and Credit Derivatives
- 12. Remo Stefano Chiari Figure che fanno conoscere Itinerario sull'idea del valore cognitivo e espressivo della metafora e di altri tropi da Aristotele e da Vico fino al cognitivismo contemporaneo
- 13. Anders McIlquham-Schmidt Strategic Planning and Corporate Performance An integrative research review and a meta-analysis of the strategic planning and corporate performance literature from 1956 to 2003
- 14. Jens Geersbro The TDF – PMI Case Making Sense of the Dynamics of Business Relationships and Networks
- 15 Mette Andersen Corporate Social Responsibility in Global Supply Chains Understanding the uniqueness of firm behaviour
- 16. Eva Boxenbaum Institutional Genesis: Micro – Dynamic Foundations of Institutional Change
- 17. Peter Lund-Thomsen Capacity Development, Environmental Justice NGOs, and Governance: The Case of South Africa
- 18. Signe Jarlov Konstruktioner af offentlig ledelse
- 19. Lars Stæhr Jensen Vocabulary Knowledge and Listening Comprehension in English as a Foreign Language

An empirical study employing data elicited from Danish EFL learners

- 20. Christian Nielsen Essays on Business Reporting Production and consumption of strategic information in the market for information
- 21. Marianne Thejls Fischer Egos and Ethics of Management Consultants
- 22. Annie Bekke Kjær Performance management i Procesinnovation – belyst i et social-konstruktivistisk perspektiv
- 23. Suzanne Dee Pedersen GENTAGELSENS METAMORFOSE Om organisering af den kreative gøren i den kunstneriske arbejdspraksis
- 24. Benedikte Dorte Rosenbrink Revenue Management Økonomiske, konkurrencemæssige & organisatoriske konsekvenser
- 25. Thomas Riise Johansen Written Accounts and Verbal Accounts The Danish Case of Accounting and Accountability to Employees
- 26. Ann Fogelgren-Pedersen The Mobile Internet: Pioneering Users' Adoption Decisions
- 27. Birgitte Rasmussen Ledelse i fællesskab – de tillidsvalgtes fornyende rolle
- 28. Gitte Thit Nielsen *Remerger* skabende ledelseskræfter i fusion og opkøb
- 29. Carmine Gioia A MICROECONOMETRIC ANALYSIS OF MERGERS AND ACQUISITIONS

- 30. Ole Hinz Den effektive forandringsleder: pilot, pædagog eller politiker? Et studie i arbejdslederes meningstilskrivninger i forbindelse med vellykket gennemførelse af ledelsesinitierede forandringsprojekter
- 31. Kjell-Åge Gotvassli Et praksisbasert perspektiv på dynamiske læringsnettverk i toppidretten Norsk ph.d., ej til salg gennem Samfundslitteratur
- 32. Henriette Langstrup Nielsen Linking Healthcare An inquiry into the changing performances of web-based technology for asthma monitoring
- 33. Karin Tweddell Levinsen Virtuel Uddannelsespraksis Master i IKT og Læring – et casestudie i hvordan proaktiv proceshåndtering kan forbedre praksis i virtuelle læringsmiljøer
- 34. Anika Liversage Finding a Path Labour Market Life Stories of Immigrant Professionals
- 35. Kasper Elmquist Jørgensen Studier i samspillet mellem stat og erhvervsliv i Danmark under 1. verdenskrig
- 36. Finn Janning A DIFFERENT STORY Seduction, Conquest and Discovery
- 37. Patricia Ann Plackett Strategic Management of the Radical Innovation Process Leveraging Social Capital for Market Uncertainty Management

1. Christian Vintergaard Early Phases of Corporate Venturing

- 2. Niels Rom-Poulsen Essays in Computational Finance
- 3. Tina Brandt Husman Organisational Capabilities, Competitive Advantage & Project-Based Organisations The Case of Advertising and Creative Good Production
- Mette Rosenkrands Johansen
 Practice at the top
 how top managers mobilise and use
 non-financial performance measures
- 5. Eva Parum Corporate governance som strategisk kommunikations- og ledelsesværktøj
- 6. Susan Aagaard Petersen Culture's Influence on Performance Management: The Case of a Danish Company in China
- 7. Thomas Nicolai Pedersen The Discursive Constitution of Organizational Governance – Between unity and differentiation The Case of the governance of environmental risks by World Bank environmental staff
- 8. Cynthia Selin Volatile Visions: Transactons in Anticipatory Knowledge
- 9. Jesper Banghøj Financial Accounting Information and Compensation in Danish Companies
- 10. Mikkel Lucas Overby Strategic Alliances in Emerging High-Tech Markets: What's the Difference and does it Matter?
- 11. Tine Aage External Information Acquisition of Industrial Districts and the Impact of Different Knowledge Creation Dimensions

A case study of the Fashion and Design Branch of the Industrial District of Montebelluna, NE Italy

- 12. Mikkel Flyverbom Making the Global Information Society Governable On the Governmentality of Multi-Stakeholder Networks
- 13. Anette Grønning Personen bag Tilstedevær i e-mail som interaktionsform mellem kunde og medarbejder i dansk forsikringskontekst
- 14. Jørn Helder One Company – One Language? The NN-case
- 15. Lars Bjerregaard Mikkelsen Differing perceptions of customer value Development and application of a tool for mapping perceptions of customer value at both ends of customer-supplier dyads in industrial markets
- 16. Lise Granerud Exploring Learning Technological learning within small manufacturers in South Africa
- 17. Esben Rahbek Pedersen Between Hopes and Realities: Reflections on the Promises and Practices of Corporate Social Responsibility (CSR)
- Ramona Samson The Cultural Integration Model and European Transformation. The Case of Romania

2007

1. Jakob Vestergaard Discipline in The Global Economy Panopticism and the Post-Washington Consensus

- 2. Heidi Lund Hansen Spaces for learning and working A qualitative study of change of work, management, vehicles of power and social practices in open offices
- 3. Sudhanshu Rai Exploring the internal dynamics of software development teams during user analysis A tension enabled Institutionalization Model; "Where process becomes the objective"
- 4. Norsk ph.d. Ej til salg gennem Samfundslitteratur
- 5. Serden Ozcan *EXPLORING HETEROGENEITY IN ORGANIZATIONAL ACTIONS AND OUTCOMES A Behavioural Perspective*
- 6. Kim Sundtoft Hald Inter-organizational Performance Measurement and Management in Action

 An Ethnography on the Construction of Management, Identity and Relationships
- 7. Tobias Lindeberg Evaluative Technologies Quality and the Multiplicity of Performance
- 8. Merete Wedell-Wedellsborg Den globale soldat Identitetsdannelse og identitetsledelse i multinationale militære organisationer
- Lars Frederiksen Open Innovation Business Models Innovation in firm-hosted online user communities and inter-firm project ventures in the music industry – A collection of essays
- 10. Jonas Gabrielsen Retorisk toposlære – fra statisk 'sted' til persuasiv aktivitet

- Christian Moldt-Jørgensen Fra meningsløs til meningsfuld evaluering. Anvendelsen af studentertilfredshedsmålinger på de korte og mellemlange videregående uddannelser set fra et psykodynamisk systemperspektiv
- 12. Ping Gao Extending the application of actor-network theory Cases of innovation in the telecommunications industry
- Peter Mejlby Frihed og fængsel, en del af den samme drøm? Et phronetisk baseret casestudie af frigørelsens og kontrollens sameksistens i værdibaseret ledelse!
- 14. Kristina Birch Statistical Modelling in Marketing
- 15. Signe Poulsen Sense and sensibility: The language of emotional appeals in insurance marketing
- 16. Anders Bjerre Trolle Essays on derivatives pricing and dynamic asset allocation
- 17. Peter Feldhütter Empirical Studies of Bond and Credit Markets
- 18. Jens Henrik Eggert Christensen Default and Recovery Risk Modeling and Estimation
- Maria Theresa Larsen Academic Enterprise: A New Mission for Universities or a Contradiction in Terms? Four papers on the long-term implications of increasing industry involvement and commercialization in academia

- 20. Morten Wellendorf Postimplementering af teknologi i den offentlige forvaltning Analyser af en organisations kontinuerlige arbejde med informationsteknologi
- 21. Ekaterina Mhaanna Concept Relations for Terminological Process Analysis
- 22. Stefan Ring Thorbjørnsen Forsvaret i forandring Et studie i officerers kapabiliteter under påvirkning af omverdenens forandringspres mod øget styring og læring
- 23. Christa Breum Amhøj Det selvskabte medlemskab om managementstaten, dens styringsteknologier og indbyggere
- 24. Karoline Bromose Between Technological Turbulence and Operational Stability

 An empirical case study of corporate venturing in TDC
- 25. Susanne Justesen Navigating the Paradoxes of Diversity in Innovation Practice

 A Longitudinal study of six very different innovation processes – in practice
- 26. Luise Noring Henler Conceptualising successful supply chain partnerships

 Viewing supply chain partnerships from an organisational culture perspective
- 27. Mark Mau Kampen om telefonen Det danske telefonvæsen under den tyske besættelse 1940-45
- 28. Jakob Halskov The semiautomatic expansion of existing terminological ontologies using knowledge patterns discovered

on the WWW – an implementation and evaluation

- 29. Gergana Koleva European Policy Instruments Beyond Networks and Structure: The Innovative Medicines Initiative
- 30. Christian Geisler Asmussen Global Strategy and International Diversity: A Double-Edged Sword?
- 31. Christina Holm-Petersen Stolthed og fordom Kultur- og identitetsarbejde ved skabelsen af en ny sengeafdeling gennem fusion
- 32. Hans Peter Olsen Hybrid Governance of Standardized States Causes and Contours of the Global Regulation of Government Auditing
- 33. Lars Bøge Sørensen Risk Management in the Supply Chain
- 34. Peter Aagaard Det unikkes dynamikker De institutionelle mulighedsbetingelser bag den individuelle udforskning i professionelt og frivilligt arbejde
- 35. Yun Mi Antorini Brand Community Innovation An Intrinsic Case Study of the Adult Fans of LEGO Community
- 36. Joachim Lynggaard Boll Labor Related Corporate Social Performance in Denmark Organizational and Institutional Perspectives

- 1. Frederik Christian Vinten Essays on Private Equity
- 2. Jesper Clement Visual Influence of Packaging Design on In-Store Buying Decisions

- Marius Brostrøm Kousgaard Tid til kvalitetsmåling?

 Studier af indrulleringsprocesser i forbindelse med introduktionen af kliniske kvalitetsdatabaser i speciallægepraksissektoren
- 4. Irene Skovgaard Smith Management Consulting in Action Value creation and ambiguity in client-consultant relations
- 5. Anders Rom Management accounting and integrated information systems How to exploit the potential for management accounting of information technology
- 6. Marina Candi Aesthetic Design as an Element of Service Innovation in New Technologybased Firms
- Morten Schnack Teknologi og tværfaglighed

 – en analyse af diskussionen omkring indførelse af EPJ på en hospitalsafdeling
- 8. Helene Balslev Clausen Juntos pero no revueltos – un estudio sobre emigrantes norteamericanos en un pueblo mexicano
- 9. Lise Justesen Kunsten at skrive revisionsrapporter. En beretning om forvaltningsrevisionens beretninger
- 10. Michael E. Hansen The politics of corporate responsibility: CSR and the governance of child labor and core labor rights in the 1990s
- 11. Anne Roepstorff Holdning for handling – en etnologisk undersøgelse af Virksomheders Sociale Ansvar/CSR

- 12. Claus Bajlum Essays on Credit Risk and Credit Derivatives
- 13. Anders Bojesen The Performative Power of Competence – an Inquiry into Subjectivity and Social Technologies at Work
- 14. Satu Reijonen Green and Fragile A Study on Markets and the Natural Environment
- 15. Ilduara Busta Corporate Governance in Banking A European Study
- 16. Kristian Anders Hvass A Boolean Analysis Predicting Industry Change: Innovation, Imitation & Business Models The Winning Hybrid: A case study of isomorphism in the airline industry
- 17. Trine Paludan De uvidende og de udviklingsparate Identitet som mulighed og restriktion blandt fabriksarbejdere på det aftayloriserede fabriksgulv
- 18. Kristian Jakobsen Foreign market entry in transition economies: Entry timing and mode choice
- 19. Jakob Elming Syntactic reordering in statistical machine translation
- 20. Lars Brømsøe Termansen Regional Computable General Equilibrium Models for Denmark Three papers laying the foundation for regional CGE models with agglomeration characteristics
- 21. Mia Reinholt The Motivational Foundations of Knowledge Sharing

- 22. Frederikke Krogh-Meibom The Co-Evolution of Institutions and Technology – A Neo-Institutional Understanding of Change Processes within the Business Press – the Case Study of Financial Times
- 23. Peter D. Ørberg Jensen OFFSHORING OF ADVANCED AND HIGH-VALUE TECHNICAL SERVICES: ANTECEDENTS, PROCESS DYNAMICS AND FIRMLEVEL IMPACTS
- 24. Pham Thi Song Hanh Functional Upgrading, Relational Capability and Export Performance of Vietnamese Wood Furniture Producers
- 25. Mads Vangkilde Why wait? An Exploration of first-mover advantages among Danish e-grocers through a resource perspective
- 26. Hubert Buch-Hansen Rethinking the History of European Level Merger Control A Critical Political Economy Perspective

- 1. Vivian Lindhardsen From Independent Ratings to Communal Ratings: A Study of CWA Raters' Decision-Making Behaviours
- 2. Guðrið Weihe Public-Private Partnerships: Meaning and Practice
- 3. Chris Nøkkentved Enabling Supply Networks with Collaborative Information Infrastructures An Empirical Investigation of Business Model Innovation in Supplier Relationship Management
- 4. Sara Louise Muhr Wound, Interrupted – On the Vulnerability of Diversity Management

- 5. Christine Sestoft Forbrugeradfærd i et Stats- og Livsformsteoretisk perspektiv
- 6. Michael Pedersen Tune in, Breakdown, and Reboot: On the production of the stress-fit selfmanaging employee
- Salla Lutz
 Position and Reposition in Networks
 Exemplified by the Transformation of the Danish Pine Furniture Manufacturers
- 8. Jens Forssbæck Essays on market discipline in commercial and central banking
- 9. Tine Murphy Sense from Silence – A Basis for Organised Action How do Sensemaking Processes with Minimal Sharing Relate to the Reproduction of Organised Action?
- 10. Sara Malou Strandvad Inspirations for a new sociology of art: A sociomaterial study of development processes in the Danish film industry
- Nicolaas Mouton On the evolution of social scientific metaphors: A cognitive-historical enquiry into the divergent trajectories of the idea that collective entities – states and societies, cities and corporations – are biological organisms.
- 12. Lars Andreas Knutsen Mobile Data Services: Shaping of user engagements
- 13. Nikolaos Theodoros Korfiatis Information Exchange and Behavior A Multi-method Inquiry on Online Communities

- 14. Jens Albæk Forestillinger om kvalitet og tværfaglighed på sygehuse – skabelse af forestillinger i læge- og plejegrupperne angående relevans af nye idéer om kvalitetsudvikling gennem tolkningsprocesser
- 15. Maja Lotz The Business of Co-Creation – and the Co-Creation of Business
- 16. Gitte P. Jakobsen Narrative Construction of Leader Identity in a Leader Development Program Context
- 17. Dorte Hermansen "Living the brand" som en brandorienteret dialogisk praxis: Om udvikling af medarbejdernes brandorienterede dømmekraft
- 18. Aseem Kinra Supply Chain (logistics) Environmental Complexity
- 19. Michael Nørager How to manage SMEs through the transformation from non innovative to innovative?
- 20. Kristin Wallevik Corporate Governance in Family Firms The Norwegian Maritime Sector
- 21. Bo Hansen Hansen Beyond the Process Enriching Software Process Improvement with Knowledge Management
- 22. Annemette Skot-Hansen Franske adjektivisk afledte adverbier, der tager præpositionssyntagmer indledt med præpositionen à som argumenter En valensgrammatisk undersøgelse
- 23. Line Gry Knudsen Collaborative R&D Capabilities In Search of Micro-Foundations

- 24. Christian Scheuer Employers meet employees Essays on sorting and globalization
- 25. Rasmus Johnsen The Great Health of Melancholy A Study of the Pathologies of Performativity
- 26. Ha Thi Van Pham Internationalization, Competitiveness Enhancement and Export Performance of Emerging Market Firms: Evidence from Vietnam
- 27. Henriette Balieu Kontrolbegrebets betydning for kausativalternationen i spansk En kognitiv-typologisk analyse

- 1. Yen Tran Organizing Innovationin Turbulent Fashion Market Four papers on how fashion firms create and appropriate innovation value
- 2. Anders Raastrup Kristensen Metaphysical Labour Flexibility, Performance and Commitment in Work-Life Management
- 3. Margrét Sigrún Sigurdardottir Dependently independent Co-existence of institutional logics in the recorded music industry
- Ásta Dis Óladóttir Internationalization from a small domestic base: An empirical analysis of Economics and Management
- 5. Christine Secher E-deltagelse i praksis – politikernes og forvaltningens medkonstruktion og konsekvenserne heraf
- 6. Marianne Stang Våland What we talk about when we talk about space:

End User Participation between Processes of Organizational and Architectural Design

- 7. Rex Degnegaard Strategic Change Management Change Management Challenges in the Danish Police Reform
- Ulrik Schultz Brix Værdi i rekruttering – den sikre beslutning En pragmatisk analyse af perception og synliggørelse af værdi i rekrutterings- og udvælgelsesarbejdet
 - Jan Ole Similä Kontraktsledelse Relasjonen mellom virksomhetsledelse og kontraktshåndtering, belyst via fire norske virksomheter

9.

- 10. Susanne Boch Waldorff Emerging Organizations: In between local translation, institutional logics and discourse
- 11. Brian Kane Performance Talk Next Generation Management of Organizational Performance
- 12. Lars Ohnemus Brand Thrust: Strategic Branding and Shareholder Value An Empirical Reconciliation of two Critical Concepts
- 13. Jesper Schlamovitz Håndtering af usikkerhed i film- og byggeprojekter
- Tommy Moesby-Jensen Det faktiske livs forbindtlighed Førsokratisk informeret, ny-aristotelisk ήθος-tænkning hos Martin Heidegger
- 15. Christian Fich Two Nations Divided by Common Values French National Habitus and the Rejection of American Power

- 16. Peter Beyer Processer, sammenhængskraft og fleksibilitet Et empirisk casestudie af omstillingsforløb i fire virksomheder
- 17. Adam Buchhorn Markets of Good Intentions Constructing and Organizing Biogas Markets Amid Fragility and Controversy
- 18. Cecilie K. Moesby-Jensen Social læring og fælles praksis Et mixed method studie, der belyser læringskonsekvenser af et lederkursus for et praksisfællesskab af offentlige mellemledere
- 19. Heidi Boye Fødevarer og sundhed i senmodernismen
 – En indsigt i hyggefænomenet og de relaterede fødevarepraksisser
- 20. Kristine Munkgård Pedersen Flygtige forbindelser og midlertidige mobiliseringer Om kulturel produktion på Roskilde Festival
- 21. Oliver Jacob Weber Causes of Intercompany Harmony in Business Markets – An Empirical Investigation from a Dyad Perspective
- 22. Susanne Ekman Authority and Autonomy Paradoxes of Modern Knowledge Work
- 23. Anette Frey Larsen
 Kvalitetsledelse på danske hospitaler
 Ledelsernes indflydelse på introduktion og vedligeholdelse af kvalitetsstrategier i det danske sundhedsvæsen
- 24. Toyoko Sato Performativity and Discourse: Japanese Advertisements on the Aesthetic Education of Desire

- 25. Kenneth Brinch Jensen Identifying the Last Planner System Lean management in the construction industry
- 26. Javier Busquets Orchestrating Network Behavior for Innovation
- 27. Luke Patey The Power of Resistance: India's National Oil Company and International Activism in Sudan
- 28. Mette Vedel Value Creation in Triadic Business Relationships. Interaction, Interconnection and Position
- 29. Kristian Tørning Knowledge Management Systems in Practice – A Work Place Study
- 30. Qingxin Shi An Empirical Study of Thinking Aloud Usability Testing from a Cultural Perspective
- 31. Tanja Juul Christiansen Corporate blogging: Medarbejderes kommunikative handlekraft
- Malgorzata Ciesielska Hybrid Organisations.
 A study of the Open Source – business setting
- 33. Jens Dick-Nielsen Three Essays on Corporate Bond Market Liquidity
- 34. Sabrina Speiermann Modstandens Politik Kampagnestyring i Velfærdsstaten. En diskussion af trafikkampagners styringspotentiale
- 35. Julie Uldam Fickle Commitment. Fostering political engagement in 'the flighty world of online activism'

- 36. Annegrete Juul Nielsen Traveling technologies and transformations in health care
- 37. Athur Mühlen-Schulte Organising Development Power and Organisational Reform in the United Nations Development Programme
- 38. Louise Rygaard Jonas Branding på butiksgulvet Et case-studie af kultur- og identitetsarbejdet i Kvickly

- 1. Stefan Fraenkel Key Success Factors for Sales Force Readiness during New Product Launch A Study of Product Launches in the Swedish Pharmaceutical Industry
- 2. Christian Plesner Rossing International Transfer Pricing in Theory and Practice
- Tobias Dam Hede
 Samtalekunst og ledelsesdisciplin
 en analyse af coachingsdiskursens genealogi og governmentality
- 4. Kim Pettersson Essays on Audit Quality, Auditor Choice, and Equity Valuation
- 5. Henrik Merkelsen The expert-lay controversy in risk research and management. Effects of institutional distances. Studies of risk definitions, perceptions, management and communication
- 6. Simon S. Torp Employee Stock Ownership: Effect on Strategic Management and Performance
- 7. Mie Harder Internal Antecedents of Management Innovation

- 8. Ole Helby Petersen Public-Private Partnerships: Policy and Regulation – With Comparative and Multi-level Case Studies from Denmark and Ireland
- 9. Morten Krogh Petersen 'Good' Outcomes. Handling Multiplicity in Government Communication
- 10. Kristian Tangsgaard Hvelplund Allocation of cognitive resources in translation - an eye-tracking and keylogging study
- 11. Moshe Yonatany The Internationalization Process of Digital Service Providers
- 12. Anne Vestergaard Distance and Suffering Humanitarian Discourse in the age of Mediatization
- 13. Thorsten Mikkelsen Personligsheds indflydelse på forretningsrelationer
- 14. Jane Thostrup Jagd Hvorfor fortsætter fusionsbølgen udover "the tipping point"? – en empirisk analyse af information og kognitioner om fusioner
- 15. Gregory Gimpel Value-driven Adoption and Consumption of Technology: Understanding Technology Decision Making
- 16. Thomas Stengade Sønderskov Den nye mulighed Social innovation i en forretningsmæssig kontekst
- 17. Jeppe Christoffersen Donor supported strategic alliances in developing countries
- 18. Vibeke Vad Baunsgaard Dominant Ideological Modes of Rationality: Cross functional

integration in the process of product innovation

- 19. Throstur Olaf Sigurjonsson Governance Failure and Icelands's Financial Collapse
- 20. Allan Sall Tang Andersen Essays on the modeling of risks in interest-rate and inflation markets
- 21. Heidi Tscherning Mobile Devices in Social Contexts
- 22. Birgitte Gorm Hansen Adapting in the Knowledge Economy Lateral Strategies for Scientists and Those Who Study Them
- 23. Kristina Vaarst Andersen Optimal Levels of Embeddedness The Contingent Value of Networked Collaboration
- 24. Justine Grønbæk Pors Noisy Management A History of Danish School Governing from 1970-2010
- 25. Stefan Linder Micro-foundations of Strategic Entrepreneurship Essays on Autonomous Strategic Action 4.
- 26. Xin Li Toward an Integrative Framework of National Competitiveness An application to China
- 27. Rune Thorbjørn Clausen Værdifuld arkitektur Et eksplorativt studie af bygningers rolle i virksomheders værdiskabelse
- 28. Monica Viken Markedsundersøkelser som bevis i varemerke- og markedsføringsrett
- 29. Christian Wymann Tattooing The Economic and Artistic Constitution of a Social Phenomenon

- 30. Sanne Frandsen Productive Incoherence A Case Study of Branding and Identity Struggles in a Low-Prestige Organization
- 31. Mads Stenbo Nielsen Essays on Correlation Modelling
- 32. Ivan Häuser Følelse og sprog Etablering af en ekspressiv kategori, eksemplificeret på russisk
- 33. Sebastian Schwenen Security of Supply in Electricity Markets

- Peter Holm Andreasen The Dynamics of Procurement Management - A Complexity Approach
- 2. Martin Haulrich Data-Driven Bitext Dependency Parsing and Alignment
- 3. Line Kirkegaard Konsulenten i den anden nat En undersøgelse af det intense arbejdsliv
 - Tonny Stenheim Decision usefulness of goodwill under IFRS
- 5. Morten Lind Larsen Produktivitet, vækst og velfærd Industrirådet og efterkrigstidens Danmark 1945 - 1958
- 6. Petter Berg Cartel Damages and Cost Asymmetries
- 7. Lynn Kahle Experiential Discourse in Marketing A methodical inquiry into practice and theory
- 8. Anne Roelsgaard Obling Management of Emotions in Accelerated Medical Relationships

- 9. Thomas Frandsen Managing Modularity of Service Processes Architecture
- 10. Carina Christine Skovmøller CSR som noget særligt Et casestudie om styring og meningsskabelse i relation til CSR ud fra en intern optik
- Michael Tell Fradragsbeskæring af selskabers finansieringsudgifter En skatteretlig analyse af SEL §§ 11, 11B og 11C
- 12. Morten Holm *Customer Profitability Measurement Models Their Merits and Sophistication across Contexts*
- 13. Katja Joo Dyppel Beskatning af derivater En analyse af dansk skatteret
- 14. Esben Anton Schultz Essays in Labor Economics Evidence from Danish Micro Data
- 15. Carina Risvig Hansen "Contracts not covered, or not fully covered, by the Public Sector Directive"
- 16. Anja Svejgaard Pors Iværksættelse af kommunikation - patientfigurer i hospitalets strategiske kommunikation
- 17. Frans Bévort Making sense of management with logics An ethnographic study of accountants who become managers
- 18. René Kallestrup The Dynamics of Bank and Sovereign Credit Risk
- 19. Brett Crawford Revisiting the Phenomenon of Interests in Organizational Institutionalism The Case of U.S. Chambers of Commerce

- 20. Mario Daniele Amore Essays on Empirical Corporate Finance
- 21. Arne Stjernholm Madsen The evolution of innovation strategy Studied in the context of medical device activities at the pharmaceutical company Novo Nordisk A/S in the period 1980-2008
- 22. Jacob Holm Hansen Is Social Integration Necessary for Corporate Branding? A study of corporate branding strategies at Novo Nordisk
- 23. Stuart Webber Corporate Profit Shifting and the Multinational Enterprise
- 24. Helene Ratner Promises of Reflexivity Managing and Researching Inclusive Schools
- 25. Therese Strand The Owners and the Power: Insights from Annual General Meetings
- 26. Robert Gavin Strand In Praise of Corporate Social Responsibility Bureaucracy
- 27. Nina Sormunen Auditor's going-concern reporting Reporting decision and content of the report
- 28. John Bang Mathiasen Learning within a product development working practice:
 - an understanding anchored in pragmatism
 - Philip Holst Riis Understanding Role-Oriented Enterprise Systems: From Vendors to Customers

29.

30. Marie Lisa Dacanay Social Enterprises and the Poor Enhancing Social Entrepreneurship and Stakeholder Theory

- 31. Fumiko Kano Glückstad Bridging Remote Cultures: Cross-lingual concept mapping based on the information receiver's prior-knowledge
- 32. Henrik Barslund Fosse Empirical Essays in International Trade
- 33. Peter Alexander Albrecht Foundational hybridity and its reproduction Security sector reform in Sierra Leone
- 34. Maja Rosenstock CSR - hvor svært kan det være? Kulturanalytisk casestudie om udfordringer og dilemmaer med at forankre Coops CSR-strategi
- 35. Jeanette Rasmussen Tweens, medier og forbrug Et studie af 10-12 årige danske børns brug af internettet, opfattelse og forståelse af markedsføring og forbrug
- 36. Ib Tunby Gulbrandsen 'This page is not intended for a US Audience' A five-act spectacle on online communication, collaboration & organization.
- 37. Kasper Aalling Teilmann Interactive Approaches to Rural Development
- Mette Mogensen The Organization(s) of Well-being and Productivity (Re)assembling work in the Danish Post
- 39. Søren Friis Møller
 From Disinterestedness to Engagement 6.
 Towards Relational Leadership In the Cultural Sector
- 40. Nico Peter Berhausen Management Control, Innovation and Strategic Objectives – Interactions and Convergence in Product Development Networks

- 41. Balder Onarheim Creativity under Constraints Creativity as Balancing 'Constrainedness'
- 42. Haoyong Zhou Essays on Family Firms
- 43. Elisabeth Naima Mikkelsen Making sense of organisational conflict An empirical study of enacted sensemaking in everyday conflict at work

- 1. Jacob Lyngsie Entrepreneurship in an Organizational Context
- 2. Signe Groth-Brodersen Fra ledelse til selvet En socialpsykologisk analyse af forholdet imellem selvledelse, ledelse og stress i det moderne arbejdsliv
- 3. Nis Høyrup Christensen Shaping Markets: A Neoinstitutional Analysis of the Emerging Organizational Field of Renewable Energy in China
- 4. Christian Edelvold Berg As a matter of size THE IMPORTANCE OF CRITICAL MASS AND THE CONSEQUENCES OF SCARCITY FOR TELEVISION MARKETS
- 5. Christine D. Isakson Coworker Influence and Labor Mobility Essays on Turnover, Entrepreneurship and Location Choice in the Danish Maritime Industry
 - Niels Joseph Jerne Lennon Accounting Qualities in Practice Rhizomatic stories of representational faithfulness, decision making and control
- 7. Shannon O'Donnell Making Ensemble Possible How special groups organize for collaborative creativity in conditions of spatial variability and distance

- 8. Robert W. D. Veitch Access Decisions in a Partly-Digital World Comparing Digital Piracy and Legal Modes for Film and Music
- 9. Marie Mathiesen Making Strategy Work An Organizational Ethnography
- 10. Arisa Shollo The role of business intelligence in organizational decision-making
- 11. Mia Kaspersen The construction of social and environmental reporting
- 12. Marcus Møller Larsen The organizational design of offshoring
- 13. Mette Ohm Rørdam EU Law on Food Naming The prohibition against misleading names in an internal market context
- 14. Hans Peter Rasmussen GIV EN GED! Kan giver-idealtyper forklare støtte til velgørenhed og understøtte relationsopbygning?
- 15. Ruben Schachtenhaufen Fonetisk reduktion i dansk
- 16. Peter Koerver Schmidt Dansk CFC-beskatning I et internationalt og komparativt perspektiv
- 17.Morten Froholdt26.Strategi i den offentlige sektorEn kortlægning af styringsmæssigkontekst, strategisk tilgang, samtanvendte redskaber og teknologier forudvalgte danske statslige styrelser27.
- Annette Camilla Sjørup Cognitive effort in metaphor translation An eye-tracking and key-logging study 28.

- 19. Tamara Stucchi The Internationalization of Emerging Market Firms: A Context-Specific Study
- 20. Thomas Lopdrup-Hjorth "Let's Go Outside": The Value of Co-Creation
- 21. Ana Alačovska Genre and Autonomy in Cultural Production The case of travel guidebook production
- 22. Marius Gudmand-Høyer Stemningssindssygdommenes historie i det 19. århundrede Omtydningen af melankolien og manien som bipolære stemningslidelser i dansk sammenhæng under hensyn til dannelsen af det moderne følelseslivs relative autonomi. En problematiserings- og erfaringsanalytisk undersøgelse
- 23. Lichen Alex Yu Fabricating an S&OP Process Circulating References and Matters of Concern
- 24. Esben Alfort The Expression of a Need Understanding search
- 25. Trine Pallesen Assembling Markets for Wind Power An Inquiry into the Making of Market Devices
- 26. Anders Koed Madsen Web-Visions Repurposing digital traces to organize social attention
 - Lærke Højgaard Christiansen BREWING ORGANIZATIONAL RESPONSES TO INSTITUTIONAL LOGICS

Tommy Kjær Lassen EGENTLIG SELVLEDELSE En ledelsesfilosofisk afhandling om selvledelsens paradoksale dynamik og eksistentielle engagement

- 29. Morten Rossing Local Adaption and Meaning Creation in Performance Appraisal
- 30. Søren Obed Madsen Lederen som oversætter Et oversættelsesteoretisk perspektiv på strategisk arbejde
- 31. Thomas Høgenhaven Open Government Communities Does Design Affect Participation?
- 32. Kirstine Zinck Pedersen Failsafe Organizing? A Pragmatic Stance on Patient Safety
- 33. Anne Petersen Hverdagslogikker i psykiatrisk arbejde En institutionsetnografisk undersøgelse af hverdagen i psykiatriske organisationer
- 34. Didde Maria Humle Fortællinger om arbejde
- 35. Mark Holst-Mikkelsen Strategieksekvering i praksis – barrierer og muligheder!
- 36. Malek Maalouf Sustaining lean Strategies for dealing with organizational paradoxes
- 37. Nicolaj Tofte Brenneche Systemic Innovation In The Making The Social Productivity of Cartographic Crisis and Transitions in the Case of SEEIT
- Morten Gylling The Structure of Discourse A Corpus-Based Cross-Linguistic Study
- 39. Binzhang YANG
 Urban Green Spaces for Quality Life
 Case Study: the landscape
 architecture for people in Copenhagen

- 40. Michael Friis Pedersen Finance and Organization: The Implications for Whole Farm Risk Management
- 41. Even Fallan Issues on supply and demand for environmental accounting information
- 42. Ather Nawaz Website user experience A cross-cultural study of the relation between users' cognitive style, context of use, and information architecture of local websites
- 43. Karin Beukel The Determinants for Creating Valuable Inventions
- 44. Arjan Markus External Knowledge Sourcing and Firm Innovation Essays on the Micro-Foundations of Firms' Search for Innovation

- 1. Solon Moreira Four Essays on Technology Licensing and Firm Innovation
- 2. Karin Strzeletz Ivertsen Partnership Drift in Innovation Processes A study of the Think City electric car development
- 3. Kathrine Hoffmann Pii Responsibility Flows in Patient-centred Prevention
- 4. Jane Bjørn Vedel Managing Strategic Research An empirical analysis of science-industry collaboration in a pharmaceutical company
- 5. Martin Gylling Processuel strategi i organisationer Monografi om dobbeltheden i tænkning af strategi, dels som vidensfelt i organisationsteori, dels som kunstnerisk tilgang til at skabe i erhvervsmæssig innovation

- Linne Marie Lauesen Corporate Social Responsibility in the Water Sector: How Material Practices and their Symbolic and Physical Meanings Form a Colonising Logic
- 7. Maggie Qiuzhu Mei LEARNING TO INNOVATE: The role of ambidexterity, standard, and decision process
- 8. Inger Høedt-Rasmussen Developing Identity for Lawyers Towards Sustainable Lawyering
- 9. Sebastian Fux Essays on Return Predictability and Term Structure Modelling
- 10. Thorbjørn N. M. Lund-Poulsen Essays on Value Based Management
- 11. Oana Brindusa Albu Transparency in Organizing: A Performative Approach
- 12. Lena Olaison Entrepreneurship at the limits
- 13. Hanne Sørum DRESSED FOR WEB SUCCESS? An Empirical Study of Website Quality in the Public Sector
- 14. Lasse Folke Henriksen Knowing networks How experts shape transnational governance
- 15. Maria Halbinger Entrepreneurial Individuals Empirical Investigations into Entrepreneurial Activities of Hackers and Makers
- 16. Robert Spliid Kapitalfondenes metoder og kompetencer

- 17. Christiane Stelling Public-private partnerships & the need, development and management of trusting A processual and embedded exploration
- 18. Marta Gasparin Management of design as a translation process
- 19. Kåre Moberg Assessing the Impact of Entrepreneurship Education From ABC to PhD
- 20. Alexander Cole Distant neighbors Collective learning beyond the cluster
- 21. Martin Møller Boje Rasmussen Is Competitiveness a Question of Being Alike? How the United Kingdom, Germany and Denmark Came to Compete through their Knowledge Regimes from 1993 to 2007
- 22. Anders Ravn Sørensen Studies in central bank legitimacy, currency and national identity Four cases from Danish monetary history
- 23. Nina Bellak Can Language be Managed in International Business? Insights into Language Choice from a Case Study of Danish and Austrian Multinational Corporations (MNCs)
- 24. Rikke Kristine Nielsen Global Mindset as Managerial Meta-competence and Organizational Capability: Boundary-crossing Leadership Cooperation in the MNC The Case of 'Group Mindset' in Solar A/S.
- 25. Rasmus Koss Hartmann User Innovation inside government Towards a critically performative foundation for inquiry

- 26. Kristian Gylling Olesen Flertydig og emergerende ledelse i folkeskolen Et aktør-netværksteoretisk ledelsesstudie af politiske evalueringsreformers betydning for ledelse i den danske folkeskole
- 27. Troels Riis Larsen Kampen om Danmarks omdømme 1945-2010 Omdømmearbejde og omdømmepolitik
- 28. Klaus Majgaard Jagten på autenticitet i offentlig styring
- 29. Ming Hua Li Institutional Transition and Organizational Diversity: Differentiated internationalization strategies of emerging market state-owned enterprises
- 30. Sofie Blinkenberg Federspiel IT, organisation og digitalisering: Institutionelt arbejde i den kommunale digitaliseringsproces
- 31. Elvi Weinreich
 Hvilke offentlige ledere er der brug for
 når velfærdstænkningen flytter sig
 er Diplomuddannelsens lederprofil
 svaret?
- 32. Ellen Mølgaard Korsager
 Self-conception and image of context in the growth of the firm
 – A Penrosian History of Fiberline Composites
- 33. Else Skjold The Daily Selection
- 34. Marie Louise Conradsen The Cancer Centre That Never Was The Organisation of Danish Cancer Research 1949-1992
- 35. Virgilio Failla Three Essays on the Dynamics of Entrepreneurs in the Labor Market

- 36. Nicky Nedergaard Brand-Based Innovation Relational Perspectives on Brand Logics and Design Innovation Strategies and Implementation
- 37. Mads Gjedsted Nielsen Essays in Real Estate Finance
- 38. Kristin Martina Brandl Process Perspectives on Service Offshoring
- 39. Mia Rosa Koss Hartmann In the gray zone With police in making space for creativity
- 40. Karen Ingerslev Healthcare Innovation under The Microscope Framing Boundaries of Wicked Problems
- 41. Tim Neerup Themsen Risk Management in large Danish public capital investment programmes

- 1. Jakob Ion Wille Film som design Design af levende billeder i film og tv-serier
- 2. Christiane Mossin Interzones of Law and Metaphysics Hierarchies, Logics and Foundations of Social Order seen through the Prism of EU Social Rights
- 3. Thomas Tøth TRUSTWORTHINESS: ENABLING GLOBAL COLLABORATION An Ethnographic Study of Trust, Distance, Control, Culture and Boundary Spanning within Offshore Outsourcing of IT Services
- 4. Steven Højlund Evaluation Use in Evaluation Systems – The Case of the European Commission

- 5. Julia Kirch Kirkegaard *AMBIGUOUS WINDS OF CHANGE – OR FIGHTING AGAINST WINDMILLS IN CHINESE WIND POWER A CONSTRUCTIVIST INQUIRY INTO CHINA'S PRAGMATICS OF GREEN MARKETISATION MAPPING CONTROVERSIES OVER A POTENTIAL TURN TO QUALITY IN CHINESE WIND POWER*
- 6. Michelle Carol Antero A Multi-case Analysis of the Development of Enterprise Resource Planning Systems (ERP) Business Practices

Morten Friis-Olivarius The Associative Nature of Creativity

- Mathew Abraham
 New Cooperativism:
 A study of emerging producer
 organisations in India
- 8. Stine Hedegaard Sustainability-Focused Identity: Identity work performed to manage, negotiate and resolve barriers and tensions that arise in the process of constructing or ganizational identity in a sustainability context
- 9. Cecilie Glerup Organizing Science in Society – the conduct and justification of resposible research
- 10. Allan Salling Pedersen Implementering af ITIL® IT-governance - når best practice konflikter med kulturen Løsning af implementeringsproblemer gennem anvendelse af kendte CSF i et aktionsforskningsforløb.
- 11. Nihat Misir A Real Options Approach to Determining Power Prices
- 12. Mamdouh Medhat MEASURING AND PRICING THE RISK OF CORPORATE FAILURES

13. Rina Hansen Toward a Digital Strategy for Omnichannel Retailing

- 14. Eva Pallesen In the rhythm of welfare creation A relational processual investigation moving beyond the conceptual horizon of welfare management
- 15. Gouya Harirchi In Search of Opportunities: Three Essays on Global Linkages for Innovation
- 16. Lotte Holck Embedded Diversity: A critical ethnographic study of the structural tensions of organizing diversity
- 17. Jose Daniel Balarezo Learning through Scenario Planning
- 18. Louise Pram Nielsen Knowledge dissemination based on terminological ontologies. Using eye tracking to further user interface design.
- 19. Sofie Dam PUBLIC-PRIVATE PARTNERSHIPS FOR INNOVATION AND SUSTAINABILITY TRANSFORMATION An embedded, comparative case study of municipal waste management in England and Denmark
- 20. Ulrik Hartmyer Christiansen Follwoing the Content of Reported Risk Across the Organization
- 21. Guro Refsum Sanden Language strategies in multinational corporations. A cross-sector study of financial service companies and manufacturing companies.
- 22. Linn Gevoll
 Designing performance management
 for operational level
 A closer look on the role of design
 choices in framing coordination and
 motivation

- 23. Frederik Larsen Objects and Social Actions – on Second-hand Valuation Practices
- 24. Thorhildur Hansdottir Jetzek The Sustainable Value of Open Government Data Uncovering the Generative Mechanisms of Open Data through a Mixed Methods Approach
- 25. Gustav Toppenberg Innovation-based M&A – Technological-Integration Challenges – The Case of Digital-Technology Companies
- 26. Mie Plotnikof Challenges of Collaborative Governance An Organizational Discourse Study of Public Managers' Struggles with Collaboration across the Daycare Area
- 27. Christian Garmann Johnsen Who Are the Post-Bureaucrats? A Philosophical Examination of the Creative Manager, the Authentic Leader 39. and the Entrepreneur
- Jacob Brogaard-Kay Constituting Performance Management 40. A field study of a pharmaceutical company
- 29. Rasmus Ploug Jenle Engineering Markets for Control: Integrating Wind Power into the Danish Electricity System
- 30. Morten Lindholst Complex Business Negotiation: Understanding Preparation and Planning
- 31. Morten Grynings TRUST AND TRANSPARENCY FROM AN ALIGNMENT PERSPECTIVE
- 32. Peter Andreas Norn Byregimer og styringsevne: Politisk lederskab af store byudviklingsprojekter

- 33. Milan Miric Essays on Competition, Innovation and Firm Strategy in Digital Markets
- 34. Sanne K. Hjordrup The Value of Talent Management Rethinking practice, problems and possibilities
- 35. Johanna Sax
 Strategic Risk Management
 Analyzing Antecedents and
 Contingencies for Value Creation
- 36. Pernille Rydén Strategic Cognition of Social Media
- 37. Mimmi Sjöklint
 The Measurable Me
 The Influence of Self-tracking on the User Experience
- 38. Juan Ignacio Staricco Towards a Fair Global Economic Regime? A critical assessment of Fair Trade through the examination of the Argentinean wine industry
 - Marie Henriette Madsen Emerging and temporary connections in Quality work
 - Yangfeng CAO Toward a Process Framework of Business Model Innovation in the Global Context Entrepreneurship-Enabled Dynamic Capability of Medium-Sized Multinational Enterprises
- 41. Carsten Scheibye Enactment of the Organizational Cost Structure in Value Chain Configuration A Contribution to Strategic Cost Management

- 1. Signe Sofie Dyrby Enterprise Social Media at Work
- 2. Dorte Boesby Dahl The making of the public parking attendant Dirt, aesthetics and inclusion in public service work
- 3. Verena Girschik Realizing Corporate Responsibility Positioning and Framing in Nascent Institutional Change
- 4. Anders Ørding Olsen IN SEARCH OF SOLUTIONS Inertia, Knowledge Sources and Diversity in Collaborative Problem-solving
- 5. Pernille Steen Pedersen Udkast til et nyt copingbegreb En kvalifikation af ledelsesmuligheder for at forebygge sygefravær ved psykiske problemer.
- 6. Kerli Kant Hvass Weaving a Path from Waste to Value: Exploring fashion industry business models and the circular economy
- 7. Kasper Lindskow Exploring Digital News Publishing Business Models – a production network approach
- 8. Mikkel Mouritz Marfelt The chameleon workforce: Assembling and negotiating the content of a workforce
- 9. Marianne Bertelsen Aesthetic encounters Rethinking autonomy, space & time in today's world of art
- 10. Louise Hauberg Wilhelmsen EU PERSPECTIVES ON INTERNATIONAL COMMERCIAL ARBITRATION

- 11. Abid Hussain On the Design, Development and Use of the Social Data Analytics Tool (SODATO): Design Propositions, Patterns, and Principles for Big Social Data Analytics
- 12. Mark Bruun Essays on Earnings Predictability
- 13. Tor Bøe-Lillegraven BUSINESS PARADOXES, BLACK BOXES, AND BIG DATA: BEYOND ORGANIZATIONAL AMBIDEXTERITY
- 14. Hadis Khonsary-Atighi ECONOMIC DETERMINANTS OF DOMESTIC INVESTMENT IN AN OIL-BASED ECONOMY: THE CASE OF IRAN (1965-2010)
- 15. Maj Lervad Grasten Rule of Law or Rule by Lawyers? On the Politics of Translation in Global Governance
- 16. Lene Granzau Juel-Jacobsen SUPERMARKEDETS MODUS OPERANDI – en hverdagssociologisk undersøgelse af forholdet mellem rum og handlen og understøtte relationsopbygning?
- 17. Christine Thalsgård Henriques
 In search of entrepreneurial learning
 Towards a relational perspective on incubating practices?
- 18. Patrick Bennett Essays in Education, Crime, and Job Displacement
- 19. Søren Korsgaard Payments and Central Bank Policy
- 20. Marie Kruse Skibsted Empirical Essays in Economics of Education and Labor
- 21. Elizabeth Benedict Christensen The Constantly Contingent Sense of Belonging of the 1.5 Generation Undocumented Youth An Everyday Perspective

- 22. Lasse J. Jessen Essays on Discounting Behavior and Gambling Behavior
- 23. Kalle Johannes Rose Når stifterviljen dør... Et retsøkonomisk bidrag til 200 års juridisk konflikt om ejendomsretten
- 24. Andreas Søeborg Kirkedal Danish Stød and Automatic Speech Recognition
- 25. Ida Lunde Jørgensen Institutions and Legitimations in Finance for the Arts
- 26. Olga Rykov Ibsen An empirical cross-linguistic study of directives: A semiotic approach to the sentence forms chosen by British, Danish and Russian speakers in native and ELF contexts
- 27. Desi Volker Understanding Interest Rate Volatility
- 28. Angeli Elizabeth Weller Practice at the Boundaries of Business Ethics & Corporate Social Responsibility
- 29. Ida Danneskiold-Samsøe Levende læring i kunstneriske organisationer En undersøgelse af læringsprocesser mellem projekt og organisation på Aarhus Teater
- 30. Leif Christensen Quality of information – The role of internal controls and materiality
- 31. Olga Zarzecka Tie Content in Professional Networks
- 32. Henrik Mahncke De store gaver
 - Filantropiens gensidighedsrelationer i teori og praksis
- 33. Carsten Lund Pedersen Using the Collective Wisdom of Frontline Employees in Strategic Issue Management

- 34. Yun Liu Essays on Market Design
- 35. Denitsa Hazarbassanova Blagoeva The Internationalisation of Service Firms
- 36. Manya Jaura Lind Capability development in an offshoring context: How, why and by whom
- 37. Luis R. Boscán F. Essays on the Design of Contracts and Markets for Power System Flexibility
- 38. Andreas Philipp Distel Capabilities for Strategic Adaptation: Micro-Foundations, Organizational Conditions, and Performance Implications
- 39. Lavinia Bleoca The Usefulness of Innovation and Intellectual Capital in Business Performance: The Financial Effects of Knowledge Management vs. Disclosure
- 40. Henrik Jensen Economic Organization and Imperfect Managerial Knowledge: A Study of the Role of Managerial Meta-Knowledge in the Management of Distributed Knowledge
- 41. Stine Mosekjær The Understanding of English Emotion Words by Chinese and Japanese Speakers of English as a Lingua Franca An Empirical Study
- 42. Hallur Tor Sigurdarson The Ministry of Desire - Anxiety and entrepreneurship in a bureaucracy
- 43. Kätlin Pulk Making Time While Being in Time A study of the temporality of organizational processes
- 44. Valeria Giacomin Contextualizing the cluster Palm oil in Southeast Asia in global perspective (1880s–1970s)

- 45. Jeanette Willert Managers' use of multiple Management Control Systems: The role and interplay of management control systems and company performance
- 46. Mads Vestergaard Jensen Financial Frictions: Implications for Early Option Exercise and Realized Volatility
- 47. Mikael Reimer Jensen Interbank Markets and Frictions
- 48. Benjamin Faigen Essays on Employee Ownership
- 49. Adela Michea Enacting Business Models An Ethnographic Study of an Emerging Business Model Innovation within the Frame of a Manufacturing Company.
- 50. Iben Sandal Stjerne Transcending organization in temporary systems Aesthetics' organizing work and employment in Creative Industries
- 51. Simon Krogh Anticipating Organizational Change
- 52. Sarah Netter Exploring the Sharing Economy
- 53. Lene Tolstrup Christensen State-owned enterprises as institutional market actors in the marketization of public service provision: A comparative case study of Danish and Swedish passenger rail 1990–2015
- 54. Kyoung(Kay) Sun Park Three Essays on Financial Economics

- 2017
- 1. Mari Bjerck Apparel at work. Work uniforms and women in male-dominated manual occupations.
- 2. Christoph H. Flöthmann Who Manages Our Supply Chains? Backgrounds, Competencies and Contributions of Human Resources in Supply Chain Management
- 3. Aleksandra Anna Rzeźnik Essays in Empirical Asset Pricing
- 4. Claes Bäckman Essays on Housing Markets
- 5. Kirsti Reitan Andersen Stabilizing Sustainability in the Textile and Fashion Industry
- 6. Kira Hoffmann Cost Behavior: An Empirical Analysis of Determinants and Consequences of Asymmetries
- 7. Tobin Hanspal Essays in Household Finance
- 8. Nina Lange Correlation in Energy Markets
- 9. Anjum Fayyaz Donor Interventions and SME Networking in Industrial Clusters in Punjab Province, Pakistan
- 10. Magnus Paulsen Hansen Trying the unemployed. Justification and critique, emancipation and coercion towards the 'active society'. A study of contemporary reforms in France and Denmark
- Sameer Azizi
 Corporate Social Responsibility in Afghanistan

 a critical case study of the mobile telecommunications industry

- 12. Malene Myhre The internationalization of small and medium-sized enterprises: A qualitative study
- 13. Thomas Presskorn-Thygesen The Significance of Normativity – Studies in Post-Kantian Philosophy and Social Theory
- 14. Federico Clementi Essays on multinational production and international trade
- Lara Anne Hale Experimental Standards in Sustainability 26. Transitions: Insights from the Building Sector
- 16. Richard Pucci Accounting for Financial Instruments in 27. an Uncertain World Controversies in IFRS in the Aftermath of the 2008 Financial Crisis
- 17. Sarah Maria Denta Kommunale offentlige private partnerskaber Regulering I skyggen af Farumsagen
- 18. Christian Östlund Design for e-training
- 19. Amalie Martinus Hauge Organizing Valuations – a pragmatic inquiry
- 20. Tim Holst Celik Tension-filled Governance? Exploring the Emergence, Consolidation and Reconfiguration of Legitimatory and Fiscal State-crafting
- 21. Christian Bason Leading Public Design: How managers engage with design to transform public governance
- 22. Davide Tomio Essays on Arbitrage and Market Liquidity

- 23. Simone Stæhr Financial Analysts' Forecasts Behavioral Aspects and the Impact of Personal Characteristics
- 24. Mikkel Godt Gregersen Management Control, Intrinsic Motivation and Creativity – How Can They Coexist
- 25. Kristjan Johannes Suse Jespersen Advancing the Payments for Ecosystem Service Discourse Through Institutional Theory
 - . Kristian Bondo Hansen Crowds and Speculation: A study of crowd phenomena in the U.S. financial markets 1890 to 1940
 - Lars Balslev Actors and practices – An institutional study on management accounting change in Air Greenland
- 28. Sven Klingler Essays on Asset Pricing with Financial Frictions
- 29. Klement Ahrensbach Rasmussen Business Model Innovation The Role of Organizational Design
- 30. Giulio Zichella Entrepreneurial Cognition. Three essays on entrepreneurial behavior and cognition under risk and uncertainty

TITLER I ATV PH.D.-SERIEN

1992

1. Niels Kornum Servicesamkørsel – organisation, økonomi og planlægningsmetode

1995

2. Verner Worm Nordiske virksomheder i Kina Kulturspecifikke interaktionsrelationer ved nordiske virksomhedsetableringer i Kina

1999

3. Mogens Bjerre Key Account Management of Complex Strategic Relationships An Empirical Study of the Fast Moving Consumer Goods Industry

2000

4. Lotte Darsø Innovation in the Making Interaction Research with heterogeneous Groups of Knowledge Workers creating new Knowledge and new Leads

2001

5. Peter Hobolt Jensen Managing Strategic Design Identities The case of the Lego Developer Network

2002

- 6. Peter Lohmann The Deleuzian Other of Organizational Change – Moving Perspectives of the Human
- Anne Marie Jess Hansen To lead from a distance: The dynamic interplay between strategy and strategizing – A case study of the strategic management process

2003

- Lotte Henriksen Videndeling

 om organisatoriske og ledelsesmæssige udfordringer ved videndeling i praksis
- 9. Niels Christian Nickelsen Arrangements of Knowing: Coordinating Procedures Tools and Bodies in Industrial Production – a case study of the collective making of new products

2005

10. Carsten Ørts Hansen Konstruktion af ledelsesteknologier og effektivitet

TITLER I DBA PH.D.-SERIEN

2007

1. Peter Kastrup-Misir Endeavoring to Understand Market Orientation – and the concomitant co-mutation of the researched, the re searcher, the research itself and the truth

2009

1. Torkild Leo Thellefsen Fundamental Signs and Significance effects

> A Semeiotic outline of Fundamental Signs, Significance-effects, Knowledge Profiling and their use in Knowledge Organization and Branding

2. Daniel Ronzani

When Bits Learn to Walk Don't Make Them Trip. Technological Innovation and the Role of Regulation by Law in Information Systems Research: the Case of Radio Frequency Identification (RFID)

2010

1. Alexander Carnera Magten over livet og livet som magt Studier i den biopolitiske ambivalens